The European Union as a Global Counter-Terrorism Actor

EUROPEAN SECURITY AND JUSTICE CRITIQUES

Series Editors: Christian Kaunert, Professor, Dublin City University and University of South Wales and Sarah Léonard, Professor, School of Social Sciences, University of the West of England

Editorial Board: Andrew Geddes, Professor, Department of Politics, University of Sheffield, UK, Emil Kirchner, Jean Monnet Chair in European Political Integration, Department of Government, University of Essex, UK, John Occhipinti, Director of European Studies, Department of Political Science, Canisius College, New York, USA, Knud Erik Joergensen, Professor, Department of Political Science and Government, Aarhus University, Denmark, Kamil Zwolski, Lecturer in Global Politics and Policy, University of Southampton, UK, Valsamis Mitsilegas, Head of the Department of Law, Professor of European Criminal Law, Queen Mary University of London, UK, Dora Kostakopoulou, Professor of European Union Law, European Integration and Public Policy, Warwick Law School, Warwick, UK and Monica den Boer, Faculty of Social Sciences, VU University Amsterdam, the Netherlands

What is security in the twenty-first century? Recently, issues such as crime, terrorism, energy security, population movements and migration, environmental and climate security, underdevelopment, health and pandemic diseases, as well as economic security have become important matters of European security. Furthermore, more traditional security challenges, such as military conflicts and the proliferation of weapons of mass destruction (WMDs), have evolved and changed character. In this context, an increasing body of scholarship has recognized the European Union (EU) as a security actor, which is well equipped to tackle these complex security challenges, due to the multi-dimensional or structural character of its policies.

Based on a broad definition of security, this unprecedented series examines the role of Europe in international security and justice matters. Taking inspiration from the European Security Strategy (ESS), it identifies more traditional security threats such as WMDs, but also points to global warming and poverty as the possible cause of conflict. Moreover, the series promises to recognize the intrinsic links between security and justice from a multitude of disciplinary perspectives and provides a platform for research that focuses on the study of Europe as a whole, going beyond only the European Union.

The series and its individual publications will be of interest to both students and academics across the areas of politics, international relations, law, sociology, history, area studies, gender studies, post-colonial studies and human and political geography.

Titles in the series include:

EU Migration Agencies The Operation and Cooperation of FRONTEX, EASO and EUROPOL David Fernández-Rojo

EU, Turkey and Counter-Terrorism Fighting the PKK and ISIS Ethem Ilbiz and Christian Kaunert

The European Union as a Global Counter-Terrorism Actor Christian Kaunert, Alex MacKenzie and Sarah Léonard

The European Union as a Global Counter-Terrorism Actor

Christian Kaunert

Dublin City University, Republic of Ireland and University of South Wales, UK

Alex MacKenzie

University of Liverpool, UK

Sarah Léonard

University of the West of England, UK

EUROPEAN SECURITY AND JUSTICE CRITIQUES



Cheltenham, UK • Northampton, MA, USA



This is an open access work distributed under the Creative Commons Attribution-NonCommercial-NoDerivatives 4.0 Unported (https://creativecommons.org/licenses/by-nc-nd/4.0/). Users can redistribute the work for non-commercial purposes, as long as it is passed along unchanged and in whole, as detailed in the License. Edward Elgar Publishing Ltd must be clearly credited as the rights holder for publication of the original work. Any translation or adaptation of the original content requires the written authorization of Edward Elgar Publishing Ltd.

Published by Edward Elgar Publishing Limited The Lypiatts 15 Lansdown Road Cheltenham Glos GL50 2JA UK

Edward Elgar Publishing, Inc. William Pratt House 9 Dewey Court Northampton Massachusetts 01060 USA

A catalogue record for this book is available from the British Library

Library of Congress Control Number: 2022938990

This book is available electronically in the Political Science and Public Policy subject collection http://dx.doi.org/10.4337/9781782548287

ISBN 978 1 78254 827 0 (cased) ISBN 978 1 78254 828 7 (eBook)

To my daughter Sophia and my son Max. May they grow up in a better world than ours. Alex MacKenzie

To Elisabeth and Raphael, whom we love so very much.

Christian Kaunert and Sarah Léonard

Contents

Preface Acknowledgements		viii ix
	oduction to The European Union as a Global unter-Terrorism Actor	1
1	EU counter-terrorism, collective securitization, and global actorness	19
2	The collective securitization of terrorism in the EU	35
3	Institutions in EU counter-terrorism	50
4	Counter-terrorism in the transatlantic relationship	71
5	EU counter-terrorism and South Asia	90
6	EU counter-terrorism, Iraq, and Syria	108
7	Conclusion to The European Union as a Global Counter-Terrorism Actor	125
Bibliography Index		138 176

Preface

This book is about European Union (EU) counter-terrorism cooperation, both conceptually and empirically. It is based on many vivid exchanges that the authors of this book have had over the years. We are grateful to a great number of people – family members, friends, and colleagues – who have helped through the various stages of the writing of this book. Without these very special people, it would not have been possible to complete this project.

The findings presented in this book are based on research supported by various institutions, including Dublin City University, the University of South Wales, the University of Dundee, the University of Liverpool, and the University of the West of England (UWE Bristol). Christian Kaunert and Sarah Léonard also wish to thank the European Union for funding their respective Marie Curie Career Integration Grants (held at the University of Dundee and Vesalius College, Vrije Universiteit Brussel). This book has also been written with the support of the Erasmus+ programme of the European Union (Jean Monnet Actions), notably Christian Kaunert's Jean Monnet Chair EUCTIRL in EU Counter-Terrorism, as well as the Jean Monnet Centre of Excellence VIRTUE, the Jean Monnet Module EUHYBRID, and, finally, the Jean Monnet Network in EU Counter-Terrorism EUCTER. While writing this book, we also benefited enormously from the comments and insights of numerous experts in the field. We thank them for their time and support.

Moreover, throughout the publication process, the entire team at Edward Elgar has provided extremely useful comments and suggestions. The support and encouragement by Alex Pettifer has been exceptional and very much appreciated. We also thank the anonymous reviewers for their helpful comments and suggestions.

We reserve a final mention for those individuals who have offered tremendous support throughout this book adventure. First and foremost, we sincerely thank our respective parents and families, who have always been a source of inspiration. We also express our appreciation to our friends, who have provided encouragement at just the right moment. Finally, Christian and Sarah would like to thank Francesco for his wonderful hospitality.

Alex MacKenzie, Christian Kaunert and Sarah Léonard Liverpool and Barry/Dublin

Acknowledgements

The authors acknowledge their debt to those many people who have helped and influenced the approach and the findings presented in this book. Some of the research findings presented in this book have been disseminated in a number of journal articles, including: C. Kaunert and S. Léonard (2019), 'The Collective Securitization of Terrorism in the EU', *West European Politics*, 41(2), pp. 261–77 (https://doi.org/10.1080/01402382.2018.1510194). The authors thank the publishers for allowing them to use this material in this book.

Introduction to *The European Union as* a Global Counter-Terrorism Actor

One of the founding fathers of the European Union (EU) highlighted the importance of crises as catalysts for change. This book illustrates this principle by analysing the development of the EU's role as a global counter-terrorism actor, which was prompted by the deadly terrorist attacks that took place on 11 September 2001 (also often referred to as '9/11'). These attacks led the EU to join in the so-called 'war on terror', whereby it collectively securitized the terrorist threat to become European, rather than national. This is the first book-length treatment of the EU as a global counter-terrorism actor. Europe has been the site of much and heterogeneous terrorist activity in the twentieth and twenty-first centuries, from the Provisional Irish Republican Army (PIRA) through the German Red Army Faction (RAF) and the Croatian Ustasha movement to al-Qaeda- and Daesh-inspired and affiliated individuals and groups. These experiences have provided European states with considerable experience in combating terrorism. Jihadist terrorism has been the main focus of counter-terrorism efforts in many Western states since 2001, although there has been growing concern about right-wing terrorism, whilst separatist terrorist groups have continued to be active in a number of European states (Europol, 2018). Europe has seen significant attacks from jihadist terrorists since 2001, including the Madrid train bombings in March 2004, which killed over 190 people and injured a further 1,841 victims (BBC, 2011).

Despite the threat that it poses to European states, the issue of terrorism needs to be placed into perspective. As pointed out by Mueller (2005), citizens of the USA and other Western states are not likely to be directly affected by terrorism and it may be dangerous to over-react. Injudicious action might marginalise ethnic minorities, polarise societies, and erode democratic norms. Having said this, most terrorist activity occurs outside of Europe and is concentrated in a small number of states. For instance, more than half of the deaths from terrorism in 2017 took place in just three countries, namely Iraq (24 per cent), Afghanistan (23 per cent), and Syria (8 per cent) (Global Terrorism Database, 2018). States such as Afghanistan, Iraq, Nigeria, Pakistan, and Syria have experienced high levels of terrorist activity for a number of years since the attacks of 11 September 2001 and the subsequent response of the USA and its partners.

What is known in the West as al-Qaeda, the group responsible for the attacks on the World Trade Center in September 2001, was subsequently overshadowed by Daesh as the leading jihadist group (Byman, 2017: 1106). There has been a debate over conceptualising the precise nature of Daesh. Cronin (2016) has argued that it has been more than a terrorist group, whilst Walt (2015) has likened it to a revolutionary state akin to revolutionary France in the late eighteenth century. What is certain is that it has inspired terrorist attacks around the world, including the coordinated attacks on Paris in November 2015, and has counted over 5,000 European citizens in its numbers (Coolsaet and Renard, 2018). It experienced a level of success in gaining a significant amount of territory in Iraq and Syria, which led other groups to pledge allegiance to it. Eventually, it was pushed back and its main stronghold Ragga taken in October 2017, although fighting has continued in the region. According to Rapoport's (2012: 41) 'wave theory' of modern international terrorism, the religious wave, which started in 1979, probably does not have long to run, with his rough prediction being that it will end sometime round about 2025 – about 40 years, or the length of a generation. Others have agreed that the fourth wave appears to be coming to an end (Weinberg and Eubank, 2010), and perhaps, in hindsight, the destruction of the central body of Daesh will come to be seen as a crucial event in that respect. Usually, a wave of terrorism ends as individuals consider that the objectives are impossible to achieve or as a result of a failure to hand over to the next generation.

There has already been speculation about the emergence of a fifth wave of international terrorism (Honig and Yahel, 2017; Kaplan, 2007). Indeed, terrorism as a method is unlikely to completely end. One can argue that the very belief that this was possible was actually one of the many errors underpinning the American ill-fated 'war on terror'. In that respect, Byman (2018) has suggested that, after its defeat in the Middle East, Daesh might go underground, disrupt politics and encourage sectarianism, continue with an insurgency, and then return, while also continuing to attack Europe. In that regard, European governments have expressed concerns about the return of foreign fighters and the decentralisation of the jihadist movement. This is a rather complex situation since about a third of foreign fighters have returned to Europe, whilst many others have been killed or imprisoned in the Middle East (European Parliament, 2018).

WHAT ROLE FOR THE EU?

This book focuses on the growth of the EU as a global counter-terrorism actor since the start of the twenty-first century, as it has gone from having almost no role in counter-terrorism in 2001 to being a significant actor 20 years later. If indeed the fourth wave of international terrorism is coming to an end, then

this book will have covered much of the arc of the EU's development so far. A further wave of terrorism may be different to the current one in terms of focus, methods used, and geographical presence. Far-right terrorism has been on the rise and has grown faster than other types of terrorism. In Western states, the far-right accounted for 17.2 per cent of attacks in 2018 and had risen by 320 per cent in the five previous years (Global Terrorism Index, 2019). Today's political environment is likely contributing to this increase, which has led respected scholars to speculate that the 'fifth wave' of terrorism might come from the right (Weinberg, 2011).

European states have cooperated on terrorism for a few years. Bures (2006: 58) traces the origins of EU counter-terrorism cooperation back to European Political Cooperation (EPC) and the Trevi Group in the 1970s – a time when many European states, such as the United Kingdom (UK), Germany, France, and Italy were experiencing significant problems with terrorism. In the early 1990s, such cooperation efforts were formally brought into the EU's framework in the form of the then new intergovernmental 'Justice and Home Affairs' (JHA) third pillar, which was introduced by the Treaty of Maastricht in 1992. However, as the 1990s progressed, terrorism declined in Europe. Many leftist groups wound down after the end of the Cold War, whilst decreasing violence in Northern Ireland led to the Good Friday Agreement in 1998. This meant that there was little urgency to improve counter-terrorism cooperation in Europe in this decade. Even so, some incremental changes did occur, such as the inclusion of terrorism as an issue for the EU in the Treaty of Maastricht, the promise of the development of an Area of Freedom, Security, and Justice (AFSJ), and the ability for the EU to negotiate and to conclude some types of security agreements with third states in the Treaty of Amsterdam in 1997 (Monar, 2007b: 272). Moreover, the European Council meeting at Tampere in 1999 provided a wish list of measures that only became possible after the attacks of 9/11 due to, among other things, entrepreneurship by EU institutions (Kaunert, 2007). The 9/11 attacks on the USA caused a major change in the EU's role in combating terrorism.

More precisely, 9/11 led to what has been referred to as the 'month of transformation' of the EU, in which the EU rushed to respond to the attacks in a somewhat hectic manner (Zimmermann, 2006). Although these activities overloaded the EU, a lack of coherence was present, and there have been some implementation problems, much has also been achieved at the EU level (Argomaniz, 2010; Bures, 2011; Bossong, 2013; De Londras and Doody, 2017). The list of achievements has grown over time, but those have often come as a response to terrorist attacks (Argomaniz, 2009a; Davis Cross, 2017). In 2008, Edwards and Meyer (2008: 1) pointed out that '[t]he governance of the European Union has been changed through its responses to international terrorism', which highlights the profound effect of terrorism on the EU after

only seven years. Indeed, international terrorism has really pushed European integration forward. A list of some important achievements is worth providing. In 2002, the EU agreed on a common definition of terrorism. This definition has both an objective and a subjective dimension. The objective dimension refers to the adoption of a list of nine categories of serious criminal conduct, such as hostage-taking, extortion, murder, and bodily injuries. In contrast, the subjective dimension concerns the aim of these acts – the specific acts that are listed are deemed to be terrorist acts when committed with a specific aim, namely (1) 'seriously intimidating a population', (2) 'unduly compelling a Government or international organisation to perform or abstain from performing any act', or (3) 'seriously destabilising or destroying the fundamental political, constitutional, economic or social structures of a country or an international organisation' (Council Framework Decision 2002/475/JHA). Following the adoption of this definition – which will be used throughout this book – all EU member states were required to incorporate it into their legal systems. Furthermore, the EU appointed an EU Counter-Terrorism Coordinator (CTC) after the Madrid bombings in 2004, and despite problems with the role, the importance attached to the CTC has grown over time (MacKenzie et al., 2013b). The EU also introduced its first European Security Strategy in 2003, a Counter-Terrorism Strategy in 2005, and an Internal Security Strategy in 2010, with some subsequent updates. With regard to externally combating terrorism, the Union has engaged heavily with the USA and announced long-term strategies for areas of the world where terrorism poses a challenge, such as the Middle East, North Africa, and Afghanistan. These are, of course, rhetorical commitments, but concrete action has followed in a number of these cases, as this book shows in its many case studies.

The initial perception of the EU's external counter-terrorism role was overall not promising. For instance, the EU was presented as an 'absent friend' by Keohane (2008). However, this piece had several limitations. First, it focused on only three counter-terrorism tools -promoting UN Conventions. dialogues on counter-terrorism, and counter-terrorism assistance - and it would certainly be difficult to identify the importance of any actor in relation to the first two components at least. Second, the paper did not consider the different roles of the EU and neglected much of an area that later became known as the 'external dimension of JHA', including EU-USA cooperation (Wolff et al., 2009). Third, the article also appeared to compare the EU to the USA. For the foreseeable future, the EU will only seriously be able to compete with the USA in the economic sphere. Comparisons with the USA are difficult to make given that the EU is not a unified state and it does not have the resources of the USA. It is certainly the case that the 'war on terror' has been problematic and something of an obstacle for the EU's alternative 'fight against terrorism' because the EU could have been 'drowned out'. Also, the EU's role has been

and will remain largely supplementary to that of the member states, although it does have a range of its own instruments and some issues are now governed at the supranational level. Nonetheless, some literature has identified the EU's growing role in counter-terrorism, highlighting EU cooperation with, for example, the USA (Argomaniz, 2009b; Kaunert, 2010a; Occhipinti, 2010) and North Africa (Eder, 2011; Joffe, 2008; MacKenzie et al., 2013a; Wolff, 2009), and the EU as a counter-terrorism actor in the making, while also demonstrating EU activity in areas beyond its established relationships and outside states in its vicinity (Beyer, 2008; MacKenzie, 2010; Monar, 2015; Spence, 2007).

Sandler (2015: 12) has stated that counter-terrorism 'corresponds to actions to ameliorate the threat and consequences of terrorism'. This is a broad definition, which could encompass a very wide range of activities ranging from countering the financing of terrorism through strengthening borders all the way to education for preventing radicalisation or health measures strengthening preparedness for an attack. The EU has resolutely viewed terrorism as a crime and has therefore focused on a 'law enforcement' approach to counter-terrorism, which refers to prioritising international legal instruments and building up law enforcement agencies to deal with a security challenge (Monar, 2015: 337). As well as being the result of hard lessons learnt by European states when deploying force against terrorism and therefore perhaps developed as part of strategic culture (Rees and Aldrich, 2005), this approach is also something of a default option for the EU given its capabilities. Indeed, much EU action against terrorism – in addition to building up law enforcement - seems to be predicated theoretically on a structural approach focusing on the root causes of terrorism. This is perhaps not surprising given EU strengths in aid, but this has posed a serious question relating to the 'securitization' of such instruments. Nevertheless, there is no reason why small-scale EU military missions could not be deployed against terrorist threats or at least play a role in enhancing the law enforcement capabilities of partner states. Although the role of the EU has rarely been significant enough to defeat a terrorist group or even to play a decisive role, Hoffman (1999: 63), even prior to 9/11, stated that 'there is reason to believe that the European way of doing things might vield more effective results in the long term'. The evidence against a military approach is now significant. Studies focusing on the end of terrorist groups have shown that few terrorist campaigns have been brought to an end through military action, while many more have ended through politics and police action (Cronin, 2009; Weinberg, 2011). Even if the EU may not be the most significant actor in counter-terrorism, it is capable of diffusing European methods around the world, which may prove effective in helping other states deal with terrorism in the long term. This book examines a range of areas that can be considered relevant to counter-terrorism, as explained in greater detail later.

The Internal Dimension of EU Counter-Terrorism

The internal dimension of EU counter-terrorism was the focus of much of the early literature on EU counter-terrorism cooperation. This is unsurprising because the EU has been much more developed internally, including as a market, than externally in terms of security. In fact, internal developments often had to occur in order to make external actions possible. den Boer and Monar (2002: 11), in one of the first important and also wide-ranging contributions to the literature that examined internal and external dimensions, discussed the challenges that international terrorism has posed to modern states, the EU's struggles to develop as a security actor and to respond appropriately alongside the USA, as well as the internal/external nexus and cross-pillar tests. They concluded that the EU's response to 9/11 had been 'at least respectable' (Den Boer and Monar, 2002: 26).

Although there has been a significant growth in studies focusing on EU counter-terrorism cooperation, the idea that the EU's counter-terrorism policy is mainly a 'paper tiger' has been rather dominant (Bures, 2006, 2011). In short, Bures (2011: 2) has argued that, in many areas, 'the EU's counter-terrorism policy is more of a paper tiger than it is an effective counter-terrorism device', although 'in some aspects and areas, the EU counter-terrorism policy is, albeit slowly and inconsistently, becoming a real tiger'. In other words, overall, the EU's counter-terrorism policy has looked better on paper than it has worked in practice, although some progress has been made. At the same time, it is important to recognise how challenging it is to measure the effectiveness of counter-terrorism measures in general. For instance, the absence of terrorist attacks cannot necessarily be equated with an effective and successful counter-terrorism policy. Keohane (2005) has formulated similar criticisms. He has discussed how the EU has put together lists about action, but how there have been implementation problems, among other challenges. Bossong (2008a) has also contributed to the debate on primarily internal issues in EU counter-terrorism cooperation, arguing that the EU's action plan on combating terrorism has been a flawed tool of governance.

Following on from Bures and Keohane and perhaps even more critically, Zimmermann (2006: 123) has asked a more fundamental question, namely whether the EU is even an appropriate forum in Europe for combating terrorism or whether this should be left to the member states or an alternative organisation. The problem with this perspective is that the EU is a major regional organisation that holds many supranational powers. Therefore, attempting to confront terrorism by another route in Europe might lead to two organisations working in different directions or to the duplication of tasks. Underpinning Zimmermann's article is a belief that past experience of combating terrorism has prevented Europe from adapting to the threat of jihadist terrorism. This

would seem to suggest that al-Qaeda is a completely new phenomenon. However, Duyvesteyn (2004) has argued that there is actually little new about the supposedly 'new terrorism' and that it is precisely by assuming a break with the past and not drawing on previous experiences that mistakes have been made. Indeed, leading terrorism scholars, such as English (2009), have argued that it is necessary to learn the lessons of the past in order to successfully deal with new threats.

Furthermore, scholars have drawn attention to a further problem, which is that EU member states have not all seen the threat of terrorism in the same way (Bures, 2010a; Monar, 2007a). In general, the older member states that have experienced terrorism have tended to be more concerned than the newer member states, which have not shared their experiences. Such a lack of a common threat perception has posed challenges to the development of an EU common counter-terrorism policy at times. In addition, an increasing range of scholars have argued that the AFSJ has focused too much on the issue of security and not enough on freedom and justice (den Boer and Monar, 2002; Guild, 2008, 2010; Ripoll Servent and MacKenzie, 2011, 2012). With counter-terrorism, this has been clear in the controversial cases of, among other things, data retention measures that have been introduced following terrorist attacks, such as the Data Retention Directive (DRD) and EU Passenger Name Records (EU PNR) (Ripoll Servent, 2013; Ripoll Servent and MacKenzie, 2017). Although it is not its main focus, this book makes a contribution to these debates. In that respect, the book is based on an understanding that the erosion of democratic norms and the idea that terrorism justifies exceptional and unorthodox means of action are extremely concerning.

Nevertheless, others have taken a more positive view of the EU's internal role in counter-terrorism. For instance, Kaunert (2007) has shown that European integration could find its way into 'high politics' through the entrepreneurship of the European Commission in the introduction of the European Arrest Warrant (EAW), one of the EU's flagship responses to 9/11. Also, as previously observed, Edwards and Meyer (2008) have highlighted how the EU has been influenced by terrorism. More specifically, they have shown how coordination and information-sharing have become of greater importance, but that integration has not followed. Spence (2007a: 168) has also pointed out how '[the EU's] response since 2001 has brought a new dynamic to EU domestic and foreign policy making and thus to the process of European integration itself'. Finally, Monar (2014: 195) has offered the optimistic view that, in the case of counter-terrorism, the EU has played an important role, looking at how 'common threat assessments guide governance responses, and specific institutional structures, cooperation mechanisms, legal instruments, and forms of external action have been put into place to respond to the cross-border nature of the terrorist challenges'. He has characterised internal

EU security governance 'as an advanced institutionalized system of cooperation and coordination between national governance frameworks constructed around a core of common instruments and procedures with a cross-border reach' (Monar, 2014: 195). For their part, in their assessment of a decade of EU counter-terrorism, Argomaniz et al. (2015: 196) have stated that '[t]he consensus view in this volume is that the European Union has accomplished a surprising amount in the past decade'. They have argued that a combination of political and institutional factors can explain the development of the EU as a player in counter-terrorism when compared to 9/11, but they have also confirmed that much more action by the EU has occurred within compared to outside Europe's borders (Argomaniz et al., 2015: 198–9).

Thus, several problems have been highlighted in the EU's internal approach to counter-terrorism, including an emphasis over quantity rather than quality of action at times, implementation problems, policy overload, the erosion of democratic norms, and heterogeneous perceptions of terrorism. More optimistically, other scholars have spoken of an increasing role for the EU in counter-terrorism despite some challenges. Ultimately, this book shows the EU as having now acquired a significantly enhanced role compared to the situation in 2001, whilst fully acknowledging some of the problems having affected EU counter-terrorism action, many of which still face the EU. It is important to keep in mind that the EU is a supranational organisation, not a state. This means that specific competences need to be granted to the EU, as it does not automatically possess them in the same way as a state. As Monar (2007a: 293) has shown:

There is no other example in the world of a group of countries agreeing on a comprehensive common strategy and action plan similar to that of the EU. As security – both in its internal and external dimension – remains arguably the area in which the European integration process has made least progress, this must be regarded as a major achievement in itself.

Thus, although various criticisms can be levelled at the EU in its efforts to combat terrorism, it is crucial to always remain clear on what it is and what it can do. The next section turns to the external dimension of EU counter-terrorism

The External Dimension of EU Counter-Terrorism

If internal measures to combat terrorism may be controversial from a sovereignty perspective, this may be even more so when it comes to the external dimension of EU counter-terrorism cooperation. States protect their ability to pursue independent foreign policies and external action in general perhaps

more than anything else, especially the large member states with the broadest range of global concerns. As previously noted, Keohane (2008) wrote one of the first articles of note on the subject of EU foreign policy counter-terrorism, in which he characterised the EU as an 'absent friend'. He largely viewed the EU as playing a minimal role in combating terrorism abroad. There were limitations to this contribution, however. In particular, he adopted a restricted conception of foreign policy counter-terrorism, focusing on areas where the EU – and any other actor assessed on the basis of such criteria – could only be seen as weak. Looking at EU external counter-terrorism in this way could only lead to the conclusion that action is often minimal. Furthermore, Keohane's contribution was at odds with a developing strand of the EU literature on the external dimension of JHA, for which EU counter-terrorism in North Africa was one among several considerations (Wolff, 2009; Wolff et al., 2009). The gradual identification and development of an external dimension to what were initially considered and intended to be internal policies of the EU (namely, JHA) have demonstrated the existence of an internal-external nexus in security policies (Weiss and Dalferth, 2009). In addition to the former Common Foreign and Security Policy (CFSP) (second) and JHA (third) pillars, certain aspects of EU counter-terrorism took place in the former first pillar (the European Community), such as combating terrorist financing (Bures, 2010b). Cross-pillar challenges posed significant problems to the EU. Not only this, but scholars began to look at individual relationships between the EU and the USA (Argomaniz, 2009b; Kaunert, 2010a; Occhipinti, 2010), the EU and North Africa (Eder, 2011; Joffe, 2008; Kaunert and Léonard, 2011a; Wolff, 2009), as well as the EU as a counter-terrorism actor more generally (Beyer, 2008; Brattberg and Rhinard, 2012; MacKenzie, 2010; Monar, 2015). In the end, these contributions have shown how the EU's 'fight against terrorism' has been considerably broader than early literature suggested and that the EU has developed as a significant player in global counter-terrorism efforts. Nevertheless, what has remained largely missing has been a joined-up assessment to evaluate the EU as a global counter-terrorism actor overall.

Over time, the literature on the external dimension of EU counter-terrorism has grown. Interest in the external dimension of EU counter-terrorism has increased to the point where a special issue on the subject was published in 2012 in the *European Security* journal (Ferreira-Pereira and Oliveira Martins, 2012; Kaunert, 2012; Oliveira Martins and Ferreira-Pereira, 2012). This has highlighted not only the increasing interest in the external dimension of EU counter-terrorism, but also, more importantly, the growing scope and geographical range of the EU's activities. Monar (2015: 333) has followed this up with an article on the external dimension of EU counter-terrorism, stating that: '[d]uring the first decade after the 9/11 attacks, the European Union (EU) has developed into an international counter-terrorism actor in its own right,

a role increasingly accepted by third countries [...] At the same time, the EU's counter-terrorism role has remained subsidiary, both legally and politically, to that of its [member states].' He has offered important evidence showing 'that the use of external relations instruments, such as political dialogues, counter-terrorism clauses, capacity building, economic assistance and others, has been a substantial part of the EU's response to the post-9/11 terrorist challenges' (Argomaniz et al., 2015: 198-9), whilst also arguing that the EU has increasingly become accepted by third states as a counter-terrorism player in its own right. Davis Cross (2017: 609) has also shown how the EU has made 'significant strides' in building up information-sharing, diplomacy, and the internal-external nexus and how the terrorist attacks of 2015-16 were 'critical junctures' that drove these developments. This book acknowledges that the idea that terrorist attacks have acted as critical junctures that have driven EU counter-terrorism has generally been convincing. Nevertheless, it also shows that such progress was built on the back of previous historical developments. Moreover, it demonstrates that, despite a number of challenges, the EU is now of much greater significance in counter-terrorism than it was in the early 2010s.

Recent work on the external dimension of EU counter-terrorism includes general approaches to the EU as an actor in counter-terrorism, but there have also been more specific studies carried out, focusing on the EU's relationship with the USA and North African countries, among others. Beyer (2008) and Brattberg and Rhinard (2012) have examined the 'actorness' of the EU in the case of counter-terrorism and have both come to similar conclusions – that the EU has been overall a modest actor in external counter-terrorism with potential for becoming more important in future. While this may be the case, it is worth examining these arguments and the EU's actorness in greater depth. These general studies necessarily gloss over much empirical data because the EU plays a variety of roles, which depend on the area of the world in question, the extent of the threat of terrorism in each region, the construction of the threat in each area, the receptivity of the interlocutor, as well as the will of the EU and of its member states. Terrorist threats have also changed and moved over time. Some places may have been important locations for terrorist activity for a time, before decreasing in importance when considering terrorism globally. Mali might be seen as such an example. Also, while the EU has cooperated on a range of issues with the USA, it has not done so with Japan, for example, because international terrorism has been overall less of a concern in Japanese foreign policy than in US foreign policy. In addition, the actions of the EU have significantly been governed by the types of cooperation that the EU has been able to offer, how strongly other states have wanted to cooperate with the EU, and whether such states have retained links with the former colonial powers (e.g. France and North Africa). Therefore, it is important to examine

EU actorness in different parts of the world in order to comprehensively assess it.

In terms of relationships, the EU–US relationship has been the most studied (Bossong, 2008a). It has been explored by various scholars, but most work has focused on EU-US agreements from different angles. In that respect, the PNR and the Society for Worldwide Interbank Financial Telecommunication (SWIFT) Agreements have been closely scrutinised (Argomaniz, 2009b; Kaunert, 2010a, 2012; Monar, 2010; Occhipinti, 2010; Pawlak, 2009a; Ripoll Servent and MacKenzie, 2011, 2012). These agreements have been significant because they have formed the basis for the EU-US relationship. In addition, although agreements could tackle a number of security challenges, cooperation has emanated from the initial intention to combat terrorism following 9/11, mostly stemming from the strong willingness of the US administration to protect its borders from terrorist infiltration after 9/11. The EU has often been seen as a 'norm-taker' and a re-producer of US security policies (Argomaniz, 2009a; Ripoll Servent and MacKenzie, 2012). To an extent, the relationship has been mutually beneficial. The USA has received the information and apparent security that it has aimed for, whilst the EU has gained recognition from the most powerful actor on the world stage and the member states have received analysed data from the USA. However, the relationship has been asymmetrical by favouring the USA. The EU-US relationship has become very significant because of the range of agreements that exist between the partners, from trade security in the form of the Container Security Initiative (CSI) to combating terrorism financing through the SWIFT Agreement. The relationship has been underpinned by a comprehensive list of agreements that are not only relevant to counter-terrorism, but can also be considered to combat organised crime. However, it is noteworthy that it is often member states or private companies that are required to meet the requirements of the agreements.

Furthermore, some literature has also considered the EU's counter-terrorism relations with North Africa. Criticisms have ranged from the suggestion that the EU has prioritised its own security concerns in its relations with North Africa to the argument that this has not been the case as EU counter-terrorism cooperation with North African countries has remained modest at best. In line with the latter argument, some have pointed out that European relations with North Africa have largely remained in the hands of the member states, especially those with a historical legacy and particular interests in the region, such as France and Italy (Eder, 2011; Joffe, 2008; Kaunert and Léonard, 2011a; Wolff, 2009). While the EU may have rhetorically emphasised the importance of security concerns in its relations with North Africa, this has not been supported in practice. There has only been a limited level of cooperation between the Union and North African countries, especially because Morocco and Algeria have preferred to cooperate on counter-terrorism issues with

member states rather than the EU institutions. For this reason, North Africa has not been selected as one of the case studies investigated in detail in this book.

Another region of the world in which the EU has been active in the field of counter-terrorism has been South Asia, in particular Pakistan and Afghanistan. There has been very limited literature on this aspect of EU counter-terrorism (see Monar, 2015; Wennerholm et al., 2010). This is rather surprising given the scale of the EU's commitment in terms of monetary and personnel resources, as well as the importance of the region in the US 'war on terror'. Much literature - as well as the media - has focused on the military campaign in Afghanistan and terrorist attacks in Pakistan. Counter-terrorism, however, is much more than military operations. This is precisely the difficulty; counter-terrorism is not a well-defined policy area and a range of actions can be considered relevant to counter-terrorism, from education through countering radicalisation to economic measures and financial sanctions against suspected terrorists (Keohane, 2005: 2-3). Because there have been many deadly terrorist attacks in Afghanistan and Pakistan, activities by the EU in these states have been mainly designed to assist in their ability to manage their own security in the face of the terrorist threat, whether in the long or short term. These activities have included capacity-building. Some scholarly literature has recently focused on the EU police mission in Afghanistan (EUPOL Afghanistan) (Larive, 2012; Moore, 2014; Oliveira Martins and Ferreira-Pereira, 2012). This reflects an emphasis on Common Security and Defence Policy (CSDP) missions, but does not fully take into account other less visible activities and contributions from other parts of the EU. Feeding into this is the fact that the EU is partially hamstrung in its ability to combat terrorism by its lack of access to certain instruments. For instance, it has little access to hard power. The EU is not a national government, it has a limited number of resources when compared to many of its member states, and the challenge of terrorism requires a response in most policy areas – a comprehensive approach (Keohane, 2005: 2–3). On the basis of the range of cases examined, this book challenges scholars to look beyond the near abroad or the USA in order to truly appreciate the scope of the EU's counter-terrorism activities.

In addition to these regions of the world, it has been suggested for some time that the EU would negotiate and sign PNR agreements with several other states, including Brazil, Russia, and Mexico (Association of European Airlines, 2014). PNR agreements are already in place with the USA and Australia, whilst an EU–Canada PNR Agreement was also signed before the Court of Justice of the EU decided that it could not be concluded in its current form in 2017. Furthermore, the EU has been active in the Sahel region of Africa, notably on the basis of a 'Strategy for Security and Development in the Sahel', which has been a high-priority concern given the activities of al-Qaeda in the Islamic Maghreb (aQIM) and the recent civil war in Mali. Moreover,

the EU has taken internal and external action in response to Daesh. Indeed, most of the policy developments mentioned by Davis Cross (2017) and the critical junctures that she has identified have been related to attacks organised or inspired by Daesh. Also, other scholars have argued that certain policy measures have become acceptable after attacks by Daesh (Ripoll Servent and MacKenzie, 2017). However, it is peculiar that this has not been a more prominent feature of the extant literature given the level of concern about Daesh among European states. Likewise, EU action related to counter-terrorism in Syria and Iraq has gone largely unnoticed.

Thus, one can question whether the global counter-terrorism role of the EU has been fully established and assessed until now. First of all, there has not been any geographically wide-ranging and comprehensive assessment of the EU's global role in counter-terrorism. The EU's counter-terrorism activities in some parts of the world have hitherto remained largely under-researched. Furthermore, overall, there has been only limited reflection on the different roles of the EU in counter-terrorism and on the range of functions that it can now perform. The EU is a more significant actor than it was 20 years ago. In that respect, the possible end of Daesh and of the fourth wave of terrorism offers a timely opportunity to reflect on how far the EU has come in the development of its counter-terrorism role. This book therefore contributes to an understanding of the importance of the EU on the contemporary world stage through an analysis of its reaction to one of the main global challenges that it faces – terrorism.

THE ARGUMENT IN BRIEF

This book contributes to two strands of the academic literature, namely that on EU counter-terrorism and that on EU integration and actorness. As outlined above, the threat of international terrorism in the aftermath of 9/11 acted as a catalyst for European integration internally and externally, including the development of EU cooperation with a growing number of third countries. First, the EU collectively 'securitized' this transnational threat, which this book analyses through an application of the 'collective securitization' model originally developed by Sperling and Webber (2019). Security, in this conceptualization, is about 'the processes of constructing a shared understanding of what is to be considered and collectively responded to as a threat' (Buzan et al., 1998: 26). It is a process for a given actor to frame a specific issue as an 'existential threat', presented to a target audience for approval in order to employ extraordinary means and measures to tackle it (Léonard and Kaunert, 2011: 57). Haacke and Williams coined the concept of 'collective securitization', which was defined as 'securitization within a regional arrangement as involving one or more securitizing actors within that arrangement identifying a particular development or issue as an existential threat to a security referent, making relevant validity claims, and finding a receptive audience among other regional actors' (Haacke and Williams, 2008: 785). Sperling and Webber subsequently argued that a regional security organisation could also constitute an agent of collective securitization, especially 'when an international organisation is possessed of legal and political authority, has agenda-setting powers, is the framework for formulating and implementing common policies, and is the repository of a common security narrative' (Sperling and Webber, 2016: 29). This book adapts the collective securitization framework originally developed by Sperling and Webber (2019) and applies it in Chapter 2, analysing the collective securitization of terrorism in the EU in the wake of 9/11. Second, this book applies the concept of 'actorness' in order to analyse the evolution of the EU as a counter-terrorism actor in different case studies. Overall, this book highlights that the EU has become a counter-terrorism actor of growing importance, which possesses an ever-diversifying number of policy options available. For this purpose, the book applies an adapted version of the set of actorness criteria identified by Brattberg and Rhinard (2012), namely context, coherence, and capability. The justification for choosing these specific criteria and their operationalisation are thoroughly discussed in the conceptual chapter. It suffices to say here that it is believed that the framework offers a tried and tested method for examining the EU as a global counter-terrorism actor, while also being flexible enough to manage an examination of each of the selected cases. The book focuses on the variety of roles that the EU has played in different parts of the world. It suggests that the extant literature has not offered a joined-up approach to EU counter-terrorism and therefore has not been able to offer a truly comprehensive and thorough assessment of the EU's actions in global counter-terrorism. This framework and the case studies allow for such an evaluation.

With regard to its empirical contributions, the book shows how the EU has been active in a number of parts of the world that are rarely written about from the perspective of the Union. These include Afghanistan, Pakistan, and the Sahel region of Africa – areas of the world that are not proximate to the EU, which is interesting because EU scholars often see the Union as primarily active in nearby states (with the important exception of trade, where the EU is one of the key actors on the global stage). After presenting a significant amount of new empirical material, this book is able to offer a comprehensive evaluation of the EU's counter-terrorism role with relevance more broadly to the literature on security and that on global politics. On the conceptual front, this book applies the concept of actorness to counter-terrorism. While this has been applied to counter-terrorism in a narrower sense and other policy areas in the past (Beyer, 2008; Brattberg and Rhinard, 2012; Kaunert, 2010a, 2010b; Kaunert et al., 2015), it has not been used to analyse EU action in

various states around the world as done here. The remainder of this book offers a detailed, thorough, and comprehensive analysis of the EU's global role in counter-terrorism. As a result, several arguments are developed. First, the EU has become a significant counter-terrorism actor in its own right, even if its abilities have been more circumscribed than those of its constituent member states and it has kept a largely supporting role to them. Second, terrorism has played a major role in pushing European integration forward into the realm of high politics and external relations. Third, internal and external players have been influential in constructing the EU's approach to counter-terrorism. Finally, a framework centred on actorness can help capture not only whether the EU is an actor in counter-terrorism, but also to what extent, in which ways and how its actorness may differ in various parts of the world. Thus, it allows for capturing the multifaceted and geographical range of the EU's role in counter-terrorism.

This book brings together a range of different projects on which the authors have worked in order to comprehensively assess the EU as a global counter-terrorism actor. It has been built through an engagement with various strands of literature, documentary evidence spanning several decades, and over thirty interviews that took place in the EU institutions, several member state representations, and member state ministries between 2008 and 2019. The interviews were conducted separately by each of the three authors and were semi-structured. Most participants agreed to take part only if they were not to be identified by name. Therefore, this book uses an appropriate designation, such as 'Commission official A' or 'MEP A' to reference them. This project would not have been possible without these interviewees, to whom the authors express their gratitude for being so generous with their time and insights.

The various case studies were chosen in order to allow for an examination of the full range of EU counter-terrorism activity. The cases are therefore the main actors and regions involved in the 'war on terror'. As a result, the book considers the EU–US relationship because the USA is the EU's closest partner, the only superpower, and the state that led the 'war on terror'. It also analyses the EU's role in South Asia, more precisely in Afghanistan and Pakistan. This has been largely overlooked in the extant literature, which is somewhat intriguing given that this region was the starting point and one of the main theatres of the 'war on terror'. Finally, the book examines the EU's counter-terrorism activities in the Middle East, in particular Syria and Iraq. It is of course impossible to ignore Daesh, which may have been synonymous with the culmination of the fourth of the waves of terrorism identified by Rapoport (2001, 2012).

STRUCTURE OF THE BOOK

This book is concerned with evaluating the global role of the EU in combating terrorism around the world. It mainly aims to examine the expansion of the EU's efforts to combat terrorism since 2001, especially in the main regions of the planet where the EU has been active, which have tended to be characterised by significant terrorist activity. It seeks to explore the reasons behind this growth and to account for the differing levels of activity in different parts of the world. For these purposes, the book is located within the broader scholarly debates about the significance of the EU as an actor in global politics and security.

Chapter 1 explains how the analysis of the role of the EU as a global counter-terrorism actor in this book proceeds in two main stages. The first aims to analyse how and why the EU became involved in the area of counter-terrorism, which had long been dominated by the individual member states. It does so using a five-stage 'collective securitization' framework adapted from Sperling and Webber (2019). The second stage of the analysis presented in this book aims to establish and to assess the role of the EU as a counter-terrorism actor on the global stage. For this purpose, an 'actorness' analytical framework is applied to three cases, which involve the USA, South Asia, and Iraq and Syria. The remainder of this chapter introduces the two analytical frameworks subsequently used and locates them in the context of the relevant scholarly debates, notably that on European integration.

Chapter 2 explores how terrorism has come to be perceived as a European security threat and how this change has led to the development of EU counter-terrorism cooperation. It analyses the collective construction of terrorism as a security threat within the EU using a five-stage 'collective securitization' framework. In so doing, the chapter establishes the beginnings of EU action in counter-terrorism.

Chapter 3 analyses the role of the EU institutions, namely the European Commission, the European Parliament, the Council Secretariat and the EU CTC, as well as the European External Action Service. It shows that EU institutional actors have played a crucial role in shaping the development of EU counter-terrorism by pushing for the EU as the best vehicle for countering terrorism. Seizing on 9/11, they developed 'European' rather than 'national' solutions, which is suggestive of increased supranational security governance in EU counter-terrorism. Institutional change followed exogenous shocks like 9/11, as well as the terrorist attacks on Madrid and London, and was facilitated by policy entrepreneurship.

Chapter 4 examines the EU's most important counter-terrorism relationship, namely that with the USA. This relationship is of historical importance given

the role played by the USA in the process of European integration and its willingness to cooperate directly with the EU rather than its constituent units. The transatlantic relationship, being the most developed, can also act as a vardstick for assessing the importance of cooperation with other states. Concern about American actions to combat terrorism rose in Europe after 9/11, especially with regard to calling it a 'war on terror'. George W. Bush's State of the Union address in 2002, in which he identified several states as belonging to an 'axis of evil', was coolly received in Europe. Transatlantic relations arguably reached their lowest point in 2002–03 in the run-up to the US-led invasion of Iraq, but ties were subsequently repaired. Despite the rhetoric, the mid-2000s was actually a very productive time in transatlantic relations, which improved even further under Barack Obama. More recently, under the Trump administration, the USA became a more unpredictable actor. President Trump made no secret of his hostility towards the EU and notably championed Brexit. It is not clear how much EU-US cooperation will grow during Joe Biden's presidency and beyond, as there may not be much impetus for significant development in the years to come.

Chapter 5 examines the EU's role in South Asia, particularly Afghanistan and Pakistan, as the starting point of the 'war on terror'. Afghanistan has received significant attention in the twenty-first century after the US-led invasion of the country. However, the country has proven much more difficult to pacify than expected, whilst the Taliban and other insurgent groups have remained active and have contributed to the destabilisation of north-west Pakistan. In all this, the EU's role in Afghanistan has gone virtually unnoticed, despite sending vast sums of money to the region, amounting to several billion euros over the past decade. This chapter considers the EU's role in Afghanistan and Pakistan. Such an assessment is crucial because it shows how the EU has developed as a global counter-terrorism actor and what it has achieved.

Chapter 6 examines the development of the EU's counter-terrorism activities in relation to Iraq and Syria, with a specific focus on how it has sought to address the threat of Daesh. The EU began to play a role in the region during the US occupation, which lasted until 2011. While it may have been expected that the EU would play only a limited role, EU action has actually grown over the past few years and will likely remain important in the reconstruction of the region. Furthermore, European states have assisted the USA and other military forces with airstrikes and Special Forces. Thus, this chapter allows for an examination of the role of the EU at its most developed point. The book will have followed the whole arc of the EU's development as a counter-terrorism actor if the defeat of Daesh in Syria and Iraq actually marks the end of the fourth wave of terrorism. Having said this, the end of fighting is far from meaning the end of terrorism, especially because of the very serious sectarian divides in both states.

The concluding chapter sums up this comprehensive examination of the EU as a global counter-terrorism actor. It reflects on the development of the EU as a global counter-terrorism actor and looks ahead in order to consider what the future may hold.

1. EU counter-terrorism, collective securitization, and global actorness

Terrorism is one of the main security threats in Europe today. Several European states have been victims of terrorism in the past and many remain possible targets nowadays. Between them, European countries have diverse experiences of terrorism, ranging from left-wing and right-wing terrorism in Germany and Italy, respectively, through British and Spanish experiences of separatist terrorism and the forms of Islamist terrorism encountered by France in the 1990s, to Islamist terrorism today. The latter is generally considered to be the greatest terrorist threat currently facing many European countries as highlighted by the UK's 'CONTEST' strategy and the EU's Counter-Terrorism Strategy (Council of the European Union, 2005a: HM Government, 2011). Many lessons can be taken from past experiences of terrorism to help combat current threats (Duyvesteyn, 2004; Spence, 2007b). Although the jihadist terrorist threat is generally considered to be the most significant of all terrorist threats in Europe at this time, other terrorist threats – be they international or domestic – cannot be neglected (Europol, 2021). For example, the case of Anders Behring Breivik in Norway has raised fears of a form of terrorism based upon right-wing militancy and Islamophobia (Guardian, 2011). This could be a one-off case, a precursor to further attacks, or an inspiration to others who hold similar beliefs, but it does serve as evidence that Islamist terrorism is not the only – nor always the most pressing – threat to European countries.

Until 9/11, the EU was largely excluded from counter-terrorism activities, which remained almost the sole preserve of its member states. However, the attacks on the USA proved to be a turning point for the EU as it reacted with impressive speed and decisiveness, including emergency meetings between the member states and the EU institutions, opening itself up to counter-terrorism cooperation with the USA, the adoption of a plan of action for combating terrorism outlining a number of areas where the EU could act and a 'month of transformation' where the EU prioritised efforts to combat terrorism (Bossong, 2008; Zimmermann, 2006). Hence, international terrorism has acted as a major catalyst for the development of measures agreed before 9/11 and has transformed the governance of the EU (Edwards and Meyer, 2008a; Zimmermann, 2006). Counter-terrorism has increasingly become a policy issue for the EU, whilst the Union has come to be recognised as a significant counter-terrorism

actor by others. This has been evidenced by the growing number of countries that have expressed an interest in cooperating directly with the EU, starting with the USA (Kaunert, 2010c). The list of countries now directly conducting counter-terrorism cooperation with the EU is remarkable, whilst such cooperation now encompasses a growing number of policy areas. Nowadays, the EU is active in the field of counter-terrorism across most continents and in many regions threatened by terrorism. This highlights the extent to which the EU has prioritised terrorism and has acted to combat it after 9/11. Without this major event, internal and external counter-terrorism action at the EU level would not have progressed so significantly.

Consequently, the EU is of much greater significance as a counter-terrorism actor nowadays than it was in 2001. This evolution is of great interest and importance (MacKenzie, 2010), notably in its external dimension. However, the external aspects of EU counter-terrorism cooperation have not received the attention that they deserve in the current scholarship. This is surprising for two main reasons: first, because the threat of terrorism to Europe remains significant; and, second, because the EU now cooperates with a large number of third countries and performs a range of counter-terrorism functions. Overall, the number of EU projects relevant to counter-terrorism has significantly grown since 9/11. This means that the EU has now fully entered the arena of 'high politics', which comprises matters of security and foreign policy (Kaunert, 2007). The fact that the EU is now involved in the previously unreachable areas of security policy and foreign policy highlights the major impact of international terrorism on the progress of European integration.

The existing literature on EU counter-terrorism has overwhelmingly focused on the internal dimension of EU counter-terrorism and how the necessity to address the terrorist threat has also fostered cooperation on a range of related issues, such as aviation security (Argomaniz, 2009a, 2010, 2011; Bossong, 2008a, 2013; Bures, 2006, 2011; Kaunert, 2007; Spence, 2007b; Brown, 2010; De Londras and Doody, 2017; Kaunert et al., 2021). This is perhaps unsurprising given the major strides forward that the Union has made internally, including the adoption of the EAW, amongst other measures. However, these internal developments have also led to external developments. The EU is now able to negotiate and sign agreements on behalf of its member states and has attempted to export its standards to partner countries through a process of cooperation and norm transfer. As these developments have occurred, there has been a slow recognition of the EU's external counter-terrorism role. For instance, the PNR and SWIFT Agreements between the EU and the USA have received significant attention, both in academia and in the media (Argomaniz, 2009b; Brouwer, 2009; Guild and Brouwer, 2006; Pawlak, 2009a, 2009b). This is because of the great importance attached to data protection in Europe and because the EU-US relationship has evidently been of great importance

to both sides of the Atlantic. Even so, the PNR and SWIFT Agreements specifically, and the EU–US relationship more generally, have only been small elements of the EU's efforts to combat terrorism, which have stretched far beyond cooperation with North America. A significant amount of EU activity has actually gone largely unnoticed.

This book therefore examines the role of the EU as a global counter-terrorism actor. This analysis proceeds in two main stages. The first aims to analyse how and why the EU became involved in the area of counter-terrorism, which had long been dominated by the individual member states. It does so using a five-stage 'collective securitization' framework adapted from Sperling and Webber (2019). Chapter 2, which applies this framework, highlights how 9/11 led to the redefinition of terrorism away from an individual (in the sense of national) security threat to a collective, European security threat. The second stage of the analysis presented in this book aims to establish and to assess the role of the EU as a counter-terrorism actor on the global stage. For this purpose, an 'actorness' analytical framework is applied to three cases, which involve the USA (Chapter 4), South Asia (Chapter 5), and Iraq and Syria (Chapter 6), respectively. The remainder of this chapter introduces the two analytical frameworks subsequently used and locates them in the context of the relevant scholarly debates, notably that on European integration.

UNDERSTANDING THE INVOLVEMENT OF THE EU IN COUNTER-TERRORISM: COLLECTIVE SECURITIZATION

As previously mentioned, before being able to map out the scope and the contents of the EU's activities in the external dimension of counter-terrorism, it is important to establish *why* the EU has become involved in counter-terrorism. This book does so using a 'collective securitization' framework based on previous work by Sperling and Webber (2019). However, prior to introducing this framework, it is necessary to first discuss the concept of 'securitization' on which 'collective securitization' builds.

Securitization

The concept of 'securitization' was initially developed by Ole Wæver to make a major contribution to the so-called 'widening-deepening' debate in security studies, which had begun in the 1980s and intensified with the end of the Cold War. The 'widening' dimension was defined as the extension of security to issues or sectors other than the military, such as the environment or the economy, whereas the 'deepening' dimension addressed the question of whether entities other than the state, such as society or individual human

beings, should be able to claim security threats (Krause and Williams, 1996: 230). Together with the concept of 'security sectors' previously developed by Buzan (1991), 'securitization' is at the heart of a new theoretical framework that, according to Wæver and Buzan, enables researchers to simultaneously widen and deepen the concept of security without rendering it too broad or meaningless. The key idea underpinning the securitization framework is that security is not about objective threats that 'really' exist out there. Rather, still for Wæver and Buzan, it is about 'the processes of constructing a shared understanding of what is to be considered and collectively responded to as a threat' (Buzan et al., 1998: 26). More precisely,

[security] is about survival. It is when an issue is presented as posing an existential threat to a designated referent object (traditionally, but not necessarily, the state, incorporating government, territory, and society). The special nature of security threats justifies the use of extraordinary measures to handle them. (Buzan et al., 1998: 21)

In other words, according to Buzan and Wæver (also known as the 'Copenhagen school'), security is a 'speech act' (Buzan et al., 1998: 26) (see also Wæver, 1995: 54-5; Roe, 2008: 617; Stritzel, 2007: 358; Balzacq, 2005: 174-9). It is an intersubjective and socially constructed phenomenon. Key concepts in the securitization framework are the 'securitizing actor', who socially constructs a specific issue as a threat to the survival of a given entity, known as the 'referent object', which therefore requires urgent protection through the use of extraordinary measures. Another important concept is that of the 'audience'. According to the Copenhagen school, '[a] discourse that takes the form of presenting something as an existential threat to a referent object does not by itself create securitization – this is a securitizing move, but the issue is securitized only if and when the audience accepts it as such' (Buzan et al., 1998: 25). To sum up, securitization is understood as a process whereby a given actor frames a specific issue as an 'existential threat', which is then presented to a target audience for approval in order to employ extraordinary means and measures to tackle it (Léonard and Kaunert, 2011: 57).

In its original formulation by the Copenhagen school and in subsequent studies by other scholars, securitization theory has overwhelmingly been applied to states, as well as nations to a lesser extent. According to Buzan and Wæver (2009: 255), 'the middle-scale "limited collectivities" have proved the most amenable to securitization as durable referent objects [because] such limited collectivities (states, nations, and as anticipated by Huntington, civilisations) engage in self-reinforcing rivalries with other limited collectivities and that such interaction strengthens their we-feeling'. This in turn facilitates securitization. In contrast, collectivities at the system level lack the mass

identity that is necessary for securitization to take place (Buzan and Wæver, 2009: 255). In other words, the level of analysis in the study of securitization processes has generally been the middle level of world politics, with a specific focus on states. This is not to say that the work of the Copenhagen school has exclusively focused on states. Their other concept of 'security constellation' '[links] across all of the levels and sectors in which securitizations occur', whilst that of 'macrosecuritization' concerns 'referent objects higher than those at the middle level' and '[aims] to incorporate and coordinate multiple lower level securitizations' (Buzan and Wæver, 2009: 257). In addition, some of Wæver and Buzan's writings also went beyond the national level to consider regions. In particular, they have developed Regional Security Complex Theory, which has centred on the concept of 'regional security complex'. This concept refers to 'a set of units whose major processes of securitization and de-securitization or both are so interlinked that their security problems cannot reasonably be analyzed or resolved apart from one another' (Buzan et al., 1998: 201; Buzan and Wæver, 2003).

Towards Collective Securitization

Overall, Buzan's and Wæver's collective work has remained focused on states, their patterns of amity and enmity, as well as the distribution of power amongst them and the role of global powers (Buzan et al., 1998; Buzan and Wæver, 2003, 2009). In particular, they did not consider how securitization processes may take place within regional arrangements. The first scholars to address this gap in the literature were Haacke and Williams (2008). They coined the concept of 'collective securitization', which they defined as 'securitization within a regional arrangement as involving one or more securitizing actors within that arrangement identifying a particular development or issue as an existential threat to a security referent, making relevant validity claims, and finding a receptive audience among other regional actors' (Haacke and Williams, 2008: 785). They also noted that one could expect securitizing moves within a regional arrangement to entail claims that a specific development 'constituted a threat either to regional security or to the respective national security of participants, and required a collective response' (Haacke and Williams, 2008: 785). Nevertheless, as highlighted by Sperling and Webber (2016: 29), Haacke and Williams's approach '[assumed] that a state [would] initiate a [securitizing] move that [would] then be generalised within a regional arrangement or organisation'. They did not consider cases where a regional organisation itself could initiate a securitizing move. This is to a significant extent linked to their case selection, as they focused on the African Union (AU) and the Association of Southeast Asian Nations (ASEAN). Those may indeed be considered 'the primary multilateral arrangements in their respective region' (Haacke and Williams, 2008: 777). However, compared to a regional organization like the EU, the AU and ASEAN have been characterized by a considerably lower degree of integration when it comes to the institutional set-up, the decision-making processes, the degree of legal integration, and the extent of political integration, amongst others. In contrast, Sperling and Webber have argued that the role of regional security organizations should not necessarily be reduced to that of a site for bargaining amongst their member states. In their view, a regional security organization can also be an agent of collective securitization, especially 'when an international organisation is possessed of legal and political authority, has agenda-setting powers, is the framework for formulating and implementing common policies, and is the repository of a common security narrative' (Sperling and Webber, 2016: 29).

In order to study this process of collective securitization, Sperling and Webber (2019) have outlined a six-stage model, which comprises (1) the status quo security discourse and policies; (2) a single precipitating event or a cascade of events; (3) the securitizing move; (4) the response of the audience; (5) the formulation and execution of policies to address the securitized threat; and (6) routinization and the emergence of a new status quo. Although it may be analytically possible to distinguish the securitizing move from the audience response, these two stages are actually 'co-dependent through [a] process of recursive interaction' (Sperling and Webber, 2019: 232). These two stages are therefore combined into one single stage in the collective securitization framework applied in this book, which therefore comprises five stages.

ASSESSING THE ROLE OF THE EU IN GLOBAL COUNTER-TERRORISM: EU ACTORNESS

Having introduced the framework that will help explain how the EU became involved in counter-terrorism, this section now turns to the 'actorness' framework that will be applied in order to establish and to assess the role of the EU in global counter-terrorism.

The Emergence of the Concept of 'Actorness'

According to Sjöstedt (cited in Bretherton and Vogler, 2006: 17), actorness or 'actor capacity' is the 'capacity to behave actively and deliberately in relation to other actors in the international system'. It can be assessed using specific criteria. Different scholars have elaborated distinct sets of criteria (see Drieskens, 2017). It has previously been acknowledged that the EU is not a traditional actor in the international system (i.e. a state) or an international organization, but rather a *sui generis* actor (Bretherton and Vogler, 2006; Carlsnaes, 2006; Ginsberg, 2001; White, 2006). Traditional state-centric approaches to

international relations, such as Realism, have often ignored what is distinctive and interesting about the EU (Bretherton and Vogler, 2006: 12). This has led to the somewhat misleading conclusion that the EU either is not an actor or is a weak actor on the global stage because state-centric theories prioritise state interactions and primarily military action over other types of activities (Bretherton and Vogler, 2006: 12). Although military power will always be an important factor on the international stage, it is not the only, or always the most effective, type of international action. Nevertheless, it is true that the EU has been limited – to some extent at least – by its lack of military capability. Furthermore, relations between actors in non-military spheres are also of great importance. Dissatisfied with state-centric approaches because of their analytical focus, scholars of the EU started experimenting with a number of approaches that deviated from those of the past and came to different conclusions, with the EU being seen as actually an important actor on the world stage in several contexts and policy areas (Bretherton and Vogler, 2006; Ginsberg, 1999, 2001; Smith, 2002). The overall difficulty of conceptualizing the EU has had significant implications for researchers because the theories chosen shape academic research. In line with this, research on the external dimension of EU counter-terrorism has generally not conceptualized the EU as an actor, which has led to the arguably premature conclusion that the EU is not important in the area of counter-terrorism (Keohane, 2008). As a result, EU actorness in counter-terrorism has remained largely under-researched.

This book applies the concept of actorness in order to gauge the extent of the EU's role and to capture the variety of its activities in global counter-terrorism. Not being a traditional type of actor (i.e. a state), there have always been challenges when conceptualizing and analyzing the EU and its role in the world. Traditional theories of international relations (e.g. realist theories) have often privileged states over other types of actors, but the very fact that the EU has external effects and conducts a range of external relations means that it cannot be ignored, despite the fact that it is not a state. Furthermore, there are many states in the world that are sovereign, but have less influence on global politics than the EU. As Dryburgh (2008: 255) has argued, the EU has often been compared to an idealized state. Although the EU's foreign policy apparatus is not as developed as that of some of its member states, there are some states whose actorness is not in question, but whose foreign policy apparatus is in some ways less significant than the EU's. She cites the example of Portugal, which in 2012 had only 52 embassies, compared to the EU's 128 delegations around the world (Dryburgh, 2008: 255; Keukeleire and Delreux, 2014: 123). In addition, traditional theories of international relations have privileged traditional foreign and security policy as a special activity of the state. As it is arguably where the EU is the least integrated, it makes the EU appear weaker on the international stage, overall.

It is important to bear in mind that the EU is more of an international actor in some areas, particularly in external trade, than in others (Ginsberg, 1999). When it comes to military matters, the EU appears to be a more modest actor. However, this lack of a military component does not make the EU or its foreign policy apparatus an international irrelevance. Bretherton and Vogler (2006: 9), when conducting interviews with representatives of EU interlocutors, found that '[w]hile a number of interviewees commented upon the lack of overall political direction and impetus to external policy, the absence of military capability was not identified as an issue by any of the third party representatives interviewed – all of whom nevertheless considered the EU to be a significant actor'. Much influence also derives from the EU's role as a major economic power, which can be used to pressure other actors. Gehring et al. (2017: 728) have even gone as far as stating that 'the EU's ample action capabilities based upon its communitarised policies can make it a great power in its own right'. This book does not necessarily subscribe to the perception of the EU as a great power, but it is underpinned by the belief that traditional perspectives have tended to underplay the role of the EU in the world. For this study, it is worth pointing out that EU actorness in counter-terrorism has generally been under-researched, particularly in certain parts of the world.

Thus, actorness is used here to establish, gauge, and highlight the variety and significance of the EU's counter-terrorism activities in each geographical region that this book examines. The concept of actorness emerged in the 1970s because groups of states, such as the EU, and transnational actors were beginning to become important in their own right. 9/11 itself has demonstrated the increasing harm that non-state entities with relatively limited resources are capable of causing. At the same time, traditional theories of international relations proved to be inadequate for analysing the role of non-state actors in global politics, such as transnational corporations, regional organizations, and international organizations. Thus, with the struggles over how to conceptualize the EU, actorness has been one of the few approaches that has allowed for an examination of the international role of the EU. It has helped establish the EU's role in world politics and its identification as an actor on the world stage. A number of contributions have convincingly demonstrated that the EU can – although it does not always do so – play an important role in world politics.

The dominant theory in international relations during much of the post-Second World War period was Realism and its many variants. Realism focuses on states (most importantly, great powers) and power as military power, in its classical, structural, and neoclassical variants. Historically, states have been by far the most important actors in international relations, but the twentieth century has seen a greater plurality of actors emerge and play an important role in global politics, including transnational corporations, international and regional organisations, as well as terrorist groups. In fact, al-Qaeda's attack

on the USA on 9/11 was arguably one of the most significant attacks ever by a non-state actor, and one that helped produce a fierce response by the USA. In contrast to Realists, Liberals attempted to put forward that states were not the only actors in world politics and that a pluralist system had emerged, allowing space for actors such as the EU to be discussed. Typically, Realists have overlooked the EU because of its non-state status. Waltz (1979: 180) viewed the EU as a potential third great power in the system (other than the USSR and the USA during the Cold War), but this was related to Europe uniting and roughly resembling a state, which he deemed unlikely. Because the EU does not have significant access to military power and is not a state, it has not been deemed important enough in these policy areas to be studied by Realists. Mearsheimer (1990: 45) even expected that European integration would break down at the end of the Cold War. In his assessment, European rivalry was muted by the threat of the USSR; with this superpower gone, suspicions amongst member states would be revived and lead to the end of European integration. This has not come to pass. The opposite has in fact occurred – European integration was relaunched towards the end of the Cold War and proceeded further following its conclusion, noticeably with the Treaties of Maastricht, Amsterdam, Nice, and Lisbon. In contrast, Joseph Grieco has attempted to tackle the European integration process head-on, admitting that 'the interest displayed by the European countries in the EU creates a problem for realist theory' (Andreatta, 2011; Grieco, 1997: 184). Others have brushed over the EU because it is an anomaly in a world where states, power, and suspicion are likely to prevail. To assume that the EU is unimportant simply because it is not a state and does not have real access to military power misses precisely the influence of the EU in the world. This book takes as its starting point that the world is characterized by a plurality of actors.

Thus, as previously mentioned, in the 1970s, interest grew in examining international actors other than states. Cosgrove and Twitchett (1970) looked at 'new actors', specifically the United Nations (UN) and the European Economic Community (EEC), as the EU was known at the time. Then, later in the 1970s, Sjöstedt (1977) discussed 'actor capacity', which he defined as the 'capacity to behave actively and deliberately in relation to other actors in the international system' (Sjostedt, cited in Bretherton and Vogler, 2006: 17). However, it was not until the 1990s that the concept of 'actorness' was added to the EU lexicon. In an important edited book, Jupille and Caporaso (1998) developed a set of criteria for actorness, including recognition, authority, autonomy, and cohesion. This was closely followed by another important contribution by Bretherton and Vogler (1999, 2006), who discussed three criteria for actorness: opportunity, presence, and capability. The difference between these two sets of criteria can be summarised as follows: while the conceptualization by Jupille and Caporaso was more formal and rationalistic, Bretherton

and Vogler's contribution offered a more constructivist slant on actorness (Brattberg and Rhinard, 2012: 559).

The concept of actorness has been examined and applied by a few scholars. often when analyzing international negotiations (Groenleer and van Schaik, 2007; Jupille and Caporaso, 1998). Dryburgh (2008) used Larsen's (2005: 59) criteria, which in turn were adapted from Bretherton and Vogler's work, to examine how the EU was an actor because of the CFSP, rather than despite it, in the case of the policy towards Iran. With regard to counter-terrorism, the EU has been examined as a global counter-terrorism actor by Beyer (2008) and Brattberg and Rhinard (2012). Actorness has also been applied to actors within the EU, such as Europol (Kaunert, 2010b) and the CTC (MacKenzie et al., 2013a). However, even in 2013, Niemann and Bretherton (2013: 262) argued that: '[w]hile several publications offer valuable conceptualisations of EU actorness, the literature contains relatively few systematic explorations of the actual extent of EU actorness'. Furthermore, Kaunert (2010a: 49) has pointed out that EU actorness in counter-terrorism has been under-researched. Although some scholars have considered the EU as a global counter-terrorism actor in general, they have not examined the various relationships that the EU has forged with a range of third countries in particular. Because the EU has performed various functions and has displayed different levels of actorness in these relationships, it is only by examining each relationship and aggregating the results that it becomes possible to gain thorough knowledge of the extent to which the EU has become a global counter-terrorism actor.

Even so, Bretherton and Vogler (2013: 375) have argued that 'that achievement of the increased capability envisaged by the Treaty of Lisbon, together with resolution of the Eurozone crisis, with its deleterious effect upon the Union's presence, would not fully compensate for the loss of opportunity provided by the changing international structure'. In short, they believe that internal problems, including the financial crisis that has engulfed Europe, and the external environment, with the rapid rise of new powers, means that the EU has declined as an actor in global politics. This argument is worth engaging with – while not disputing the central contention that the EU has declined as an actor (this is somewhat inevitable with the global recession and the rise of new powers), it is possible to examine whether the EU has become a more important global actor in one policy area: counter-terrorism.

EU Actorness in Global Counter-Terrorism

This book draws upon and adapts Brattberg and Rhinard's (2012) four criteria for actorness, which they identified as follows: context, capacity, coherence, and consistency. This specific set of criteria was chosen because it synthesises the criteria elaborated by other scholars for a comprehensive examination of

actorness and it was developed specifically with examining the EU's role in counter-terrorism in mind. Moreover, these criteria are flexible enough to take into account all actors that are examined in this book. Nevertheless, this book slightly adapts their set of criteria as it subsumes 'consistency' (which refers to the idea of 'sticking to the line' over time) under 'coherence', as the analysis of coherence is carried out over a certain period of time – about 20 years – in each of the empirical chapters.

As a result, the 'actorness' framework applied in this book comprises the three following criteria:

- Context: whether circumstances enable or constrain EU action and its recognition by partners;
- Coherence: the presence or absence of shared values, the ability to form and pursue preferences, established procedures, along both vertical and horizontal dimensions;
- Capability: the authority to act, available instruments, and the deployment of instruments in the state or region in question.

Context

'Context' relates to whether circumstances have enabled or constrained EU action – notably the extent to which an issue, such as counter-terrorism, has been securitized collectively. In this sense, the EU was evidently not much of a player in counter-terrorism prior to the attacks of 9/11. This was undoubtedly the singular most important event in thrusting the EU into this policy area. It led to the member states recognizing that greater cooperation amongst European states was needed to combat international terrorism, especially given the specific circumstances and opportunities – including the arrangements governing the Schengen zone - provided for criminals and terrorists in EU states. Simultaneously, actors within the EU have seen and used events such as 9/11, as well as subsequent attacks, to push forward their agendas (Argomaniz, 2009b; Davis Cross, 2017). Without 9/11, for example, the EAW may not have been adopted for many years because these terrorist attacks gave the European Commission a significant opportunity to play a leading role (Kaunert, 2007). Similarly, without the Madrid attacks of 2004 and the London attacks of 2005 and the British presidency that year, perhaps there would not have been any EU CTC or EU Counter-Terrorism Strategy, respectively. On the external front, the need to show solidarity with the USA and to build the EU into a partner for the USA to work with was important for actors within the EU (Bossong, 2008a). Also, the Bali bombings of 2002, which led to the death of a number of Europeans, but also escalating violence around the world, some of which could also be harmful to Europe and Europeans, motivated the EU to play a greater role in global counter-terrorism action. Continuing attacks and violence have

led to an evolving EU role, noticeably with the emergence and global activities of Daesh, which committed terrorist attacks in several European states. Thus, EU counter-terrorist action could be split up into two phases: concern about al-Qaeda following attacks between 2001 and 2013 and then the focus largely on Daesh from 2013 until now. This is not to say that other terrorist threats do not matter. Nevertheless, jihadist terrorism has done more than other types of terrorism to facilitate cooperation amongst EU member states.

The announcements of a 'war on terror' and of the existence of an 'axis of evil' were received coolly in the EU and perhaps contributed to the announcement of an alternative and lower key 'fight against terrorism'. In this way, the EU was putting forward an alternative model mainly based on law enforcement and tackling the root causes of terrorism. None of this would have been possible without 9/11 and the USA then proceeding to respond in the way it did. The US invasion of Afghanistan led to the EU playing a significant role in the country. Also, while the EU was initially not very active in Iraq after the US-led invasion and was divided because of the decision of some member states to support it, the EU eventually became involved, not least because of the rise of Daesh years later. The USA has set the path and the EU has responded to US activities over the past 20 years. Without such circumstances and the actions of others, EU action in this area would not have been possible.

Second, context also refers to external recognition by the partners cooperating with the EU. While an actor can still act without recognition, entities are probably unlikely to respond to others that are not relevant, do not grant their actions legitimacy or do not have useful capabilities. Greater recognition, especially by powerful actors, can enhance the reputation of actors and lead to a phenomenon akin to a snowball effect, where agreements with one state can lead to agreements with another once it is recognized that an actor plays a role in a given policy area. Thus, in practice, recognition of the EU by the USA for its role in counter-terrorism has been important because it has led to the EU having a certain status or reputation in the area. Thus, recognition that the USA had signed agreements related to counter-terrorism with the EU led to the adoption of other agreements. For example, US global influence on PNR data opened the door for such data to be requested by other states. The EU–US agreements led to the negotiation of similar agreements between the EU and other states, such as Australia and Canada.

To summarise, context will be operationalized by examining the global context, but also the situation in each state and whether this has enabled or constrained EU action. In addition, under context, the quality of the relationship between the EU and its interlocutor will also be considered.

Coherence

In the actorness framework, 'coherence' refers first of all to shared values and preferences (Brattberg and Rhinard, 2012: 561–2). One cannot say that all member states are similarly concerned about the threat of terrorism. They face a range of threats, with some Eastern European states often focused on Russia's intentions in Eastern Europe, for instance. The older member states that have experienced protracted terrorist campaigns have typically seen terrorism as more of a threat than the newer member states (Bures, 2010a; Monar, 2007a). However, it can be observed that these positions are now closer than they were in the past.

In addition, not all member states have to be highly concerned about terrorism for action to occur. Often, there has been solidarity with other member states or external actors, which was evident in Afghanistan, for example. Few member states had reason to have a strong interest in the region. Nevertheless, they followed the USA into Afghanistan because they wanted to support and show solidarity with Washington. Some did the same in Iraq, although the EU member states were more divided on that occasion. The fact that the EU adopted a European Security Strategy in 2003, a Counter-terrorism Strategy in 2005, an Internal Security Strategy in 2010, and a Global Strategy for the European Union's Foreign and Security Policy in 2016 shows that there has been at least a reasonable amount of similarities in the preferences of the EU and its member states. The adoption of common strategies – which could also be regional strategies – reflects some level of unity amongst member states.

The second component of coherence concerns procedures. The European Commission is able to use certain tools itself, but there have been occasions on which this has been contested by the European Parliament. For instance, Wennerholm et al. (2010: 20–21) have discussed how the European Commission originally attempted to fund a border-management project in the Philippines, but the European Parliament argued that this exceeded the Commission's powers and the Court of Justice of the EU eventually annulled the decision. Also, procedures have been important sites for contestation, as different institutions and member states have pulled in different directions over specific issues. For example, turf wars have occurred amongst institutions for power advantages, as has happened on a few occasions in relation to counter-terrorism issues. For example, data retention measures have been hotly contested, with the European Parliament often seen as a protector of civil liberties, while the European Commission and the Council have had their own motives for pursuing data retention measures. This has been seen in the Parliament's resistance to the DRD, the EU-US PNR Agreements, and the EU-US SWIFT Agreement (Monar, 2010; Ripoll Servent and MacKenzie, 2011). Furthermore, there have been some cultural differences amongst member states. For example, the UK - when it was still an EU member

state – often called for stronger cooperation between the EU and third states and strong counter-terrorism measures, whereas Germany has typically been more resistant (Monar, 2010; Rees, 2008; Ripoll Servent and MacKenzie, 2017). This is an over-simplification to some extent, as states may alter their position over time, notably as a result of changes to the composition of their government.

One can also distinguish between the horizontal and vertical dimensions of coherence. The horizontal dimension of coherence refers to cohesion among the institutions or among member states, while vertical coherence deals with cohesion between member states and EU institutions. Coherence, therefore, deals with values and preferences, policies, procedures, and vertical and horizontal issues. The analysis presented in this book also spans a number of years, which highlights whether the EU is able to 'stick to the line' (Brattberg and Rhinard, 2012: 563) over time (or not). In that way, 'coherence' also subsumes 'consistency', which was a separate criterion in Brattberg and Rhinard's (2012) set of criteria.

Capability

'Capability' starts with the authority that the EU has to act. At the highest level, this means that counter-terrorism is one of the responsibilities of the EU in the treaties, which is the case. Following on from this, it is worth pointing out that the EU's role in counter-terrorism could mostly be seen as that of a coordinator, with a focus on law enforcement and the root causes of terrorism. This is related to a large extent to the methods that European states have developed over time. This has also been where the EU possesses tools that it can use (Hoffman, 1999; Monar, 2015; Rees and Aldrich, 2005). To delimit counter-terrorism and ensure that not everything is defined as such, it can be considered to have three main components: (1) instruments specifically focused on the 'fight against terrorism', (2) instruments dealing with complex threats and disasters, and (3) capacities within EU institutions to help member states (Rhinard et al., 2007: 89).

Terrorism is an extremely complex threat to address, which requires finance, education, health, and many other types of combined policy action. Therefore, counter-terrorism cannot be limited to military or traditional foreign policy tools as a result.

To offer further specifics of the EU's efforts to combat terrorism, it is important to consider the counter-terrorism tools that are funded by the EU and deployed under its flag – which will be considered in greater detail in each of the case studies in Chapters 4–6:

1. Agreements with third states: the EU has signed a number of agreements with third states, most importantly the USA, which have been motivated

by the terrorist threat and/or have had counter-terrorism as one of their purposes.

- Europol and Eurojust: these two EU agencies the former to facilitate
 police cooperation and the latter for judicial cooperation have grown in
 importance over the past few decades and have increasingly developed
 their counter-terrorism activities, including in cooperation with third
 states and organisations.
- 3. Capacity-building programmes related to counter-terrorism: the EU has funded a range of programmes intended to build up the counter-terrorism capabilities of recipient states.
- 4. Instrument contributing to Stability and Peace (IcSP): this instrument succeeded the Instrument for Stability (IfS) and the Rapid Reaction Mechanism (RRM).
- 5. Development Cooperation Instrument (DCI): while primarily concerned with development issues, the ability to provide money for institutional reform and some of the thematic programmes being directly related to counter-terrorism mean that this instrument is relevant to this analysis.
- European Instrument for Democracy and Human Rights (EIDHR): this instrument can also be seen as making an important contribution to counter-terrorism, given the links between political grievances and terrorism.
- 7. European Neighbourhood Partnership Instrument (ENPI): this instrument focuses on neighbouring states, which are part of the European Neighbourhood Policy (ENP).
- Common Security and Defence Policy (CSDP) missions: while these EU missions might not include terrorism in their mandate, they can have a direct bearing on a state's capabilities for combating terrorism, including the capabilities of the army and the police (see Wennerholm et al., 2010).

This book is underpinned by an understanding that these eight components all contribute to the EU's counter-terrorism capabilities. The EU does not make use of all of these capabilities in every case, which is why each case study focuses on the most relevant instruments. Capability is therefore concerned with the EU's legal authority in the area and the instruments that could help combat terrorism. This framework helps gauge the extent of EU actorness in each area of the world

CONCLUSION

This chapter has introduced the analytical frameworks that will be applied in the remainder of this book. 'Collective securitization' will be applied in Chapter 2 in order to analyse how the EU originally became involved in the

field of counter-terrorism. 'Actorness' will be applied in Chapters 4–6 in order to establish and to evaluate the exact role of the EU in counter-terrorism in various regions of the world. Indeed, this book provides a comprehensive discussion of the development of the EU as a global counter-terrorism actor in order to gauge how far the EU has come since 9/11. The following chapters will show that the EU has made huge strides forward as a counter-terrorism actor since these devastating terrorist attacks. While it may lack all the capabilities of its member states, it has become a global counter-terrorism actor in its own right. The EU can fulfil a number of important functions. It can be used by individual member states to gather support for individual objectives; it can act on behalf of member states that are unable to act in some parts of the world; and it can help all member states speak with one voice.

2. The collective securitization of terrorism in the EU

This chapter examines how terrorism has come to be perceived as a European security threat and how this change has led to the development of EU counter-terrorism cooperation. It is necessary to first understand how and why the EU became involved in the field of counter-terrorism before being able to establish and assess what exact role it has played as a terrorism actor on the global stage.

A month after the terrorist attacks in Brussels in March 2016, a debate on existing tools and possible future measures to combat terrorism took place in the European Parliament and notably featured interventions by Commission President Jean-Claude Juncker and the Dutch Minister of Defence Jeanine Hennis-Plasschaert who represented the Council Presidency. A few months earlier, Europol, the European Union (EU)'s law enforcement agency, had launched the European Counter Terrorism Centre (ECTC) as a hub of expertise and an operational centre aiming to strengthen the EU's response to the threat of terrorism. A year later, the EU adopted Directive (EU) 2017/541 on combating terrorism, which further harmonized the definition of terrorist offences in all EU member states, in particular in relation to foreign terrorist fighters and terrorist financing. Nowadays, any significant terrorist attack in Europe tends to lead to questions about possible failures of the EU and calls for strengthened counter-terrorism cooperation amongst member states. This strongly suggests that, although some have remained sceptical about the effectiveness of its policies (e.g. Bures, 2011), the EU has acquired an increasingly significant role in the efforts to combat terrorism across Europe.

This was not always the case. For years, several European countries, including Spain, Italy, the UK, and West Germany, tackled the terrorist threat that each of them was facing largely on their own. The general perception was that each state was confronted with a distinct threat, namely ETA ('Basque Fatherland and Liberty') in Spain, the Red Brigades in Italy, the Irish Republican Army (IRA) in the UK, and the Red Army Faction in Germany (Chalk, 1996a). Cooperation amongst European states remained extremely limited. Some European governments even accused others of actually enabling

the growth of terrorism. For example, as explained by Heiberg (2007: 42), after 1982.

to the intense irritation of the Spanish socialists, the French government insisted on continuing its policy of offering ETA a safe haven in southern France, viewing the militants as freedom fighters and political refugees. This safe haven offered ETA the opportunity to train its recruits, plan its operations, headquarter its leadership, and maintain its weapon supplies. After operations in Spain ETA militants would withdraw to the impunity of France.

Thus, compared to the lack of a shared perception of the terrorist threat and the virtual absence of counter-terrorism cooperation amongst European states in the 1970s and 1980s, the existence of EU-wide debates, legislative instruments, and practical cooperation nowadays is particularly remarkable. This chapter explores this change and seeks to explain it using the framework of 'collective securitization' introduced in the previous chapter. There are five sections in this chapter, each of which focuses on a specific stage of the collective securitization process.

STATUS QUO SECURITY DISCOURSE AND POLICIES

Although one may consider that there have been various terrorist events affecting European countries throughout history, it is generally agreed that modern terrorism in Europe can be traced back to the 1970s (Bossong, 2013: 25). Several European states saw attacks committed by various ethnonationalist-separatist groups, including ETA in Spain and the IRA in the UK. Others were hit by left-wing terrorist groups, such as the Red Brigades in Italy and the Red Army Faction in West Germany. In addition, various Middle Eastern groups, including Black September, committed terrorist attacks in Europe in an attempt to influence the foreign policies of Western European states (Bossong, 2013: 25).

However, these terrorist attacks did not lead to the collective securitization of terrorism within the predecessors of the EU, the EEC and the European Community (EC). Each state confronted with a terrorist threat tended to consider it a domestic issue that had to be dealt with 'at home'. Terrorism was not collectively securitized in Europe and counter-terrorism cooperation was therefore severely limited. This was due to several reasons (Chalk, 1996a). The first was the lack of a shared perception of terrorism as a significant threat. Although some European states were hit by terrorist attacks, others were not, which also meant that the latter did not see any need for counter-terrorism cooperation. The second factor was the existence of differences amongst the various terrorist threats affecting European states. Those were not under attack from the same terrorist groups. In some cases, they were not even facing the

same type of terrorist threat. For example, the UK was mainly preoccupied with the IRA, whereas Spain's counter-terrorism efforts focused on ETA. Other European countries were also hit by terrorist groups, but those were not ethnonationalist-separatist in nature, but rather inspired by a violent left-wing ideology. Examples in that respect included the Red Brigades in Italy and the Red Army Faction in West Germany. A third factor explaining the initial lack of collective securitization of terrorism in Europe was the persistence of concerns for national sovereignty, which went hand in hand with the existence of different historical, political, and legal traditions amongst various European states. For example, the French authorities were willing to negotiate openly with terrorists and accommodate their demands, which stood in stark contrast to the approach favoured by other governments, such as the British government. Such differences made cooperation across Europe more difficult for a long time.

Nevertheless, the terrorist attacks that took place in the 1970s, in particular the 1972 Munich massacre during the summer Olympic Games, led to the start of some limited cooperation on counter-terrorism amongst European states. First of all, counter-terrorism officers began to develop bilateral contacts, which led to a growth in mutual trust (Chalk, 1996a: 121-2). In addition, some first attempts at multilateral counter-terrorism cooperation took place, but outside the framework of the EU's predecessors, the EEC and the EC, since they did not possess any competence in the field of counter-terrorism or even security more broadly. In particular, one can identify five venues in which European states began to cooperate on counter-terrorism from the 1970s onwards. First of all, European states started coordinating their counter-terrorism strategies within Interpol. This organization has a global membership, but has developed a regional infrastructure to specifically address European challenges, including a European regional assembly, a European committee, and a European bureau (Chalk, 1996a: 125). Second, counter-terrorism cooperation also developed within the framework of the Council of Europe, where the European Convention on the Suppression of Terrorism, which entered into force in 1978, was drawn up. It mainly aimed to facilitate the extradition of persons having committed acts of terrorism by attempting to limit the use of the 'political offence' exception, which had hitherto been a major obstacle to the extradition of terrorists. However, it was not ratified by several of its early signatories, including Belgium, France, Greece, and Italy for a considerable number of years, which limited its impact in practice (Bossong, 2013: 26). Third, some counter-terrorism cooperation at the practical level also developed amongst European states within the Trevi Group, which was an intergovernmental forum independent of the EEC, from the mid-1970s onwards (Argomaniz, 2009a: 152–3). In particular, two operational working groups (WGs) dealing with terrorism matters were established. WG1 focused on information exchange

and mutual assistance regarding terrorist activities, whereas the remit of WG2 was mainly training matters and the exchange of technical and scientific information to facilitate the fight against terrorism. Although the achievements of Trevi cooperation were rather modest in practice, the existence of this group helped develop trust amongst individual police and intelligence officers across the EEC. Fourth, the Police Working Group on Terrorism (PWGOT) was a semi-permanent working group independent of Trevi, which promoted closer working relationships amongst police officers across Europe – in contrast with Trevi, which was a ministry-level initiative. Fifth, counter-terrorism was also part of the nascent Schengen cooperation. This is because terrorism was identified as one of the illicit activities that might benefit from the lifting of the internal border controls within the Schengen zone, which therefore called for it to be addressed by Schengen compensatory measures. Although Schengen cooperation did not specifically focus on counter-terrorism, it contributed to the development of counter-terrorism cooperation amongst EEC/EC member states by facilitating cross-border law enforcement and police cooperation (Chalk, 1996a: 127).

Thus, counter-terrorism cooperation amongst European states remained severely limited for a long time, actually up to 2001, as will be explained in the next section. One can identify three main explanations for this lack of progress, namely the intergovernmental nature of counter-terrorism cooperation attempts, the lack of a common perception of the terrorist threat, and the concomitant absence of a shared definition of terrorism, as well as the wide range of institutional venues, both inside and outside the EEC/EC, in which member states had begun to cooperate in a somewhat haphazard manner in the 1970s (den Boer, 2000).

PRECIPITATING EVENT

The single precipitating event that was so serious that it disrupted the status quo in the field of counter-terrorism in Europe was the 9/11 attacks. On 11 September 2001, four airplanes were hijacked in the USA by 19 terrorists associated with al-Qaeda. Two aircraft were flown into the twin towers of the World Trade Center in New York City, whilst the third plane hit the Pentagon and the fourth one crashed in a field in Pennsylvania. In total, almost 3,000 people lost their life in these terrorist attacks. With reference to the collective securitization framework, the terrorist attacks on 11 September 2001 were very serious due to their large scale, the high number of victims, as well as the original modus operandi of the attackers, which made them unique. Another key factor was the fact that these attacks received extensive media coverage, including the live broadcast of the plane slamming into the second of the twin towers and their subsequent collapse, and were watched by millions of people

around the world (Goncalves, 2016: 1). These terrorist attacks also disrupted the status quo in the field of counter-terrorism by confirming what a growing number of experts had been claiming in recent years, namely that the character of the terrorist threat was changing. In 1999, Laqueur (1999) had argued in The New Terrorism that the nature of terrorism was evolving as it was moving away from the calculated use of violence for political gain towards fanaticism and the pursuit of catastrophic destruction. Other experts had warned against the threat of what had been variously described as 'super-terrorism', 'mega-terrorism' or 'hyper-terrorism' in order to denote that the world was now facing a new type of terrorist threat, which was thought to be more lethal, more dangerous, and less predictable than it had ever been before (Neumann, 2009: 3). In other words, the terrorist attacks on 11 September 2001 reinforced the growing perception that Western governments now faced a new terrorist threat, namely that of Islamist terrorism, which was seen as being of far more concern than the threat that had been posed by left-wing or ethnonationalist terrorist groups in some European countries. Even in the European states that continued to experience terrorist acts by ethnonationalist groups, such as France and Spain, the threat of Islamist terrorism was perceived to be more serious because of the characteristics that have just been discussed.

Thus, this chapter argues that 9/11 was the single precipitating event for the collective securitization of terrorism in the EU. As demonstrated earlier, the terrorist attacks on 11 September 2001 were so grave and unique in character that they were sufficient on their own for the collective securitization of terrorism to take place in the EU in their aftermath. This is not to say that the numerous Islamist terrorist attacks that happened later in Europe, including in Madrid, London, Paris, Brussels, and Nice, did not have any impact on the EU counter-terrorism policy, for they certainly did. However, from an analytical viewpoint, they can be seen as belonging to the final stage of the collective securitization framework, which concerns routinization and the emergence of a new status quo, rather than playing a role in the initial collective securitization process.

THE SECURITIZING MOVES AND THE RESPONSE OF THE AUDIENCE

Very shortly after the terrorist attacks on 11 September 2001, President Bush made a key speech, which some called the 'act of war speech', which constituted a platform for the emerging norm to join the 'war on terrorism' (BBC, 2001a). He declared the following:

The deliberate and deadly attacks, which were carried out yesterday against our country, were more than acts of terror. They were acts of war [...] This enemy

attacked not just our people but all freedom-loving people everywhere in the world [...] We will rally the world [...] This will be a monumental struggle of good versus evil, but good will prevail.

These statements show that the US administration considered these acts to be acts of war, even though, conventionally, they would have been defined as terrorist attacks. However, the definitions and norms were about to change. The acts were perceived as an act of war, against which the USA needed to take necessary military action. Also, the USA needed other countries to join in the fight against terrorism in order to legitimate the war and to combat the terrorists.

One should note the enormous pressure for other countries to adopt this norm. Bush defined appropriate action in terms of fighting in the 'war against terrorism' and made an even stronger case by distinguishing between 'good and evil'. Since nobody would want to be associated with 'evil', it was imperative to join the war in order to count as a 'force of good'. Later, Bush enforced this emerging norm by stating that 'you are either for us or against us' (BBC, 2001a). In this presentation, there was no grey area. The countries supporting terrorism would later be called the 'axis of evil'. It was therefore appropriate and even imperative to support the USA, in whichever way needed. NATO ambassadors also invoked the mutual defence clause in the organization's founding treaty for the first time ever in its history (NATO, 2001), even though, subsequently, the organization was to be largely side-lined throughout the military campaign against Afghanistan.

After NATO, the international community decided to join in. Broadly speaking, it would have been considered inappropriate to oppose this, or even to be lukewarm about it. The UN Secretary-General Kofi Annan said that the international organisation's host country and its host city had just been subjected to a terrorist attack of a kind one hardly dared imagine 'even in our worst nightmares' (United Nations, 2001c). He also asserted the fact that the UN Assembly had continuously condemned terrorism on numerous occasions and had called on all states to adopt measures to prevent it and strengthen international cooperation against it. 'All nations of the world must be united in their solidarity with the victims of terrorism, and in their determination to take action – both against the terrorists themselves and against all those who give them any kind of shelter, assistance or encouragement (United Nations, 2001c). The importance of these statements becomes apparent when one considers the speed and timeframe in which they were made. In the space of one day, almost the entire world had already rallied behind the USA. Even more importantly, there was a widely shared view that there was an obligation for the international community to join the 'war on terror' as these terrorist attacks had been attacks on the 'civilized world'

Several European political leaders also reacted shortly after the terrorist attacks against the USA and rallied behind President Bush and his administration. In addition to extending their sympathy to American citizens, they also securitized terrorism, notably by presenting it as an act of war. For example, the then German Chancellor Gerhard Schröder described the attacks as 'a declaration of war against the entire civilized world', whilst the then EU External Relations Commissioner Chris Patten called 9/11 'an act of war by madmen' (Bossong, 2013: 39). The German Chancellor also called on European nations to come together within the framework of the EU to combat global terrorism: 'Only if we put in place common policing and judicial resources can we ensure that there will be no hideouts for terrorists and other criminals in the European Union.' Interestingly, he also noted that, in the aftermath of 9/11, '[we] are ready to make Europe into an international player with global influence' (BBC, 2001b). What is particularly noticeable here is that the referent object in the securitizing discourse of these European leaders was not Europe or the EU. This is understandable since, although some of the perpetrators had links to Europe, the terrorist attacks on 11 September 2001 did not target Europe. Rather, the referent object was broader and encompassed 'the civilized world', of which the USA and European states are part. The securitization of terrorism also played an agenda-setting role. This was notably evidenced by Jacques Chirac's statement in New York on 19 September 2001 – in the context of a meeting with President Bush – that 'the fight against terrorism had evidently become an absolute priority in today's world'.

FORMULATION AND EXECUTION OF POLICIES

The EU rapidly moved beyond statements of sympathy and solidarity with the USA to discussing the development of an EU counter-terrorism policy. As early as 12 September 2001, the Council requested the Presidency, the High Representative for the Common Foreign and Security Policy, and the Commission to submit, as soon as possible, a report on concrete measures that may be recommended to '[increase] the capacity of the European Union to effectively fight, together with the United States and other partners, international terrorism' (Council of the European Union, 2001: 4; italics added). On 14 September, in a Joint Declaration issued by the Heads of State and Government of the European Union, the President of the European Parliament, the President of the European Commission, and the High Representative for the Common Foreign and Security Policy, the EU emphasised two specific policy objectives in the fight against terrorism, namely '[strengthening the] intelligence efforts against terrorism' and '[accelerating] the implementation of a genuine European judicial area, which will entail, among other things, the creation of a European warrant for arrest and extradition, in accordance with the Tampere conclusions, and the mutual recognition of legal decisions and verdicts' (European Union, 2001: 2).

The JHA Council met on 20 September 2001, agreeing on a package of measures to combat terrorism. Those were endorsed by an extraordinary European Council meeting on the next day. At the extraordinary Council Meeting on 21 September 2001, the European Council launched an ambitious 'Action Plan to Combat Terrorism'. Its cornerstones were 'close cooperation between all the Member States of the EU' and the adoption of a 'coordinated and interdisciplinary approach embracing all Union's policies' (European Council, 2001a). The European Council called for the use of all the tools at the EU's disposal, amongst which are legislative and operational, repressive and preventive, internal and external measures. This made the European collective response as comprehensive as possible and essentially multidimensional.

It was now rather clear that the EU would take part in the 'war on terror', although the exact modalities of this engagement still had to be established. Speaking before the European Parliament, the Commissioner responsible for the Area of Freedom, Security and Justice, Antonio Vitorino, declared the following (*Financial Times*, 2001):

Terrorist acts are committed by international groups with bases in several countries, exploiting loopholes in the law created by the geographical limits on investigators and often enjoying substantial financial and logistical resources. Terrorists take advantage of differences in legal treatment between States, in particular where the offence is not treated as such by national law, and that is where we have to begin.

Thus, Vitorino highlighted the crucial character of legal harmonization to tackle terrorism.

One of the key policy achievements during that phase was the adoption of the EAW. In that respect, the Commission followed the above-mentioned rhetoric with a very timely proposal (Kaunert, 2007, 2010c), which had already been under preparation for about two years by the time it was tabled. Vitorino initially intended to launch it under the Spanish presidency in the first half of 2002 due to Spain's strong support of the issue in order to solve its own problems with the ETA terrorists. Yet, with the emerging norm of the 'war on terror', it became apparent that fast action was required. Ministers in the AFSJ would be under intense pressure to behave appropriately and settle their differences rapidly. Vitorino remarked: 'If we do not get agreement, and it should be a substantial agreement to cope with the global threat, it will be difficult to explain to the public why we failed' (*Financial Times*, 2001). Therefore, the European Commission's strategy was for the EAW to be presented as a counter-terrorism measure and to be amalgamated with other such measures,

such as the Framework Decision on Combating Terrorism, which established the EU's definition of terrorism (Kaunert, 2007, 2010c).

The EAW is exemplary for the scope of integration achieved in EU counter-terrorism after 9/11, as it abolishes extradition amongst EU member states. Vogel describes the importance of the introduction of the principle of mutual recognition in the EAW as a 'revolution in extradition law' (Vogel, 2001: 937). Until the adoption of the EAW, extradition between EU member states was based on several intergovernmental measures in the realm of international law (Peers, 2001). The EAW has not created international law, but rather transnational or 'European' law (Wagner, 2003). It replaces all previous international legal instruments amongst the different member states with an EU legal instrument. In effect, this is a form of European extradition law, which is revolutionary in many ways. For example, it abolishes the term 'extradition' and replaces it with the term 'surrender' (Douglas-Scott, 2004). The legal effect of this measure is subject to the jurisdiction of the Court of Justice of the EU (Peers, 2001) if member states sign a declaration approving this. The EAW – which applies to 32 categories of crimes – abolishes the principle of double criminality for serious offences (Douglas-Scott, 2004), which means that an EAW may not be contested on the basis that it concerns an activity not criminalized in a surrendering member state.

In addition to the EAW, a key legislative development in EU counter-terrorism cooperation during that phase was the adoption of Council Framework Decision 2002/475/JHA of 13 June 2002 on Combating Terrorism. This was a crucial advance in the development of EU counter-terrorism cooperation, because it meant that, for the first time, there was a common definition of what constitutes terrorism at the EU level. This was particularly remarkable given that a number of member states did not even have a legal definition of terrorism until then (Douglas-Scott, 2004). The EU's Framework Decision on Combating Terrorism, agreed politically in December 2001, first defined what was meant by terrorist acts in three parts: (1) the context of an action; (2) the aim of the action; and (3) the specific acts being committed. Terrorist acts

must be intentional acts [...] which given their nature or context, may serve to damage a country or an international organisation. These acts must be committed with the aim of either seriously intimidating a population or unduly compelling a Government or international organisation to act or fail to act, or seriously destabilizing or destroying the fundamental political, constitutional economic or social structures of a country or international organisation.

In addition, a list identified eight specific acts. The definition also covered behaviours which may contribute to terrorist acts in third countries. Thus, the Framework Decision on Combating Terrorism ensured that terrorist offences were to be punished by heavier sentences than common criminal offences would have been in all EU member states. Furthermore, it approximated the level of sanctions amongst member states according to the principle that sentences have to be both proportional and dissuasive. Member states were made legally responsible for acting in cases of terrorist incidents taking place on their own territory or committed against their own people. Thus, this Framework Decision was also favourable for EU–US cooperation in the fight against terrorism, because it ensured that terrorism was now recognized as a criminal offence on both sides of the Atlantic.

Thus, the terrorist attacks on 11 September 2001 became a normative defining point – for European integration and for the European Commission's role in this process. The European Commission played a significant role in leading the EU into this emerging 'war on terror'. It managed to construct an important role for the EU in this policy area by being a 'strategic first mover' in order to shape the debate, as indicated above. It also allied with the USA to put pressure on reluctant member states. However, the European Commission had to use a 'carrot and stick' approach in its relations with the USA, as symbolized by the fact that the former US Homeland Security Secretary Thomas Ridge stated that his greatest regret was to have not worked more closely with the EU from the start (Lebl, 2006: 125). The European Commission managed to contribute to the construction of a role for the EU in the 'war on terror', and thereby rhetorically and practically 'securitized terrorism'. Furthermore, it used security reasons to move the negotiations faster, in order to secure the adoption of the EAW and an increase in EU-US cooperation. This underlines the argument that securitizing actors use securitization as a political strategy and have clearly defined political objectives.

However, it is important to underline that these significant policy developments at the EU level were not immediately followed by changes at the domestic level. As argued by Argomaniz (2009a: 157), '[the] reality [...] is that the implementation record during this period [was] poor'. This became particularly evident in the wake of the terrorist attacks in Madrid in March 2004. On that occasion, the European Council (2004a: 3) adopted a 'Declaration on Combating Terrorism', which notably emphasized that 'the effective combating of terrorism requires that measures adopted by the Council be effectively and comprehensively implemented by Member States'. The Declaration listed six key legislative measures that had to be adopted by all member states by June 2004. Even then, as highlighted by Monar (2005: 142), 'nine Member States had, by [December 2004], still not fully implemented the framework decision on combating terrorism. In addition, nine (partly different) EU countries had not implemented the legislation on joint investigation teams and one Member State (Italy) had still parliamentary ratification problems with the European arrest warrant.'

THE ROUTINIZATION OF EU COUNTER-TERRORISM COOPERATION AND THE EMERGENCE OF A NEW STATUS QUO

After this intense phase of collectivization of the terrorist threat in the EU, a period of relative inertia followed until the terrorist attacks against Madrid in March 2004. This is actually a pattern that has tended to repeat itself over the years (Argomaniz, 2009a, 2010). Terrorist attacks tend to lead to a spike in the number of EU counter-terrorism policy initiatives, which is then followed by a period of deceleration until a new terrorist attack pushes terrorism to the top of the policy agenda again. In so doing, the new terrorist attack sheds light on the remaining challenges and possible problems of implementation, which may then be addressed by new initiatives. For that reason, the implementation of EU measures at the domestic level often lags one terrorist attack behind. Nevertheless, despite what some may perceive as the slow growth of EU counter-terrorism cooperation (Argomaniz, 2011; Bures, 2011), policy developments clearly point towards the gradual strengthening of the EU's role in counter-terrorism governance in Europe. Institutional innovations reinforce and solidify this trend, as explained in the remainder of this section.

Following the 11 March 2004 attacks on Madrid, which killed almost 200 people, the EU stepped up again its efforts against terrorism, with a particular emphasis on the international dimension. The 'solidarity' Declaration on Combating Terrorism of 29 March 2004, agreed upon by the European Heads of State and Government (European Council, 2004a), strongly emphasized the need 'to reduce the access of terrorists to financial and other economic resources' and 'to address the factors contributing to the support for and recruitment into terrorism'. The EU also made counter-terrorist clauses an important element of the agreements concluded with third countries, which it described as 'a basis for political dialogue and a tool for EU external action' (European Council, 2004a: 7). Further to the adoption of the Framework Decision on Combating Terrorism, other significant EU counter-terrorism measures were adopted during that phase, such as the EU-US Container Security Initiative (CSI) in 2004, which was negotiated to ensure greater protection for containers being transported from Europe (and a number of other important ports around the world) to the USA (MacKenzie, 2012). Furthermore, the European Commission negotiated the first EU-US PNR Agreement in 2003-04 on the basis that data was dealt with under the then first pillar, but the Court of Justice of the EU ruled this to be an incorrect legal basis and invalidated the agreement, thereby pushing the responsibility to negotiate the next agreement to the member states on the basis of the then second and third pillars. The Commission used the momentum of these negotiations to later push for an EU PNR system. A greater role for the European Commission did not really emerge until 2009, when the Treaty of Lisbon came into force (Kaunert, 2010c). Through this treaty, the European Commission gained greater powers to negotiate agreements with third states and from then on was able to negotiate agreements in the former third pillar, starting with the re-negotiation of the SWIFT Agreement between February and July 2010 after its rejection by the European Parliament in February of that year (Ripoll Servent and MacKenzie, 2012). Further agreements that have been negotiated by the European Commission include the EU's PNR Agreements with the USA (2012), Australia (2012), and Canada (2014). Since the early 2010s, the European Commission has played a more significant role in counter-terrorism, notably in the external dimension.

This section also considers in greater detail the role of an underestimated body – the office of the EU CTC. In 2004, following the Madrid bombings, the EU decided to appoint a CTC, the first incumbent being Gijs de Vries. This was no easy position given that the EU is an organization of more than 25 member states, all of which have had different experiences with terrorism, making it inevitable that some member states will be more concerned about terrorism than others (Bures, 2011; Monar, 2005). The weakness of the position appeared to be confirmed on the grounds that de Vries resigned in March 2007, with the post being left vacant for six months and rumours abounding that the post itself might thereafter be discontinued (Bossong, 2013). Nevertheless, Gilles de Kerchove – an EU insider – took up the position in September 2007 and held it until 2021. A committed transatlanticist, he made significant efforts to improve counter-terrorism cooperation with certain key partners, particularly the USA, making the CTC active not only within Europe, but around the world. There is no specific legislation establishing the post and setting out the Coordinator's powers. Formally, the post was established by the European Council in its Declaration on Combating Terrorism that was adopted shortly after the Madrid bombings (Council of the European Union, 2004d). As a consequence, although quickly nicknamed the EU's 'Mr Terrorism' or the 'European terrorism czar' by the media, the actual formal mandate of the EU CTC is rather general and his precise tasks and powers are not entirely clear.

In practice, the EU CTC has audited the progress of the 200+ measures from the EU Counterterrorism Action Plan. For this purpose, he publishes regular reports on the implementation of the Strategy and Action Plan to Combat Terrorism, as well as other occasional and more specialised reports and discussion papers on various aspects of the EU counter-terrorism policy. Although these reports have sometimes been rather critical of both the deficiencies of the EU member states and the EU agencies involved in the fight against terrorism, at least in the latter case, the content of the report is generally

an outcome of prior consultations. The Coordinator should also coordinate a plethora of other institutions and committees that have a role in different aspects of the EU counter-terrorism policy, and encourage greater cooperation between the Commission (which drafts legislation) and the Council (where national Interior and Foreign Affairs ministers meet to decide EU policies). However, the problem has been that the Coordinator formally lacks both powers and resources to coordinate the work of the Commission, the Council, and the other EU agencies involved in the fight against terrorism. With a very small office, one can hardly expect a hands-on coordination of the complex area of counter-terrorism that is overcrowded with diverse EU and non-EU structures and agencies. Another and arguably even more important role of the EU CTC is to cajole the member states into timely implementation of EU-level counter-terrorism agreements and initiatives. The ability of the EU CTC to do so in practice has, nevertheless, been significantly circumscribed because they cannot force national governments to act, have no independent budget, and cannot propose legislation.

Europol has also become an important player in EU counter-terrorism. Its role was reinforced in the aftermath of the 9/11 attacks and has steadily increased since then. Given the importance of the threat of Islamist terrorism, compared to other forms of terrorism, Europol's counter-terrorist efforts have been mainly directed towards tackling this form of terrorism. Its activities in the field of counter-terrorism have been as follows: (a) analyzing gathered information from strategic, tactical, and operational perspectives; (b) undertaking threat and risk assessments, and, based on their results, crafting and implementing awareness activities; (c) on request, supporting operational investigations in the member states through various means; (d) monitoring, tracking and preventing all forms of illicit trafficking of nuclear material, strong radiological sources, arms, ammunition, explosives, and weapons of mass destruction; (e) establishing regular contacts and a sound relationship with experts in coping with terrorism and counter-proliferation.

In order to strengthen Europol's counter-terrorist efforts, the European Counter Terrorism Centre (ECTC) was created within Europol in January 2016. The official website of Europol and serving officers have presented the ECTC as an operations centre and hub of expertise that reflects the growing need for the EU to strengthen its response to terrorism. The ECTC focuses on (a) tackling foreign fighters; (b) sharing intelligence and expertise on terrorism financing amongst member states (through the Terrorist Finance Tracking Programme and the Financial Intelligence Unit); (c) monitoring and suggesting preventive measures against online terrorist propaganda and extremism (through the EU Internet Referral Unit); (d) illegal arms trafficking; and (e) international cooperation among counter-terrorism authorities. Thus, over the years since 2001, counter-terrorism cooperation has become increasingly insti-

tutionalized in the EU, as epitomized by the creation of the post of CTC and of the ECTC within Europol. This institutionalization has greatly contributed to the routinization of EU counter-terrorism practices.

CONCLUSION

This chapter set out to explore the significant growth in EU cooperation on counter-terrorism, which includes various important legislative instruments that have had a significant impact on the EU member states. It did so by drawing upon the concept of 'collective securitization' introduced in Chapter 1. The analysis highlighted how 9/11 was a precipitating event, which led some EU leaders to call for the governments of the EU member states to agree on developing an EU counter-terrorism policy and step up their counter-terrorism cooperation with the USA. Various counter-terrorism legislative instruments have been adopted, such as that defining terrorist acts, some of which have had a significant impact on national counter-terrorism policies. The 9/11 attacks were therefore used by some actors to convince the EU member states that, although some of them might continue to face threats from ethnonationalist-separatist terrorist groups, they now all faced one major, collective terrorist threat, embodied at the time by al-Oaeda, rather than each of them facing a distinctive threat as had hitherto been a common view across Europe. This was a crucial development as it paved the way for the development of EU counter-terrorism cooperation for the first time. The subsequent, growing institutionalization of this cooperation, in particular through the establishment of the post of CTC and the creation of the ECTC within Europol, has significantly contributed to the routinization of counter-terrorism practices in the EU.

However, there are limitations to the argument developed in this chapter. Given its focus on the EU itself, it has been unable to do justice to the richness and complexity of the EU-member state relationship on terrorism. While on some occasions the EU led member states in the fight against terrorism and those then enacted national legislation, it is important to acknowledge that counter-terrorism largely remains the preserve of member states, as has been particularly noticeable when it comes to its policing and intelligence dimensions. The EU mainly aims to add value in counter-terrorism above the national efforts of the EU member states. Nevertheless, more than 20 years after 9/11 and the initial collective securitization of terrorism, the idea that the EU has a significant role in combating terrorism alongside its member states is no longer in question.

NOTE

 Most sections of this chapter have previously appeared in the following article: C. Kaunert and S. Léonard (2019a), 'The Collective Securitization of Terrorism in the EU', West European Politics, 41(2): 261–77 (https://doi.org/10.1080/01402382.2018.1510194). The authors thank the publishers for allowing them to use this material in this book.

3. Institutions in EU counter-terrorism

This chapter analyzes the role of the EU institutions in counter-terrorism, namely the European Commission, the European Parliament, the Council Secretariat and the EU CTC, as well as the European External Action Service. This chapter suggests that EU institutional actors have played a crucial role in shaping the development of EU counter-terrorism in particular ways. Traditionally, member states tended to see security threats as calling for national solutions. However, as previously discussed, 9/11 led member states to acknowledge that greater cooperation was needed on counter-terrorism and chose the EU as the best vehicle for such endeavours. Some EU institutions also seized on 9/11 and managed to channel member states towards developing 'European' - rather than 'national' - solutions. This chapter shows an increase in supranational security governance in counter-terrorism since the early 2000s. This can be mainly explained as follows: cross-border security issues led to a greater demand for EU legislation and supranational organizations had considerable influence on the supply of this legislation (Stone Sweet et al., 2001: 11). Institutional change has been facilitated by exogenous shocks like 11 September, as well as the terrorist attacks on Madrid and London, and policy entrepreneurship. As observed by Stone Sweet et al. (2001), '[p]olicy entrepreneurs generate and attempt to propagate ideas that will define problems and solutions in ways that other actors find convincing and useful'.

More precisely, this chapter analyzes the respective roles of the EU institutions in the EU's global counter-terrorism efforts. It shows how they have developed over the years. By way of structure, this chapter examines the four main institutions involved in EU counter-terrorism. It starts by examining the European Commission, which has traditionally sought to enhance its powers and those of the EU, both internally and externally. It has not always been successful in doing so, but it has sometimes been able to use 'windows of opportunity' to strengthen EU powers. The chapter then turns to the European Parliament. It shows how its role in counter-terrorism has evolved, from emphasizing the protection of civil liberties initially to becoming a more consensual – and, some would argue, responsible – actor following the entry into force of the Treaty of Lisbon (Ripoll Servent, 2015). The chapter then considers the Council Secretariat, which also has a range of instruments at its disposal relevant to counter-terrorism. The Secretariat has played an influential role in EU counter-terrorism on occasions. This has been most evident when in alli-

ance with the European Commission. Of particular importance here is also the role of the EU CTC, who is based in the Council Secretariat. Thus, this section also considers the role of the CTC who has, over time, become an important player in EU counter-terrorism. While the post has few formal powers, it has been able to exert some pressure on member states regarding the importance of implementing the measures agreed at the EU level through its monitoring activities (Argomaniz, 2009a). In addition, the CTC has emerged as the face of EU counter-terrorism and has regularly engaged in discussions with third states, as well as being able to set a number of priorities for the EU. In short, despite a lack of formal powers, the EU's CTC has been a largely underestimated actor in EU counter-terrorism who has increasingly been seen as one of the faces of EU counter-terrorism to external actors. Finally, this chapter examines the role of the European External Action Service, which has, since the Treaty of Lisbon, become a very significant player in EU counter-terrorism, despite certain overlaps with the EU CTC.

THE EUROPEAN COMMISSION AND EU COUNTER-TERRORISM

The formal responsibilities of the European Commission include proposing legislation, enforcing EU law, setting objectives and priorities for action, managing and implementing EU policies, and representing the EU in negotiations with third parties. However, the European Commission has also been well known for its policy entrepreneurship, notably with regard to the single market (Sandholtz, 1993), but also some aspects of internal security, including more recently the EAW (Kaunert, 2007). Since 9/11, the European Commission has developed a range of tools that have been relevant to counter-terrorism (Wennerholm et al., 2010). In addition, the Treaty of Lisbon has helped the European Commission extend its role beyond the former first pillar (i.e. primarily economic activities) into policy matters that were previously included within the third JHA pillar – an area that was primarily intergovernmental in nature. From 2009 onwards, the European Commission has also been able to negotiate agreements on security matters with third states.

The European Commission's Financial Instruments

The European Commission now has a number of instruments at its disposal to support counter-terrorism activities in external relations, including the IcSP, the EIDHR, the DCI, and the ENPI (Wennerholm et al., 2010: 11–12). All of these are relevant to counter-terrorism, although none of them has combating terrorism as its sole purpose.

First, the IcSP is a Commission-based instrument created in 2014 in order to succeed the IfS, which had itself replaced the RRM in 2007. The IcSP can be mobilised in two main ways, namely providing a crisis response or assisting with emerging crises (short-term component) or longer-term capacity-building (long-term component) (European Commission, 2015a). Counter-terrorism efforts can be supported by the IcSP, which had a budget of ϵ 2.3 billion for the period 2014–20 (European Commission, n.d.c).

Second, the EIDHR, which was established in 2007, focuses on supporting and promoting human rights in third states through technical assistance (European Commission, n.d.a; Wennerholm et al., 2010). The key objectives of the EIDHR are 'supporting, developing, and consolidating democracy', as well as 'enhancing respect for and observance of human rights and fundamental freedoms' (European Commission, n.d.a). In the 2007–10 and 2011–13 strategy papers, the EIDHR could be seen as helping combat what could be considered the root causes of terrorism by seeking to spread human rights and fundamental freedoms, as well as combating problems like poverty and inequality (European Commission, 2010a; Wennerholm et al., 2010: 11). However, terrorism appeared to be absent from the 2014–17 Multiannual Indicative Programme, suggesting that this instrument might have shifted away from combating terrorism (European Commission, n.d.b).

Third, the DCI is one of the most significant instruments of the European Commission with a budget of €19.6 billion for the period 2014–20. Albeit not specifically linked to combating terrorism, the DCI can be seen as seeking to address the supposed root causes of terrorism by supporting development and democracy, as well as preventing conflicts. Thus, a number of projects supported by this instrument can be seen as contributing to the combating of terrorism around the world (European Commission, 2016a).

The final tool in the European Commission's arsenal is the ENPI, which is linked to the ENP. The ENPI had a budget of €11.2 billion between 2007 and 2013. This instrument is specifically mentioned as playing a significant role in combating terrorism (European Commission, 2016b). This is particularly important because the ENPI specifically targets states in the EU's neighbourhood, some of which have been the sources of terrorism that has affected Europe in the recent past.

While it can be argued that none of these tools specifically focuses on addressing terrorism and that all these instruments could be better resourced, the fact remains that the European Commission has a number of tools at its disposal, which it can use autonomously for combating terrorism.

The European Commission as a Negotiator of International Counter-Terrorism Agreements

In addition to managing the aforementioned tools, the European Commission has been responsible for negotiating a number of agreements for the EU. The European Commission has long had competences for negotiating trade deals, which explains why it negotiated the EU–US CSI in 2004, although it had a security dimension. This agreement aimed to ensure greater protection for containers being transported to the USA (MacKenzie, 2012). Furthermore, the European Commission negotiated the first EU–US PNR Agreement in 2003–04 on the basis of its first pillar competences. However, the Court of Justice of the EU annulled the agreement on the basis of an incorrect legal foundation, which led to the negotiation of a new agreement based on alternative terms. Nevertheless, the European Commission used the momentum created by these negotiations with the USA to promote the development of an EU PNR system (Pawlak, 2009a).

A greater role for the European Commission did not really emerge until 2009 and the entry into force of the Treaty of Lisbon. On that occasion, the European Commission gained greater powers to negotiate agreements with third states. From then onwards, it was able to negotiate agreements in the former third pillar, starting with the re-negotiation of the SWIFT Agreement between February and July 2010 after it had originally been rejected by the European Parliament in February of that year (Ripoll Servent and MacKenzie, 2012). Further agreements that have been negotiated by the European Commission include the EU's PNR agreements with the USA (2012), Australia (2012), and Canada (2014), although the Court of Justice of the EU ruled that the latter could not be signed in its current form in July 2017. Thus, a gradual expansion of the European Commission's powers has taken place since the early 2010s, which has enabled it to play a more significant role in counter-terrorism, including in its external dimension.

In sum, the European Commission's role in counter-terrorism has developed significantly since 9/11. To begin with, it played primarily an internal role. Then, its external capabilities also grew over time to encompass a wide range of activities. Moreover, it has also sought to import some ideas developed in international negotiations into the EU's internal context, although this has not always been popular, such as in the case of using PNR data for counter-terrorism purposes.

THE EUROPEAN PARLIAMENT AND EU COUNTER-TERRORISM

Prior to the entry into force of the Treaty of Lisbon in 2009, the European Parliament had almost no role in external counter-terrorism, although it had attempted to increase its powers. Originally, it had been critical of a number of practices developed during the US-led 'war on terror' in the name of protecting civil liberties in general and data protection in particular (Ripoll Servent and MacKenzie, 2011). Notably, it had mounted a legal challenge against the Council and the European Commission with regard to the adoption of the 2004 EU-US PNR Agreement. This agreement was subsequently annulled by the Court of Justice of the EU, which considered that the internal market was the wrong legal basis for its adoption. The European Parliament had criticised the agreement on the grounds of necessity, data protection, and proportionality. Furthermore, in 2006, the European Parliament carried out an enquiry into the extraordinary renditions being carried out by the USA in cooperation with some European states (European Parliament, 2006). In short, in the pre-Lisbon period, the European Parliament appeared to be a critical actor in counter-terrorism and was rather vocal against some of the counter-terrorism measures adopted by the EU without much of its own involvement. However, this changed with the entry into force of the Treaty of Lisbon in 2009. The increase in its competences has rendered it more consensual and more receptive to the security concerns of member states (Acosta, 2009; Ripoll Servent, 2015).

In terms of its powers, the European Parliament had always had the right to protest from the sidelines, notably through the adoption of resolutions, although the member states had no legal obligation to engage with its arguments. It also had the competence to mount legal challenges if it considered that its competences had not been respected. Importantly, the European Parliament gained the power of consent for most of the EU's international agreements with the entry into force of the Treaty of Lisbon. It used it to great effect only a couple of months later when it refused to consent to the EU-US SWIFT Agreement in February 2010, as further discussed later in this book. Through its power of consent, the European Parliament can, in one plenary vote, either accept or reject an international agreement on a security matter. The political consequence of this legal competence has been that the European Parliament has managed to gain some indirect influence over the negotiation of such international agreements, as negotiators now already take into account the preferences of the European Parliament – this is what Ripoll Servent and MacKenzie (2011: 400) have termed 'quasi-co-decision' (original emphasis). In short, the European Parliament has managed to gain influence beyond what was envisaged in the treaties. Although this may have come about as the result of turf wars amongst EU institutions (Pawlak, 2009b), this reinforcement of the role of the European Parliament has also had the effect of pushing EU counter-terrorism towards supranational governance.

It is also striking that, more recently, the European Parliament has accepted the priority given to security by member states – as well as the USA – although it has sought to promote greater respect for data protection and a strengthening of scrutiny, as shown by the cases of the PNR and SWIFT Agreements with the USA (European Union, 2010a). Although some observers have been critical of the influence of US norms in European countries, most of which have a tradition of stronger protection of private life and data protection standards, the negotiation of agreements with the USA has also led some to try and replicate the EU–US arrangements amongst EU member states (Argomaniz, 2009b; Ripoll Servent and MacKenzie, 2012).

In sum, the European Parliament used to be largely side-lined in counter-terrorism, but its powers have grown over time. It has moved from being a critical to a significantly more accommodating policy actor as it has gained more competences over time. Being awarded the power of consent to most international agreements has been a particularly important step up in terms of its role in the EU. Negotiations with third states have also helped push the European Parliament towards new possibilities for counter-terrorism cooperation within the EU. While the European Parliament does not enjoy the right to propose new measures, it has sometimes appeared to be in favour of exploring some new ideas for EU cooperation, including some emanating from the USA – a move viewed with suspicion at times by some observers.

THE COUNCIL SECRETARIAT AND THE EU COUNTER-TERRORISM COORDINATOR

The Council Secretariat is responsible for assisting the European Council and the Council of the European Union. The Secretariat, among other tasks, drafts agendas, provides legal advice, and circulates documentation. Ultimately, it is supposed to ensure continuity between the rotating six-month presidencies. The Secretariat may appear to hold a relatively modest position at first sight. Yet, it can act as an ally to the European Commission, it can attempt to persuade others and push them in a certain direction, and it has gained executive power in the intergovernmental aspects of the Union. Above all, the Council Secretariat can exercise significant influence when working in unison with the Commission, as it did in the case of the EU's counter-terrorism financing regime (Kaunert and Della Giovanna, 2010: 279–80).

This section now turns to considering the role of the EU Counter-Terrorism Coordinator (CTC). In 2004, following the Madrid bombings, the EU decided

to appoint a CTC, with the first incumbent being Gijs de Vries, a Dutch politician. Yet, several problems have plagued the EU's position, not least the fact that the EU is an organization of (now) 27 member states, all of which have had different experiences of terrorism, making it inevitable that some member states will be more concerned about terrorism than others (Bures, 2010a; Monar, 2007a). This starting point puts the post-holder in an unenviable situation, before looking at the restrictions on the position. Since the appointment of the first CTC in 2004, it has generally been assumed that the Coordinator is ineffectual, with no mandate; a lack of real power to force member states to cooperate and a general dependence upon their good behaviour. Also, in the case of de Vries, he was considered to be inexperienced when it came to intelligence matters and to have insufficient knowledge of the internal workings of the EU (Bossong, 2008a; Keohane, 2005; Mukhopadhyay, 2004; Zimmermann, 2006). The weakness of the position appeared to be confirmed on the grounds that de Vries resigned in March 2007, with the post being left vacant for six months and rumours abounding that the post itself might thereafter be dropped. However, Gilles de Kerchove – an EU insider – took up the position in September 2007 and held it until 2021. Although de Vries is known to have become disillusioned with the lack of any real power vested in his position, Spence (2007b: 19) suggested that his resignation may also have had something to do with a European Parliament report criticizing his reticence to give information about member state compliance with the controversial activities of the Central Intelligence Agency (CIA) in Europe and the more general lack of transparency regarding his position. However, under de Kerchove's leadership, significant efforts were made to improve cooperation with certain key partners, making him active not only within Europe, but around the world.

There is no specific legislation establishing the post and setting out the Coordinator's powers. Formally, the post was established by the European Council in its Declaration on Combating Terrorism that was adopted shortly after the Madrid bombings (Council of the European Union, 2004a). As a consequence, although quickly nicknamed the EU's 'Mr Terrorism' and the 'European terrorism czar' by the media, the actual formal mandate of the EU CTC is rather general and their precise tasks and powers are not entirely clear. According to Javier Solana, the then EU's High Representative and thus also the direct superior of the EU CTC, the expectation was that the incumbent of the new post would take action in three main areas:

- To present proposals aimed at better organizing and streamlining the work of the EU Council Secretariat on the fight against terrorism;
- 2. To prepare proposals for better EU coordination between EU Councils of Ministers and their preparatory bodies;

3. To maintain regular contacts with [member states] so as to ensure coordination between EU and national action (Rhinard et al., 2007: 18).

In practice, the EU CTC has published a broad range of reports on various aspects of the EU counter-terrorism policy. Although these reports have sometimes been rather critical of both the deficiencies of EU member states and agencies involved in the fight against terrorism, at least in the latter case, the content of the report is generally an outcome of prior consultations. As a senior Europol official put it,

it is not that the CTC is saying to the outside world Europol is going to do this or that without talking to us first. And if we think it is not feasible, we tell him. It would not be handy for the CTC to announce something that is not feasible. (Interview with a Europol official, September 2009)

The Coordinator should also coordinate a plethora of other institutions and committees that have a role in different aspects of the EU counter-terrorism policy and encourage greater cooperation between the Commission (which drafts legislation) and the Council (where national Interior and Foreign Affairs ministers meet to decide EU policies). As one high-level EU official noted, the very creation of the post of the EU CTC

was about providing a visible symbol and a feeling that there needs to be somebody specifically charged with terrorism simply because the subject is spread out over so many parts of a huge and complex bureaucracy, which tends to work very effectively but also slowly, and is not always very good in making horizontal connections. (Interview with EU official A, October 2008)

The problem has been, however, that formally the Coordinator has lacked both powers and resources to coordinate the work of the Commission, Council, and the other EU agencies involved in the fight against terrorism. At the very best, according to one Commission official, 'the EU CTC can manage coordination within the Council' (Interview with a Commission official, October 2008). Moreover, some Commission officials were originally suspicious of de Vries because formally he was supposed to work for the national governments represented in the Council. As such, they feared that 'as an "agent" of the governments, de Vries would try to limit the Commission's role in EU counterterrorism efforts' (Keohane, 2005: 19). Therefore, they 'took great pains to emphasize that the work of the Co-ordinator should be restricted to the inner workings of the Council and the overseeing of implementation' (Argomaniz, 2008: 197–8). According to Argomaniz (2008: 197), this emphasis on the complementary role of the Coordinator to that of the Commission was a 'result of the Commission's instinct to protect their competencies in those counter-terror

areas that intersect with the common market and their independence from the Council's institutional influence'. Similarly, Spence (2007b: 18) has argued that the origin of the Commission's opposition to this post resides in their perception that the Council Secretariat was proposing structures that would put the then first and third pillars at the service of the second by locating policy initiative and coordination under High Representative Solana. The official view of the Commission's Directorate General (DG) Justice, Liberty and Security (JLS) Director was that he simply does 'not believe that the co-ordinator has the resources, the mandate or the institutional position to co-ordinate operational action of the various bodies in the European Union' (cited in Argomaniz, 2008: 198).

Another, and arguably even more important, role of the EU CTC has been to cajole the member states towards timely implementation of EU-level counter-terrorism agreements and initiatives. The ability of the EU CTC to do so in practice has been, nevertheless, severely circumscribed by the fact that they cannot force the national governments to act, have no independent budget, cannot propose legislation, and cannot chair meetings of national Justice, Interior and/or Foreign Affairs ministers to set the EU's counter-terrorism agenda. In fact, even at the press conference announcing de Vries tenure as the first EU CTC, the new incumbent was asked whether his mandate was sufficient, which led Zimmermann (2006: 133) to suggest that

[t]his question, again, is symptomatic of the EU's inherent, and some would say ingrained, weakness of lacking a robust executive mandate. In the final analysis, to efficiently utilize the EU's counterterrorism capacities, Mr. de Vries would have to be given the same powers on the Union level that interior ministers have at their disposal in the Member States.

One Council official commented that, in the absence of such powers, de Vries 'occasionally managed to "shame" laggard member states into speeding up their implementation processes. Otherwise, he mostly fulfilled a purely representational role towards the media and third countries' (Bossong, 2008b: 7).

Although some observers noted that de Vries successfully pushed the EU into developing some new counter-terrorism policies, the position of the CTC has not changed much since 2004. In fact, the lack of real powers and the EU member states' 'deeply ingrained dislike of sharing information on counterterrorism activities' were cited by the press as the key explanations for de Vries' decision to step down in March 2007 after only three years in the job (EUObserver, 2007). Disagreements over the exact scope of the job of the EU CTC not only prompted de Vries to quit, but also delayed his replacement. On the one hand, there was almost unanimous agreement among ministers that the new EU CTC should be a bureaucrat with established knowledge of

the inner workings of the EU, rather than a politician. This was due to the criticism of de Vries' attempts to assume an ambassadorial role that, in the view of most member states, should be reserved for national authorities. On the other hand, several EU member states had been particularly keen for the new CTC to enjoy greater powers in JHA. Their calls were echoed by the EU's High Representative for the CFSP, who suggested that de Vries' successor should have a 'deeper' mandate and enjoy 'a much closer relationship' with EU governments and institutions: '[c]ircumstances have changed since the first mandate was agreed. We have learned during this period and it is for intelligent people to adapt to the new circumstances' (Euractiv, 2007). Some member states, however, not only opposed the idea of a more politically visible CTC, but were actually in favour of abolishing the post altogether (Argomaniz, 2008: 114, 204).

In the end, Gilles de Kerchove was appointed to the post on 19 September 2007, after a summer full of failed or foiled terrorist plots in Germany, Denmark, and the UK, which once again revived political interest in counter-terrorism. As an EU law professor and the director for JHA within the Council Secretariat, de Kerchove had already personally managed much of the EU's counter-terrorism agenda and was also a choice that addressed most of the aforementioned concerns of the EU member states. This was confirmed in an interview with a Council official who argued:

There was a shift of priorities between the two coordinators. The first coordinator was a former MEP and thus was clearly someone who was used to having a very public role at a high level. His successor is someone whose main prior experience was in the Council bureaucracy in JHA and is therefore more experienced in getting the bureaucratic machine fully engaged and this is where the current focus lies. (Interview with EU official A, October 2008)

Formally, de Kerchove's mandate remained identical to that of his predecessor – he was supposed to 'coordinate the work of the Council of the EU in the field of counterterrorism, maintain an overview of all the instruments at the Union's disposal, closely monitor the implementation of the EU counterterrorism strategy, and ensure that the Union plays an active role in the fight against terrorism' (European Union, 2007a). This was also confirmed in his first public interview, where de Kerchove stated that his primary role was 'to coordinate the work of the various Councils in the field of counterterrorism' and 'to ensure that the Commission and the Council work together in this area' (European Union, 2007b). In practice, according to EU officials in Brussels, de Kerchove concentrated on more low-key coordination within the Council and followed a more hands-on approach than his predecessor on information transfer within the Council. His background in JHA also shifted the attention

of the post towards the internal side in contrast with de Vries' emphasis on the external dimension of counter-terrorism (Argomaniz, 2008: 207).

Overall, it appears that despite some calls to the contrary, the role and powers of the EU CTC are likely to remain rather limited. Moreover, the very future of the post was actually in question prior to the Treaty of Lisbon, raising questions about its overall usefulness. The treaty established the new post of High Representative of the Union for Foreign Affairs and Security Policy, thus merging the responsibilities of the then High Representative for the CFSP and Secretary General of the Council of the EU with those of the EU Commissioner for External Relations, which 'raises interesting questions regarding the future of the Counter-terror Coordinator. The position is a creation of Solana's cabinet, but it is questionable whether the new HR will keep Solana's practice of special representatives' (Argomaniz, 2008: 114, 211). Having established the mandate and possible contribution of the CTC to EU counter-terrorism efforts, this section now turns to examining the responsibilities of the Coordinator in the external dimension of counter-terrorism, as they are outlined in EU documents.

The External Responsibilities of the EU Counter-Terrorism Coordinator

It was previously stated that the EU CTC's general responsibilities are not entirely clear. However, based upon their actions and EU documents, one can identify certain responsibilities that have been assumed by the Coordinator over the past few years. Upon taking on the position, the main focus of de Kerchove's tenure as EU CTC became internal counter-terrorism efforts, whilst the implementation of existing counter-terrorism measures became a specific area of attention. This focus on the internal is not unsurprising given the problems that have arisen with EU counter-terrorism legislation, such as member states being slow to transpose EU law into their national law. However, this internal action does not preclude external action; in fact, the EU has become increasingly active on external counter-terrorism issues over the last ten or so years. The Coordinator has at times even become the EU's face in the presentation of its counter-terrorism policies to the outside world (Bures, 2011: 172). Perhaps most notably, this includes the UN, where he has appeared several times to brief the UN Counter-Terrorism Committee that oversees the implementation of UN Security Council Resolution 1373 (Eling, 2007: 113). This may have also prompted the UN to entertain the idea of establishing a counter-terrorism coordinator of its own. Even so, many caveats exist regarding what the EU is able to do internationally. Perhaps the most high-profile counter-terrorism actions by the EU are its agreements with third states – such as with the USA, Canada, and Australia – and technical assistance projects in countries across the world, but it must be remembered that the EU is not able to undertake military missions and is restricted to the personnel that the member states will provide for any kind of action abroad.

According to a UK House of Lords report, the EU CTC is responsible for the following:

- 1. Coordinating the counter-terrorism work of the Justice and Home Affairs Council (including a multitude of working groups and working parties);
- 2. Maintaining an overview of the relevant EU instruments in this area;
- 3. Ensuring effective follow-up of Council decisions;
- 4. Monitoring the implementation of the EU Counter-Terrorism Strategy, including making reports to the Council;
- 5. Fostering better communication between the EU and third countries;
- 6. Ensuring that the EU plays an active role in the fight against terrorism as a whole (House of Lords, 2011).

Numbers five and six in this list are of particular interest because they appear to comprise the CTC's external responsibilities. However, these responsibilities are in no way specific about what the Coordinator can do. The CTC does not have any power to make agreements or to initiate projects in third countries, but it is expected that they will improve counter-terrorism cooperation with third states and that they will ensure that the Union plays an active role in the fight against terrorism, which provides them with the scope to undertake a broad range of possible, but non-binding, actions. In terms of practical actions undertaken by the EU CTC in 2011 (for example), de Kerchove held discussions on terrorism with Russia and India and was also involved in an Asia-Europe Meeting (ASEM) and a major international UN conference in Riyadh (Council of the European Union, 2011b). From this, it is evident that the EU CTC has become an increasingly important voice in EU counter-terrorism activities abroad. Therefore, while it is difficult to detect actual growth in cooperation between the EU and these third countries, it is clear that other countries do recognize the CTC's position and will discuss terrorism issues with them. This recognition by, and interaction with, third states is an important part of the CTC's role in EU counter-terrorism. In the same report, we see that the EU has launched a strategy for security and development in the Sahel and that the CTC visited Algeria in order to discuss this strategy (Council of the European Union, 2011b). Clearly, in this instance, the EU CTC was involved in detailing the counter-terrorism aspects of the strategy and articulating them to the third states in the area. In other words, the CTC is involved in discussing the EU's counter-terrorism actions with third states and in the formulation of EU counter-terrorism activity. In this case, he also worked closely with the European External Action Service Senior Coordinator for the Sahel, showing that the CTC cooperates with other institutions (Council of the European Union, 2011b).

Furthermore, the EU CTC has worked closely with the UN, going beyond simple communication. For example, the CTC has engaged with UN agencies, specifically the Counter-Terrorism Committee Executive Directorate (CTED), to deliver technical assistance to third countries (Wennerholm et al., 2010: 10). According to one interviewed EU official, although the Office of the EU CTC does have links to many multilateral organisations,

probably the closest relationship exists with the UN Counterterrorism Executive Directorate, in part because we are sort of similar business and also because for the CTED, the EU is an interesting partner because we can do more with 27 states than they can do with 192 states. We are also significant donors to the UN system so we can be of significant assistance. (Interview with EU official A, October 2008)

Evidently, the CTC lacks the ability to perform any executive tasks or to force member states to do anything, but they can coordinate activities and bring the EU and third states closer together on terrorism issues; help the EU and the member states formulate counter-terrorism action in certain areas around the world; and deal directly with the UN, including regarding the provision of technical assistance

THE EUROPEAN EXTERNAL ACTION SERVICE

The Treaty of Lisbon, through its article 27(3), established the European External Action Service (EEAS). It serves as a platform for EU diplomacy to support the functions of the High Representative and Vice-President (HR/ VP) of the EU and, in parallel, is the primary liaison between the diplomatic services of each member state, embodying therefore the key instrument for future European diplomacy actions (European Council, 2010). Following the events of the Arab Spring, the EEAS entered the history of the EU as one of the fastest services to be designed and created (Drieskens and Van Schaik, 2010: 7; Helwig and Rüger, 2014: 6). It enables a more consistent and harmonised EU external coordination with other EU policies, including member states, guaranteeing the effective functioning and coordination of European external action. While the Treaty of Lisbon was one of the most transformative constitutional pieces for the counter-terrorism agenda, introducing key amendments and creating new core actors to enhance coherence in the objectives of the EU's external action, it also generated more ambiguity and sometimes function overlaps, particularly with regard to decision-making between the different actors in counter-terrorism (Ciascai, 2011; Balfour et al., 2015). Therefore, it is

of particular interest to examine the dynamics amongst the institutional actors of EU counter-terrorism. This section focuses on the role of the EEAS.

The position of High Representative for the Common Foreign and Security Policy and Secretary-General of the Council was created by the Treaty of Amsterdam. Despite the review of this role by the Treaty of Nice, their functions were still contained and somewhat limited – supporting the Presidency and the Council in matters of common foreign and security policy and, whenever necessary, assuming a diplomatic role on behalf of the Council at the Presidency's request and building bridges with third parties (Amadio Viceré, 2015: 4). However, Javier Solana's appointment to the post brought about a dynamic that would informally transcend the provisions of the Treaty (Helwig, 2015: 90; Denza, 2012: 484; Amadio Viceré, 2018). Throughout his mandate. Solana balanced the role overlaps between his tasks as HR and the duties as the External Relations Commissioner. In 1999, he publicly endorsed the idea that the EU, as an important global actor, should be given a permanent seat on the UN Security Council. This caused some friction amongst some member states that quickly came forward against this unapproved EU policy (Bates, 1999; Denza, 2012: 482). In 2000, Chris Patten, EU Commissioner for External Relations, publicly criticised him for interfering in his own sphere of influence during the Balkan crisis:

The creation of the High Representative of the CFSP, which also assumes the function of secretary general of the Council, has not contributed to resolving the tension [that exists between the Governments of the Fifteen and the community powers]. In fact, it has given rise to new institutional complications [...] It could also have increased the tendency of the CFSP to usurp functions incumbent on the Commission. (*El País*, 2000)

To overcome this institutional conundrum, the Treaty of Lisbon established the role of the HR and VP of the EU, creating an effective platform for a new institutional approach to this policy at institutional level and smoothing over the inter-pillar incoherence. The role of the HR thus translated into a new institutional approach by merging and reinforcing the functions previously held by the High Representative for the Common Foreign and Security Policy, the Secretary-General, and the Commissioner for External Relations. The HR/VP was conceived not only as the EU's diplomatic emissary, but also as a mediator with intergovernmental and supranational responsibilities (European Commission, 2012b: 29).

In the pursuit of a coherent institutionalization in counter-terrorism, in addition to now combining the HR functions set out in the Treaty of Amsterdam with those of the hitherto Commissioner for External Relations, the Treaty of Lisbon also renamed the post 'High Representative of the Union', while adding the important role of Vice-President of the Commission. Moreover, the second

HR/VP was also put in charge of the Presidency of the Foreign Affairs Council (FAC), a function previously assigned to the member state presiding over the rotating presidency. In short, the new superposition of the HR/VP, in addition to becoming a bridge between both EU institutions, now has direct responsibility for all the different dimensions of EU external relations (Bretherton and Vogler, 2006: 168-9; Vanhoonacker and Reslow, 2010: 2; Piris, 2010: 244). Assuming that one single individual would assume the same three functions, colloquially defined as 'hats' (Piris, 2010: 243; Griller and Ziller, 2008), has led many scholars to characterize this position as a true mission impossible (Euroactiv, 2009; Dietrich, 2019). The amendments to the functions of the HR have been introduced by Articles 18 and 27 Treaty on European Union (TEU). They include a significant coordinating role, while introducing a hybridization of HR/VP functions that oscillates between the supranational and intergovernmental arenas, with the aim of enhancing the robustness of institutional coherence between CFSP actors and other member states in the fight against terrorism. From a supranational standpoint, with regard to CFSP, and under the umbrella of 'ensuring the unity, consistency and effectiveness' of Union action, the HR/VP additionally has the right of initiative, both in High Representative matters (CFSP) and as Vice-President of the European Commission on matters of external relations managed by the Commission (De Ruyt, 2005: 15). The role of HR is also extended and now encompasses the power to conduct ministerial political dialogues with third countries, not only on CFSP issues, but also acting and actively participating as a policy entrepreneur in the Commission's external relations (Amadio Viceré, 2015, 2016). Concurrently, the HR/VP also heads the Union's Special Representatives (EUSRs), while also having the possibility of attending the work of the European Council, despite not being a member or being able to vote (Denza, 2012; Helwig, 2015; Helwig and Rüger, 2014). The wide range of HR/VP representation functions also embody the roles concerning 'common foreign and security policy issues' and, additionally, the task of representing the EU in international organizations and international conferences, including the United Nations Security Council. Lastly, the HR/VP also heads the negotiation of 'international agreements relating exclusively or principally to CFSP matters on behalf of the Union', aside from also shouldering the consultation and information obligations on CFSP matters towards the European Parliament. Given the transnational dimension of terrorism, the post-Lisbon era gives the HR/VP opportunities for actively participating in CFSP proposals, opening up the possibility of further extending the dialogue on such delicate matters as the strategy against terrorism, the fight against organised crime and fraud, or even discussing de-radicalization approaches with third countries. This competence, which previously lay solely with member states, enables a whole new externalization of EU security. The EU agenda for counter-terrorism has been markedly event-driven and the fact that it has always been shaped by precipitating security events stressed the need to ensure an articulated relation especially among core EU actors, namely between the HR/VP and the CTC. However, if one looks more closely at the spectrum of its functions one may arguably find that many of the CTC functions overlap with those of the HR/VP, as outlined in the previous section (MacKenzie et al., 2013a: 328–9).

With regard to the links between the EEAS and the implementation of the counter-terrorism agenda, the EEAS now controls five external relations committees and working groups of the Council. One of the most visible effects of this post-Lisbon organization was precisely the suppression of the biennial rotation of the Council Presidency (Wouters and Chané, 2016: 313). Moreover, the EEAS, led by the HR/VP, also plays a part in counter-terrorism in the Council Working Group via the Director of the EEAS Security Policy Division, as well as in the Standing Committee on Operational Cooperation on Internal Security (COSI) and the Working Party on Terrorism – International Aspects (COTER). Finally, the EEAS actively cooperates with COSI (that gathers the Ministries of Interior and/or Justice of each member state, Commission and EEAS representatives, as well as Europol, Eurojust, Frontex, and the European Union Agency for Law Enforcement Training (CEPOL)) through the European Union Intelligence and Situation Centre (EU INTCEN), whose role is to provide non-public information to the EU HR/VP and the member states aiming to advise the EU decision-making bodies in the fields of CFSP and CSDP and counter-terrorism (Plater-Zyberk, 2017; Gruszczak, 2016a, 2016b).

The role of the EU HR has celebrated 20 years of existence, having gone through a number of politically critical moments, particularly those that have shaken the security of the EU as a whole. During that time, three officials have held the position of High Representative of the Union, all of them with different profiles, all marking the post at the institutional and instrumental levels, but above all revealing the entrepreneurial power of the HR/VP. For that reason, it is important to seek to understand how the performances of Javier Solana, Lady Ashton and Federica Mogherini have oscillated between diplomatic officer, mediator, and policy entrepreneur.

Javier Solana will retain the feat of having been the first to answer Kissinger's question: 'What is Europe's phone number?' (Reuters, 2009). Javier Solana's mandate is intrinsically linked with the seminal counter-terrorism institutionalization and policies in the EU as it was during his tenure that the Madrid bombings in 2004 and London in 2005 took place. The 9/11 attacks had obviously influenced the course of action of EU policies against terrorism but, in truth, between the creation of the CTC and the Madrid terrorist attacks, Europe was largely characterized by inertia when it came to its counter-terrorism policy agenda. The terrorist attacks against Madrid were therefore a wake-up call to

develop more robust institutionalization and more coherent counter-terrorism policies and tools. As the Madrid attack was seen as targeting European values, during this period all the steps forward were rarely questioned. George W. Bush's earlier global call for a united response against the transnational 'evil' triggered the initial adoption of external and internal security frameworks (de Vries, 2008; Kaunert and Léonard, 2012: 423; Argomaniz, 2009a). Solana personally engaged in an action plan on combating terrorism brought forward by the European Council and, despite the limitations of his mandate under the Treaty of Amsterdam, acted not only as a mediator between the Council and the Commission, but also retained the merits of policy entrepreneur in the aftermath of 9/11. Javier Solana authored the European Security Strategy adopted by the European Council in 2003, identifying not only the globalized nature of terrorism, but also breaking old paradigms, namely in how to address these 'dynamic' threats (European Council, 2003a, 2003b). The number of speeches and media engagement is quite significant during this period. Solana personally engaged not only in designing the counter-terrorism strategy, but also in actively seeking to get public support for his political counter-terrorism strategy (Bates, 1999; El País, 2000; Irish Times, 2003; Solana, 2004).

The years 2004 and 2005 were heavily marked by a trigger-reaction policy period. After the Madrid and London bombings, Javier Solana created a new coordinating role, designating Gijs de Vries as CTC whose mission was to coordinate the Council's counter-terrorism tasks, supervise the implementation of the EU Counter-Terrorism Strategy, and play the role of EU mediator in the fight against terrorism, under the pillars of 'prevent, protect, pursue and respond', introduced by the EU Counter-Terrorism Strategy in 2005 (European Council, 2005). In 2007, Gilles de Kerchove replaced de Vries after three years of limited activity. Solana, once again, stepped forward, announcing that the new CTC would have enhanced functions and would be working in synergy with other EU bodies in order to effectively put in place the Counter-Terrorism Strategy. This period marks a new stage for the EU security agenda where a conjoint vision for the external dimension of freedom, security, and justice set in motion the prospective actions within this framework. In terms of intelligence cooperation. Solana was also involved as a policy entrepreneur bringing forward the Joint Situation Centre to start producing intelligence-based classified assessments. At his request, the Council of the European Union agreed in June 2004 to establish within the existing Joint Situation Centre (SITCEN) a counter-terrorist cell that would later become the EU INTCEN (Keohane, 2005: 15; Bossong, 2013: 77-8). He had previously shaped this outcome through several proposals to the Council and discursively in released summary remarks on terrorism and intelligence cooperation (European Council, 2004b). However, 'his biggest achievement was the development of his supporting bureaucratic organisation (...) When he was appointed in 1999, he had some seventy civil servants working for him. Ten years later when he left office, there were approximately 600–700 civil servants in Brussels directly reporting to him' (Dijkstra, 2011: 66).

Catherine Ashton's mandate will be forever linked to some distinctive features: she was the first HR/VP with the new double-hat formulation; she set up the entire EEAS from scratch in the midst of a European economic crisis; and she began her mandate during another trigger-event, namely the Arab Spring. Despite being considered by many a weak mandate without much impact, in truth Catherine Ashton had to deal with the fact that she was succeeding a charismatic leader who was also the architect of the entire post-Lisbon CFSP structure. Ashton's main task was to set up the EEAS and perhaps because she was an 'accidental diplomat', her diplomatic and negotiating experience was weak (O'Connor, 2010). Ashton began her mandate at about the same time as the Arab Spring and just as new warnings regarding a growing terrorist threat began to be voiced. The instability caused by the Arab Spring uprisings opened up possibilities for al-Oaeda to look for new niches in North Africa and strengthen the organization. The MI5 Director-General made it public that there were already signs of self-organisation in the UK and that there were already some British Jihad candidates. They also warned that there was a possibility that some would return to the UK and that this could pose a threat to the European territory. Europol's own report also stated that the Arab revolutions and the economic crisis could increase the risk of terrorist attacks in the EU by Islamists (Hopkins, 2012). Gilles de Kerchove also warned that the Arab Spring could mean enhanced danger and that one of his concerns was the 'dismantling of the security services in Tunisia and Egypt' provoking 'a security vacuum', which led him to call for EU support, saving 'that's where I see an urgent need for the EU to help' (Sommers, 2011). In northern Mali, radical Islamic groups closely linked to the terrorist organisation aQIM controlled part of the territory. This was a significant warning of the EU's southern neighbourhood's instability. Consequently, the EEAS released a new strategy focused solely on the Sahel where it assumed a mid-to-long-term intervention in order to counter terrorism and radicalization at the source (EEAS, 2011).

From the beginning of her term, Federica Mogherini advocated active debate and effective investment in counter-terrorism policies. At the end of her term, Mogherini could be satisfied in having actively promoted all the functions under the 'hats' she wore (Dietrich, 2019). Her mandate, similarly to what happened during Solana's tenure, was also marked by several terrorist attacks on European soil, such as those against the *Charlie Hebdo* magazine and the Bataclan theatre in Paris, as well as the attacks in Brussels, Nice, Manchester, London, Barcelona, and Berlin, in addition to the biggest political challenge the EU has faced in recent years, the 'refugee crisis'. The exercise of Mogherini's mandate differed greatly from others, namely through the

vehicles she used to communicate with citizens and their counterparts. All the diplomatic machinery was oriented towards new technologies and statements were often made in advance on social platforms in order not only to spread the message faster, but especially to act autonomously. And this translated into a significant number of statements between 2014 and 2019. Actually, HR/VP Mogherini made more statements on terrorism and counter-terrorism than any of her predecessors.

As with Solana, Mogherini somehow took advantage of the fact that there was unanimity against the terrorist attacks on European soil. This enabled her to become a policy entrepreneur in ways that had not been widely imagined before, reinforcing the spectrum of her message and thus shaping the member states' own political decisions and policy action in the EU. Particularly, she actively worked towards the better integration of the internal and external dimensions of security and highlighted the importance of the implementation of the European Agenda on Security linking the external and internal dimensions of the security realm (Aspenia, 2015). In her speech at the EU Global Strategy Conference in Barcelona, she mentioned terrorism as an example of the need to bridge the 'different dimensions of action', as one of the key issues for the EU internal—external security nexus (Mogherini, 2015b), which must be outlined under a consistent and responsible engagement framework:

A starting point for our work is the close link between our internal and external policies and the effect this has on our security. (...) I will focus on two main principles. One is consistency, the other is responsible engagement. Consistency means that our external and our internal action must be guided by the same interests and the same values. (Mogherini, 2015b)

If for Solana the turning point had been the Madrid terrorist attacks, for Mogherini the paradigm shift was the *Charlie Hebdo* attack. In addition to joining the joint march in Paris, with the heads of state and government, she became committed to including the issue of counter-terrorism on the agenda of the subsequent Foreign Affairs Council Meeting where the core statement was that '[for] the first time, there was a real awareness that we need to work together to fight a common threat. We took a decision within the Foreign Affairs Council to cooperate in a much more active way than it has been the case so far'.

Moreover, as a demonstration of real commitment and need to build synergies, Gilles de Kerchove (CTC) was invited to attend this meeting (Mogherini, 2015a). After the Paris attacks in 2015, the FAC set a point of departure to a novel approach to terrorism. As Mogherini put it:

I have to say that, probably for the first time, there was a real deep awareness of the need to work together and – I was surprised to know for the first time – the Foreign

Affairs Council had a discussion with our EU Counter terrorism coordinator Gilles de Kerchove. I think this was not only a good exchange of views but also discussions on some decisions to take that is obviously only the first step. (Mogherini, 2015a)

Additionally, not only as a policy entrepreneur, but also as a bridge-maker bringing together the Council, the Commission, the Parliament, and the member states, she stated skilfully that:

We have re-invited the European Parliament to work on the Passengers name record, the PNR. This is not the competence of the Foreign Affairs ministers, but still they are politicians and members of the national cabinets; so our Council gave a strong call for moves in that direction, as well as the need to increase our work preventing financing of terrorist networks. (Mogherini, 2015a)

In one single statement she managed to unite the different EU bodies towards the same objective: to fight terrorism. If it is true that the terrorist threat has been changing in shape and techniques, it is also true that it changed more rapidly during Mogherini's mandate. From the overwhelming nature of big terrorist attacks, Europe is nowadays increasingly used to the diffuse and pervasive nature of lone wolf attacks that were predominant during the last two years of her tenure.

Moreover, for the last three years, the threats were also of a hybrid nature calling for different actions. She engaged in a robust dialogue not only with different agencies, but also underlined the need for a global response to a transnational dimension of terrorism. Countering cyber-threats has become one of the core areas of counter-terrorism within the agenda of EU counter-terrorism and, as such, there has also been much progress by the HR/VP in this area. Her policy entrepreneurship has been increasing on both internal and external levels, notably in relation to NATO:

We are all affected, in Europe and beyond, and we all need to mobilise (...) We cannot do this alone, and we are not doing this alone. Our coordination with member states – starting from the digital one [Estonia] is constant, and we have strong and important partnerships, starting with NATO (...) We got to a stage of cooperation with NATO, especially in this field, that has no precedent in our history. (Mogherini, 2017; EEAS, 2017)

Mogherini has been actively working towards greater harmonization and joint synergies between the supranational and intergovernmental spheres that characterize the EU decision-making process, especially with regard to counter-terrorism. Perhaps the most consolidated incarnation of her vision is the European Union Global Strategy, which points towards a stronger Europe (EEAS, 2018) where, among other issues, counter-terrorism appears as a cornerstone to sustain the EU's external resilience (EEAS, 2018). On the external

level, Mogherini has been working to build as many partnerships as possible. Today, the outreach of the EEAS is quite remarkable. The launch of the EU Global Strategy has not only strengthened cooperation with the EU's neighbourhood, notably the Middle East, North Africa, and the Western Balkans, but also with the Gulf Cooperation Council and the Arab League, which has also been strengthened by the posting of counter-terrorism and security experts across EU delegations and CSDP missions.

CONCLUSION

This chapter has identified and examined the respective roles of the EU institutions in counter-terrorism, namely the European Commission, the European Parliament, the Council Secretariat and the EU CTC, as well as the EEAS. It has shown that EU institutions played a significant role in EU counter-terrorism. From 9/11 onwards to the more recent terror attacks in Paris, Brussels, Berlin, Barcelona, Manchester, and London, they have channelled EU counter-terrorism towards 'European' – rather than 'national' – policy. This is clearly suggestive of increased supranational security governance. These changes have been facilitated by exogenous shocks like September 11 and subsequent terror attacks, as well as the policy entrepreneurship displayed by EU institutional actors.

4. Counter-terrorism in the transatlantic relationship

The transatlantic relationship is of crucial importance to the EU.¹ The USA has encouraged the integration of Europe through, among other things, the Marshall Plan of 1948, has traditionally underwritten the security of the region through its leadership of the North Atlantic Treaty Organisation (NATO) since the end of the Second World War, led the 'war on terror', and remains the world's most powerful state.

Despite occasional disagreements, European states and the USA have tended to share broadly similar world views and values surrounding the desirability of a liberal world order and a preference for democratic political systems. Hill (2004: 145) has summed up European views on the USA as follows: '[f]or all the anti-Americanism that is to be found in parts of western Europe, and widespread hostility to right-wing Republicanism, democratic Europeans have not forgotten American sacrifices in two world wars, and know which side they are on in any conflict between pluralism and theocracy'. Indeed, links between European countries and the USA run deep. The EU and the USA cooperate on a vast array of issues ranging from trade to counter-terrorism. One can also argue that it would be mutually beneficial to the EU and the USA to deepen their cooperation in future given the significant shifts that are currently taking place globally, including the rise of China and the concomitant relative decline of the USA on the international stage. By cooperating with the EU, Washington has conferred a certain status on Brussels. It has significantly contributed to establishing the EU as a recognized security actor and has signalled this new status to others. The European Commission (2018a) has stated that '[t]he European Union and the United States have the largest bilateral trade and investment relationship and enjoy the most integrated economic relationship in the world', together accounting for about half of world gross domestic product (GDP) and roughly one-third of world trade. For its part, and even though there have been some tensions in transatlantic relations at times – and most recently under the Trump administration – the USA has also benefited from a strong EU. This is epitomized by the website of the US Mission to the EU, which argues that

[a] strong, united, and integrated Europe as envisioned by the EU's founders benefits Europeans and Americans alike. Europe's security and success are inextricably linked to our own. Our economies, cultures, and peoples are intertwined, and a Europe whole, free, and at peace helps to uphold the norms and rules that maintain stability and promote prosperity around the world.

In short, both sides need each other and are in closer alignment on several global issues than the vast majority of other actors in global politics.

Against that backdrop, it is no surprise that the EU's most important and most developed counter-terrorism relationship has been with the USA, where cooperation has reached a high level of intensity, ranging from dialogues to various agreements that require the transfer of personal data. This importance has been reflected in the literature on transatlantic ties, where the security relationship - often in relation to terrorism - has become a major area of research in the last few years (Argomaniz, 2009b; Kaunert, 2010a, 2010b; Kaunert et al., 2012; MacKenzie, 2010, 2012; Monar, 2010; Occhipinti, 2010; Pawlak, 2009a; Rees, 2006, 2009; Rees and Aldrich, 2005; Ripoll Servent and MacKenzie, 2011, 2012; Tzanou, 2015; Soare, 2020). Importantly, prior to 9/11, there was almost no cooperation between the EU and the USA on security matters, to the frustration of Washington. Now, two decades after 9/11, it can be concluded that the attacks on New York and the Pentagon made possible cooperation and measures that would not have been acceptable otherwise, chiefly because of concerns of the EU member states surrounding sovereignty or possible infringements of the civil liberties of private individuals.

Thus, international terrorism and the response of the US administration to it have acted as a major catalyst for the EU's development in at least four ways. First, 9/11 created a new international environment enabling actors to suggest policy solutions within the USA and the EU. In times of crisis, great changes not otherwise thought possible can occur. Second, actors able and willing to bring about change included those considered to be neo-conservative pushing their agenda in the USA and, in the EU, the European Commission taking the opportunity to create a greater security role for the EU. Third, the EU offered its support to the USA after 9/11. The US administration produced a list of over forty demands² in response to the attacks and has continued to put direct pressure on the EU on occasion since then – for example, when negotiating the agreements over PNR and SWIFT. Fourth, against the backdrop of the 'war on terror', the EU and its member states have felt under indirect pressure to act in support of the USA. It is worth noting though that this relationship has been asymmetrical, as the USA has been the dominant partner. Indeed, the

USA has had more influence on the EU than vice versa (Argomaniz, 2009b), despite many European states being more experienced in counter-terrorism than the USA. Having said that, European support for the USA in its 'war on terror' has not been unconditional. Concern about the unilateralism of the US administration, its willingness to violate global norms under certain circumstances, and its emphasis on military methods certainly troubled a number of European leaders after 9/11. The 'Snowden revelations' about significant US telecommunication 'snooping' around the world, including on German Chancellor Angela Merkel, were not well received in Europe either (Hopkins and Wintour, 2013).

It is also important not to approach the EU as a monolith in this policy area. In the aftermath of 9/11, there were some significant disagreements amongst member states, notably with regard to the invasion of Iraq in 2003. On this occasion, the UK, Italy, and Spain chose to support the USA, while Germany and France opposed that decision. Furthermore, not all EU member states have pulled in the same direction on data protection in the context of growing data exchange – a crucial issue in the EU–US relationship, given the types of cooperation requested by the USA and traditionally higher European data protection standards. The European Commission often acted to enhance its own role and facilitated EU-US cooperation by attempting to construct the EU as a partner to the USA (Argomaniz, 2009b; Bossong, 2008a; Ripoll Servent and MacKenzie, 2011, 2012; Kaunert et al., 2012). Traditionally, the European Parliament attempted to challenge counter-terrorism measures and agreements infringing upon the rights of citizens. However, this has changed in the last few years for several reasons, including its increased powers in the form of consent to international agreements following the entry into force of the Treaty of Lisbon and the strength of the European People's Party (EPP). As a result, the European Parliament has become less critical, although this trend already had its precedents (Pawlak, 2009b; Ripoll Servent, 2009, 2010). Despite the existence of some pockets of resistance at times, the forces encouraging security ties have mostly won out and EU institutions are now probably more aligned than they were in the past. Remarkably, the EU member states, the European Commission, and the European Parliament have internalised US norms by proposing or requesting EU programmes similar to those developed with the USA, such as an EU PNR programme.

One should also note that member states remain important for the USA. Nevertheless, Washington has sought cooperation with Brussels in counter-terrorism, notably because it is more expedient to negotiate with one party than with more than 25. Thus, the EU and its institutions can now be seen as major players in Europe's relationship with the USA, alongside the member states.

This chapter is structured into two main parts. First of all, it examines the historical development of the relationship between the EU and the USA. This includes their respective experiences of terrorism and counter-terrorism. This discussion provides context for the developments following 9/11 and highlights the significance of the changes that took place after 9/11. Second, the chapter analyses EU actorness in this relationship, drawing upon the revised version of the conceptual framework for actorness, with its focus on context, coherence, and capability. Overall, this chapter makes three main points: that the EU is now a significant actor in its own right in this relationship, especially in terms of agreements concerning transfer of personal data and criminal justice; that EU institutions and agencies have been important in processes of cooperation with the USA and have helped push elements of the security relationship to the supranational level; and that the relationship has been asymmetrical, which, importantly, has suited some actors in the EU and has enabled them to achieve their objectives of greater EU-US cooperation and of a stronger security role for the EU more generally.

THE DEVELOPMENT OF THE EU-US RELATIONSHIP

The USA played a vital role in the reconstruction and the integration of Europe after the Second World War. Diplomatic relations between the USA and the European Coal and Steel Community (ECSC) were established in 1953, with the focus being firmly on trade. In the climate of the Cold War, security concerns were mainly military, and with the failure of the European Defence Community (EDC) in 1954, Europe did not seem likely to make progress on military integration. Therefore, the main vehicle for military cooperation became NATO for the duration of the Cold War – and beyond. The USA has also had historical ties with many member states individually – arguably the UK, first and foremost – whilst the 'Five Eyes' alliance remains key for the USA and the UK, alongside Australia, New Zealand, and Canada, when it comes to intelligence matters.

The end of the Cold War changed the entire structure of the international system. Russia declined, the USA was no longer allegedly kept in check by its rival, and the world appeared to be entering an unprecedented period of unipolarity. This coincided with what appeared to be a major leap forward for European integration in the form of the Treaty of Maastricht, which was signed in 1992. With the launch of the EU's CFSP, it appeared to the USA that the Europeans were – at long last, in their view – considering helping with the burden of managing global security. Washington therefore looked for closer direct cooperation with the EU throughout the 1990s: 'seeking information sharing, improved extradition of criminal suspects and the creation of a transatlantic judicial network. But the Europeans were more concerned with

generating their own patterns of collaboration and rebuffed the US advances as premature' (Rees, 2006: 38). Even so, the EU and the USA did put terrorism forward as an international challenge in the 1990 Transatlantic Declaration and the 1995 New Transatlantic Agenda (NTA) (Transatlantic Declaration, 1990; New Transatlantic Agenda, 1995).

In the 1990s, Washington viewed global terrorism with growing concern. The USA was heavily affected by the first World Trade Center bombing in 1993, the East Africa Embassy bombings in 1998, and the attack on the USS Cole in 2000, which, for the Americans, pointed to a growing and potentially global jihadist threat (Kfir, 2008; Raymonda, 2006). Even so, Washington stopped short of declaring an all-out 'war on terror' in the 1990s. By contrast, terrorism in Europe appeared to be finally declining following high levels of violence in a number of states from the 1960s onwards. For example, circumstances in Northern Ireland appeared to be edging towards a peaceful conclusion after decades of conflict. Perhaps the exception and an indicator of things to come was France's experience with the Algerian Islamic Army Group (GIA) in the 1990s (Gregory, 2003). The USA was frustrated by the lack of enthusiasm for counter-terrorism cooperation across the Atlantic in the 1990s, chiding the Europeans for not sharing the burden of global security and for being generally soft on terrorism. However, Hoffman (1999: 63) disagrees, as he argues that Europeans simply saw terrorism in a different way - as primarily a domestic issue, rather than a global threat. During the 1990s, therefore, the USA had effectively moved from focusing on the global threat of the USSR to the global threat of terrorism. Still according to Hoffman (1999: 65), this could be mainly explained by '[t]he paucity of domestic terrorism itself', which 'only accentuated America's preoccupation with international terrorism'. Furthermore, terrorism was seen as being linked to a nexus of other threats by the USA in the 1990s, including nuclear proliferation (Rees and Aldrich, 2005: 909). One caveat is important here, however. The USA did experience a number of one-off attacks in the 1990s, the most infamous arguably being the Oklahoma City bombing in 1995, which killed 168 people and injured many more (New York Times, 1995). What the USA did not experience was any high-profile, long-term campaigns of violence on the scale that several European states did. Hard lessons were learnt by the Europeans in these instances, not least that significant military force could rarely end terrorist campaigns on its own and could even become inflammatory.

The 9/11 attacks were a turning point that sent shockwaves around the world, as almost 3,000 people died in the largest terrorist attack ever (BBC, 2013a). To show how 9/11 changed US views on terrorism, Crelinsten (2009: 13) differentiates between 10 September and 12 September (2001) thinking, comparing how the criminal justice model still prevailed on 10 September, before being overtaken by the war model on 12 September, among a range

of other serious changes. All of a sudden, terrorism leapt to the top of the international security agenda, which notably jolted the EU and its member states into action. Without their knowledge, European states had acted as bases for planning 9/11, whilst several of the hijackers had passed through Europe to reach the USA. For instance, the infamous 'Hamburg Cell' that assisted with the planning of 9/11 had been based in Germany (BBC, 2005). For many political leaders, 9/11 exposed the weaknesses of the EU, and they set about enhancing the security dimension of the organisation. European states took unprecedented steps to enhance internal and external cooperation on counter-terrorism, an ambiguous and potentially wide-ranging policy area. Thus, 9/11 provided new impetus to European integration and in a new direction, although it is worth recalling that member states have often preferred coordination and cooperation instead of true integration in this area since 2001 (Monar, 2007b: 267). Even so, for the first time, European integration began to penetrate areas that were previously considered too important to the sovereignty of member states for them to consider integrating in – areas of 'high politics'. The EAW has been an excellent example of this trend (Kaunert, 2007). From then onwards, the EU's role in counter-terrorism has significantly grown, as has EU-US counter-terrorism cooperation. To summarize, this section has discussed the historical development of the EU-US relationship. demonstrated how US thinking on counter-terrorism changed after 9/11, and highlighted the ways in which EU-US counter-terrorism cooperation was given strong impetus by that event.

EU ACTORNESS IN THE TRANSATLANTIC COUNTER-TERRORISM RELATIONSHIP

As previously explained, the conceptual framework used for the analysis of the EU–US relationship comprises three criteria: context, cohesion, and capability (Brattberg and Rhinard, 2012). Context concerns the level and types of recognition by the partner in question and the extent to which circumstances have enabled EU action. Cohesion refers to agreement and consistency regarding values, procedures, and outputs, both horizontally and vertically. Finally, capability relates to the EU's authority to act and the availability and deployment of instruments. This chapter shows that the USA was ready for cooperation in the 1990s, while the EU was not until 9/11. Since then, although circumstances have changed, notably with regard to threats and leaders on both sides of the Atlantic, the context since 9/11 has broadly enabled EU–US cooperation, despite some periods of tensions. In terms of cohesion, there have been many divides over the years, but member states and the EU institutions have, overall, a lot more in common with the USA than that which divides them. In addition, internal cohesion has grown over the years. The USA has exercised significant

influence over the EU and its role has arguably compelled or encouraged cohesion in a number of instances. Finally, with regard to capability, the chapter shows that EU–US security cooperation – often motivated by terrorism – has developed to be extensive and broad-ranging. Overall, it demonstrates that the EU is now an important actor and component in a transatlantic security constellation, which plays an especially important role in relation to agreements over high-technology information-sharing and criminal justice.

Context

Recognition by the USA

The EU-US relationship has been more informal than many others, given the absence of any overarching, legal agreement, such as an association agreement, between both parties. It is instead based on the Transatlantic Declaration of 1990 and the NTA of 1995. The Transatlantic Declaration outlined a number of common goals; principles of the partnership; economic cooperation; education, scientific, and cultural cooperation; transnational challenges; and an institutional framework for consultation (Transatlantic Declaration, 1990). Cooperation was taken forward with the NTA, which included four broad objectives: promoting peace and stability, democracy, and development around the world; responding to global challenges; contributing to the expansion of world trade and closer economic relations; and building bridges across the Atlantic (New Transatlantic Agenda, 1995). It is of note that both documents contain references to a joint commitment to tackle international terrorism. However, these were primarily rhetorical and did not lead to substantive engagement on the issue at the time. There are regular summits, meetings, dialogues, and other forms of interaction that indicate the USA's recognition of the EU and its close cooperation with it. Furthermore, the Commission's Delegation to the USA was established in 1954, the US Mission to the ECSC was created in Luxembourg in 1956, and the US Mission to the EU opened in 1961. Over the past few decades, cooperation has broadened and deepened into numerous policy areas. President George W. Bush's letter of October 2001 requesting cooperation directly with the EU and developments in terms of security agreements and cooperation with EU agencies demonstrate that the USA recognises the EU as an actor in counter-terrorism.

Have circumstances enabled or hindered a role for the EU?

In the 1990s, the USA wanted to cooperate with the EU on security challenges. While some actors in the EU would happily have responded affirmatively to such requests, the pre-9/11 context did not allow for such moves. Some small internal moves by the EU in the 1990s set the stage for cooperation with third states after 9/11, such as those enabling agreements between the EU and third

states on JHA issues, as well as the establishment of Europol in 1999. Both of these developments became important later for the EU-US relationship after 9/11. The USA requested cooperation with Europol as early as the 1990s (New Transatlantic Agenda, 1995). Nevertheless, Europol's initial role was rather modest. It started limited drug-related operations in 1994, whilst its mandate did not initially extend to counter-terrorism tasks (Bures, 2008: 501). The USA attempted to push the EU forward and probably hoped to influence the direction of the EU's evolution, but cooperation was not possible at that stage. From Washington's point of view, the benefits of cooperating directly with the EU on as many matters as possible were significant. For example, the USA could deal with only one legal system that would bind all the member states. rather than having to engage with many different legal systems. However, the EU member states did not perceive any pressing reason to significantly involve the EU in counter-terrorism matters at that time. However, 9/11 brought about several significant changes. First, terrorism jumped to the top of the security agenda. Second, there was a general outpouring of international sympathy towards the USA, since 3,000 people had died in the biggest terrorist attack of all time and it would have been unacceptable for European states not to help. Even in France, often one of the greatest Western critics of the USA, the headline of Le Monde daily newspaper was '[w]e are all Americans' (BBC, 2013a; Guardian, 2006). Third, the USA responded robustly and often by coercing even its closest allies. Fourth, the apparent global nature of the threat arguably left Europe uniquely vulnerable due to its level of integration.

Prior to 2001, terrorism was primarily a national threat for most European countries. In other words, for European terrorist groups, the main target was generally the government or institutions representing the power of the state (Wilkinson, 2007: 371), although they may have had transnational links to other countries, like the USA for funding or Libya for weapons as in the case of the PIRA. However, it is noteworthy that, in the French intelligence community, it had already been observed in the 1990s that Islamist groups were becoming increasingly linked to al-Qaeda. Nevertheless, this focus on France could be at least partially explained by the close relationship between France and Algeria and its intervention in the Algerian Civil War in the 1990s (Gregory, 2003). Al-Qaeda operatives were being tracked in various places across the world, such as in Kenya and Tanzania, Yemen and Bosnia, which suggested that al-Qaeda was active all over the world. Even so, at the time, the group was arguably viewed as more of a nuisance than a global threat. More broadly, many probably underestimated how much destruction non-state actors could and might be willing to cause. Then, 9/11 made it clear that terrorists could cause immense amounts of damage with modern technology and were not just focused on a single state as their target. It is not possible to replicate the fear, uncertainty, and lack of available accurate information about al-Oaeda after 9/11, but many people saw this event as ushering in a new era. It was seen as a harbinger for a future in which huge acts of violence by non-state actors had become possible, not the one-off that it has turned out to be. The US administration argued at the time that al-Oaeda terrorists, and those who harbour them, were a threat to civilization (White House, 2002). With more measured language, the European Security Strategy (European Council, 2003a, 2003b) saw terrorism as putting 'lives at risk; it imposes large costs; it seeks to undermine the openness and tolerance of our societies, and it poses a growing strategic threat to the whole of Europe. Increasingly, terrorist movements are well-resourced, connected by electronic networks, and are willing to use unlimited violence to cause massive casualties.' An abundance of literature followed from these terrorist attacks, which often depicted al-Qaeda as a truly global threat with affiliates all over the world waiting to attack. For instance, in 2003, Schultz and Vogt (2003: 10) claimed that al-Qaeda '[had] established cells from New Zealand to India to the United States'. Similarly, David Whittaker (2003: 44) noted that al-Oaeda and its affiliates were thought to be a threat in perhaps over 50 countries. The threat posed by al-Qaeda was arguably less severe than governments believed at the time – and had notably been led to believe by the group itself (Kaunert and Léonard, 2019b). However, given the perception of a serious transnational threat, a European response was viewed as vital.

Despite differing perceptions and interpretations amongst observers as to the exact nature of al-Qaeda (Byman, 2017), it appears that al-Qaeda, its affiliates and successors had a core, alliances, and inspired, unconnected individuals (or 'freelancers') (Errera, 2005: 72-5). It was this third group that often caused the greatest anxiety because of its unknown size and the difficulties inherent in identifying its 'members'. Piazza (2009: 65) provided a useful way of categorising al-Qaeda as an 'abstract/universal group', with 'highly ambitious, abstract, complex, and nebulous goals that are driven primarily by ideology'. Bin Laden's fatwa of 1998 can be seen as a general guide to what al-Qaeda wanted under his leadership. First, they believed in establishing strict sharia religious law rule. Second, they wanted to expel the USA and other 'infidels' from the Middle East and from Muslim lands anywhere in the world. Third, they aimed to topple Muslim regimes that they claimed were betraying 'true' Islam and collaborating with the USA and its allies. Fourth, they wished to ultimately establish a pan-Islamist caliphate uniting all Muslims. Fifth, al-Oaeda had declared a *jihad* or holy war against the USA and its allies and had set up a World Islamic Front for Jihad, declaring it the 'duty of all Muslims to kill US citizens - civilian and military - and their allies everywhere' (Wilkinson, 2006: 40). Such goals are not compatible with or acceptable to governments around the world, but this should not lead to the assumption that jihadist 'terrorists' are utterly incorrigible. However, it did mean that action to combat such ideas and groups, as well as supporting the USA, were vital. After all, the aims, ideology and religious motivations behind al-Qaeda appeared to suggest that anybody anywhere could be a potential target. Later, Daesh would take up this mantle and continue the threat of transnational jihadist terrorism, albeit with a number of differences, such as having a political centre and attempting to present itself as a state ('Islamic State') with a specific territory and a population.

While the level and the type of threat have been important in terms of context, so have the US administrations. Broadly speaking, they have been supportive of EU–US cooperation, especially under George W. Bush and Barack Obama. The first Bush presidency (2001–05) was more coercive in its actions, but this did have the effect of pushing along EU activity. The second Bush presidency (2005–09) was a productive time in terms of EU–US cooperation and was less coercive in its approach to its allies. Obama continued cooperation with the EU through his presidencies (2009–17) and broadly adopted a cooperative posture. By contrast, the Trump administration (2017–21) generally attempted to sideline the EU, which meant that EU–US counter-terrorism cooperation did not significantly develop further. Some observers have pointed out that the responsibilities of large online platforms in the context of the debates on freedom of expression could be an area where both sides of the Atlantic would gain by cooperating under the new Biden administration (Bossong, 2021).

Ultimately, the context created by al-Qaeda – and later by Daesh – since the early 2000s has fluctuated in terms of threat level, but the terrorist threat has remained. Other issues have emerged, such as the financial crisis, Russian resurgence, and China's continuing growth, but terrorism has continued to be on the political agenda. As previously mentioned, EU activity has followed attacks, primarily those committed against Europe, and 9/11 and subsequent attacks transformed the focus of the EU member states on national security and pushed them towards working more together at the supranational level. Also, the need for supporting the USA and Washington's interest in developing counter-terrorism cooperation with the EU created a new environment in which EU–US cooperation was seen not only as desirable, but actually necessary for securing Europe.

Coherence

The EU, the USA, and values

Notwithstanding the existence of some disagreements amongst European states or between Europe and the USA at times, these states largely share the same values. There have traditionally been some differences between Europe and the USA, but these have often been overstated (Kagan, 2003). Also, there have been some recent challenges to transatlantic ties, including the stance

taken by the Trump administration on various global issues. Before then, the Bush years were also marked by disagreements over the conduct of the 'war on terror' and the designation of the so-called 'axis of evil', including the violation of international norms and some of the methods employed in military campaigns. Nevertheless, broadly speaking, Europe and the USA have faced many challenges together and have been deeply interconnected over the past decades. They share a continuing support for a liberal rule-based order and a preference for democracy over other forms of political system. This has been well summed up in the preamble to the 'EU–US and Member States 2010 Declaration on Counterterrorism':

Noting that the enduring EU–U.S. partnership is and will continue to be built upon shared values and the corresponding actions which give them substance and meaning; [...] and Recommitting to the proposition that respect for the rule of law and international law, including human rights law, international humanitarian law, and refugee law, are fundamental in the national and international efforts in the fight against terrorism and are at the basis of our shared action, in contrast to the acts of terrorists which undermine the enjoyment of human rights and fundamental freedoms. (Council of the European Union, 2010)

This highlights how enduring EU-US cooperation is, including in the field of counter-terrorism, whatever the political leadership on either side of the Atlantic. All EU member states felt the need to assist the USA after 9/11, especially with a view to cooperating directly with Washington, but also in terms of external action where considered legitimate in regions such as South Asia. The 2005 EU Counter-Terrorism Strategy stated that '[c]ontinuing to make counter-terrorism a high priority in dialogue with key partner countries, including the USA, will also be a core part of the European approach', the US being mentioned before any other state in the document (Council of the European Union, 2005a). More recent summaries of the strategy published by the EU have given the USA even greater importance – '[c]ooperation with the US is a fundamental component of the EU's strategy. In recent years, cooperation agreements have been reached in areas such as terrorism financing, transport and borders, mutual legal assistance and extradition. The US authorities are working more and more closely with Europol and Eurojust' (EUR-Lex, 2018). While there have been disagreements over the exact methods used to combat terrorism and the invasion of Iraq at the political level, EU-US cooperation has substantially grown over the past twenty years. There is arguably no other counter-terrorism relationship where the actors are so aligned in terms of values and preferences. If 9/11 and other threats have highlighted anything, it is how close Europe and the USA are and, many would argue, need to be. It will be interesting to see how Brexit influences EU-US relations. President Trump had voiced his support for Brexit, which had not been warmly received in much of Europe, whilst the Biden administration has been more critical. In addition, the UK used to act as a supporter of EU–US cooperation and a 'bridge' between the two when it was an EU member state. Time will tell how a new UK outside the EU decides to position itself in relation to the EU–US relationship.

Methods and outputs

Methods relate to ways of doing and achieving things, but also to procedures. The book considers the extent and the consistency of agreement on the ways of combating terrorism amongst the actors and whether outputs occur from within or outside the EU. There have been major disagreements over the methods of dealing with terrorism between the USA and European states, including the invasion of Iraq, as well as the use of controversial methods, such as torture, extraordinary rendition, and mass surveillance. Nevertheless, only the last element has been directly relevant to the EU with reference to the policy areas where it possesses competences. The most significant issue of contention both horizontally and vertically where the EU has competence has arguably concerned data protection, especially given the types of cooperation that have developed between the Atlantic partners. In the aftermath of 9/11, European states typically viewed the USA as having low data protection standards. Data transfers have therefore proven problematic in the transatlantic relationship, whilst the importance of data and the capacities for data transfer have significantly increased.

Capability

Here, the EU's legal authority to act has been embedded in each area of substantive cooperation. In this section, the legal authority of the EU to act is first outlined. Then, the process through which cooperation has emerged is discussed. Finally, the substance of the cooperation is examined and assessed. While general disagreements and trends have been covered earlier – over data protection, for instance – this section examines in greater detail specific instances and processes of cooperation that have mainly aimed to foster security cooperation. It is shown that EU–US cooperation has centred around information-sharing and judicial cooperation where the EU has competence. The section also highlights how the EU now plays an important role in transatlantic relations, notably through its agencies. EU–US counter-terrorism cooperation has the following main components: police cooperation, judicial cooperation, trade security cooperation, air and border security cooperation, as well as cooperation against the financing of terrorism.

Police cooperation

Police cooperation across the Atlantic has mainly developed as a result of the signing of an agreement between the US authorities and Europol (formerly the European Police Office and now the European Union Agency for Law Enforcement Cooperation). Europol started limited anti-drugs operations in 1994, with its role expanding into other criminal activities from then onwards. In terms of counter-terrorism, a special preparatory group was set up in 1997, before the Treaty of Amsterdam subsequently extended Europol's role to formally include counter-terrorism tasks. Europol fully began its activities in July 1999 (Bures, 2008: 501). With regard to cooperation with third states and international organisations, Europol used to have the competence to conclude two types of cooperation agreements: strategic agreements, which include threat tips, crime patterns and risk assessments; and operational agreements, which allow for the transfer of personal information, such as names, addresses, and criminal records. Once the Europol Regulation of 2017 began to apply, Europol lost the competence to sign new cooperation agreements from 1 May 2017, although the agreements concluded until then have been preserved (Léonard and Kaunert, 2021a). In addition, liaison officers from a variety of different states have been stationed at Europol, whilst Europol has also had the ability to send liaison officers to partner states. Over time, Europol has become of increasing importance in counter-terrorism (Kaunert, 2010c; Mounier, 2009). In 2010, it became an agency of the EU following the entry into force of the Treaty of Lisbon. More recently, in the context of the preparations for Brexit, it has been claimed that Europol was 'transformed (...) into a much more effective organisation' under the management of Sir Rob Wainwright, its director up until May 2018, which has been involved in over 40,000 international criminal investigations per year, and has been described as 'the jewel in the crown of EU law enforcement co-operation' (Parliament.uk, 2018). Even so, Bures (2016: 57) has warned against expecting it to become a genuine European intelligence agency in the near future.

The USA had already requested cooperation with Europol back in the 1990s. US information from EU agencies was also one of Bush's demands in his October 2001 letter to the EU. Europol and the USA then signed two agreements in 2001. The first was an operational agreement, whilst the second was a supplemental agreement on the exchange of personal data and related information. Moreover, the USA has stationed personnel from a range of departments at Europol, including the FBI, the Secret Service, and Customs and Border Protection (CBP) (Europol, 2017a). Also, Wainwright indicated in a lecture at George Washington University in April 2018 that Europol and the USA have worked together on terrorist financing and travel, which has generated 82,000 intelligence leads (George Washington University, 2018). Thus, cooperation between Europol and the USA appears to have grown over time.

Judicial cooperation

EU-US judicial cooperation for counter-terrorism purposes also developed after 9/11. It was underpinned by the signing of several agreements, including one with Europol's judicial counterpart, Eurojust, which was established in 2002 and is now known as the European Union Agency for Criminal Justice Cooperation (Eurojust, 2019a, 2019b). First of all, negotiations on a mutual legal assistance (MLA) treaty and an extradition treaty between the USA and the EU were initiated in the wake of 9/11. On 20 September 2001, the EU's Justice and Home Affairs Council decided that a series of measures needed to be taken in order to enhance cooperation with the USA on criminal matters. On the basis of the then Articles 24 and 38 of the TEU – which stipulated that the conclusion of international agreements in third pillar matters was granted to the Council and should be done on the basis of a recommendation by the Presidency – an agreement on MLA and an agreement on extradition between the USA and the EU were signed in June 2003. Even so, some member states were slow to implement these agreements. Rees (2008: 101) has noted the 'inordinate' amount of time that some member states took to pass the MLA in their own parliament, which meant that it only came into force on 1 February 2010. Nevertheless, these agreements were impressive on several grounds. First of all, they were rapidly negotiated, especially given the sensitivity of a matter such as extradition (Council of the European Union, 2009c). Of particular note in that respect was the acceptance by the USA that the death penalty would not be applied to anyone extradited to the USA on the basis of the EU-US Extradition Agreement. Moreover, these agreements modernised and supplemented existing bilateral treaties, also making extradition to the USA possible for the first time for some countries, such as Bulgaria, Latvia, Malta, and Romania (Occhipinti, 2010). In addition, the MLA Agreement updated previous bilateral agreements of this sort, but also represented the first MLA agreement between the USA and Bulgaria, Finland, Malta, Portugal, Slovakia, and Slovenia (Occhipinti, 2010). A review of the MLA Agreement was carried out in 2016. The general consensus appeared to be that the agreement has added value as thousands of requests for legal assistance were registered by both sides between 2010 and 2014, although six areas were identified for improvement (Council of the European Union, 2016).

Furthermore, the USA concluded an agreement with Eurojust in 2006 in order to enhance cooperation on serious transnational crime. Eurojust has played a role in MLA with the USA. It has been presented as a 'facilitator of casework, training and policy discussions in the context of judicial cooperation between EU Member States and the U.S.', which can add value where several member states are involved in a case with the USA, as shown by the 135 cases involving the USA that were opened between 2010 and 2015 (Council of the European Union, 2016).

Trade security

With regard to trade security, EU–US cooperation has mainly revolved around the CSI programme, which was established by the Department of Homeland Security in 2002. Millions of containers are shipped into the USA and not all used to be systematically checked for their contents. Thus, it seemed possible that containers could be used to smuggle dangerous materials into the USA, especially by terrorist groups after 9/11. Initially, the USA targeted the world's 20 busiest ports, which included 14 ports in eight EU member states. The USA made bilateral agreements with each country, stipulating that US cargo would get its own terminal and that US officials would screen the containers side-by-side with their European counterparts. This led the European Commission to open proceedings against these member states because it potentially jeopardised the customs union. At the same time, it sought a mandate for an EU-wide agreement with the USA given that external trade is a communitarised policy area (Cameron, 2007; Occhipinti, 2010). An agreement was eventually signed between the EU and the USA in April 2004. By 2010, there were 23 CSI ports in ten EU member states (Occhipinti, 2010). In 2018, the US CBP claimed to have 58 operational CSI ports around the world that now pre-screen over 80 per cent of US-bound containers (Customs and Border Protection, 2018). Evidently, it is the USA that has most benefited from this cooperation, whilst the EU – like others – has had to respond to US demands, whilst also benefiting from this increase in global security. In addition, the CSI programme has helped identify other illicit activities and has acted as a deterrent, although criminals may have adapted by choosing different means of transport or superior ways of concealing what they transport into the USA.

Air and border security

In the aftermath of 9/11, US authorities examined ways in which they could strengthen border security. In November 2001, the US Aviation and Transportation Security Act introduced the requirement that airline companies operating passenger flights to, from or through the USA, provide the US CBP and the Transportation Security Administration (TSA) with electronic access to PNR data. This refers to information that airlines and other passenger service operators collect as part of their business, such as the information required to process a booking including passengers' names and addresses and credit card details (de Hert and de Schutter, 2008). It can also include 'sensitive data', such as meal choices or health information. However, there was a significant problem with passing on European PNR data to US authorities. It contravened Article 25 of the EU's Data Protection Directive (DPD), which prevented the EU from transferring data to a country not considered to have a sufficient level of data protection (European Parliament and European Council, 1995). The

USA failed this requirement at the time – and some are still critical of data protection standards in the USA, although workarounds have been agreed (Fuster et al., 2008: 191). Thus, the US demand placed European air carriers in a difficult situation. On the one hand, if they did not pass PNR data on to US authorities, airlines could face large fines or even lose their flying rights. On the other hand, if they broke the EU's DPD and handed data to the USA, air carriers would also face fines (Argomaniz, 2009b; Brouwer, 2009; Guild and Brouwer, 2006). Therefore, significant pressure was placed on the EU to reach an agreement with the USA. After negotiating a series of requirements with the US authorities aiming to ensure an adequate level of data protection, the European Commission adopted a decision on adequacy based on Article 25 of the DPD (European Commission, 2004c). It subsequently negotiated an agreement on the transfer of data with the US administration, using as a legal basis the implicit legal power provided by Community law concerning transport and data protection (Kaunert et al., 2012: 484). The EU-US PNR agreement was approved by the Council on 17 May 2004 and subsequently signed on 28 May 2004 in Washington, However, the European Parliament, which considered that it had not been sufficiently involved during the negotiations of the agreement, sought to have it annulled by the Court of Justice of the EU. In May 2006, the latter did annul the agreement, ruling that it should not have been based on the EU transport policy, given that its main aim was to combat terrorism and strengthen security (Pawlak, 2009b: 37-8).

Consequently, this first EU–US PNR Agreement had to be re-negotiated on the basis of third pillar provisions. Thus, this time, the negotiations were led by the Council Presidency with the assistance of the European Commission (de Hert and de Schutter, 2008: 330). An interim PNR agreement – negotiated under police and judicial cooperation – was adopted in the meantime and entered into force in October 2006, which ensured similar levels of data protection as before. The Council of Ministers approved a new EU–US PNR Agreement in July 2007, which began to apply provisionally. Under this new EU–US PNR Agreement, the US administration agreed to receive fewer fields of data – 19, instead of 34 compared to the first PNR agreement – although this decrease was partially due to the merging of some categories of data (Kaunert et al., 2012: 485; Rhinard et al., 2007: 21). In exchange, the Council agreed that US authorities would be allowed to share the data with a larger number of federal authorities and to store them for longer, namely 15 years, instead of three and a half years (Occhipinti, 2008: 19).

The entry into force of the Treaty of Lisbon in December 2009 meant that the European Parliament had now the right to consent to most international agreements. In May 2010, the European Parliament withheld its consent to the 2007 agreement, which led to new negotiations led by the European Commission. The Council signed the agreement in December 2011 and the

European Parliament voted on the agreement in April 2012. Despite the rapporteur in the Civil Liberties, Justice, and Home Affairs (LIBE) committee, Sophie in 't Veld, advising its rejection, the European Parliament gave its consent to the agreement by 409 votes to 226, with 33 abstentions (European Parliament, 2012). Despite its controversial character (see EDRI, 2017), the EU–US PNR Agreement remains in place and is subject to regular reviews (European Commission, 2017a).

The European Commission (2019a) has argued that many states consider PNR data to be important for law enforcement, although not much detailed evidence for this claim has been provided to date, which could also be due to security reasons (European Commission, 2019a). Some information was provided in the context of the debates on an EU PNR system, which indicated that PNR data was solely or predominantly responsible for 95 per cent of drug seizures in Belgium in 2009, solely responsible for 65–75 per cent of drug seizures in 2009 in Sweden, whilst, in the UK, within a six-month period in 2010, PNR data enabled the seizure of over 200 kilos of cocaine and heroin (European Commission, 2011a). Relating this back to the EU–US relationship, the USA has coerced states around the world to submit PNR data to its authorities. Thus, the USA has exercised a significant level of influence over other states, including the EU member states. It has diffused its norms to the EU, which eventually established its own PNR system in 2016 (Argomaniz, 2009b; Ripoll Servent and MacKenzie, 2012).

Countering the financing of terrorism

The US Treasury instigated a Terrorist Financing and Tracking Program (TFTP) after 9/11. SWIFT, which is responsible for worldwide financial messaging and facilitating worldwide bank transfers (Fuster et al., 2008: 192) found itself obliged to cooperate with the US authorities, because, although it is a company based in the EU, it had a mirror of its data in Virginia, USA. The company is responsible for the standardised financial messages of more than 11,000 banking organisations, securities institutions, and corporate customers (SWIFT, n.d.). Furthermore, and most importantly, SWIFT possessed 80 per cent of the electronic transfer market at that time (Deutsche Welle, 2010). Crucially, these messages frequently include personal data of the payer and the payee (Fuster et al., 2008: 192), which made them interesting to US authorities in the framework of the fight against terrorism.

The TFTP system was kept secret until 23 June 2006 when the *New York Times* leaked information that SWIFT had provided the USA with access to international bank data. This came as a surprise 'almost for everybody' (Fuster et al., 2008: 194). Fuster et al. (2008: 194) pointed out that '[f]rom a European perspective, the main problem with data processing as foreseen by counter-terrorist US law is in general terms not the absence of checks and

balances to ensure data protection, but the exclusion of non-US citizens from access to such protection in the cases where it is granted'. In 2009, SWIFT began to relocate the mirror of its data from the USA to Switzerland – thus requiring the USA to request data from Europe. As a result, the adoption of an agreement between the EU and the USA became necessary.

Member states rapidly recognised the security benefits of the TFTP system. For instance, the USA passed on over 1,500 leads to European authorities, some of which have led to the prevention of terrorist attacks and arrests (Archik, 2012: 5; Occhipinti, 2010: 137). A nine-month interim agreement was concluded by the Council on 30 November 2009 – that is, one day before the Treaty of Lisbon granted the European Parliament consent over the agreement. Unsurprisingly, this move was not well received by the European Parliament, which rejected the agreement in February 2010. In addition to 'flexing its muscles' in the context of the EU's inter-institutional relations, the European Parliament was also attempting to raise its concerns about the transfer of bulk data to the USA, also citing what it perceived as US coercion as one of the reasons for the rejection of the interim agreement (Ripoll Servent and MacKenzie, 2011: 397).

As a result, the Commission had to re-negotiate the SWIFT Agreement. This time, it had to take into account the concerns of the European Parliament in relation to data protection. Without some changes being made to the agreement, it was possible that the European Parliament would once more reject it. A second – permanent – SWIFT Agreement was then accepted by the European Parliament in plenary in July 2010. It differed from the first one in several ways. It included a role for Europol as a clearing house, which had to 'green light' the transfers of data, data protection provisions were strengthened, a 'scrutineer' was to be employed by the EU for oversight purposes, whilst the idea of an EU TFTP was launched (Archik, 2012; European Union, 2010a). Moreover, as discussed by an interviewee, this agreement arguably got the EU involved in intelligence gathering for the first time (Interview with EU official B, June 2010). Although this is a development that increases the security role of the EU, this second SWIFT Agreement has been criticised by some for being actually very similar to the first (Council of the European Union, 2011a; Europol Joint Supervisory Body, 2011). A review of the agreement was also undertaken in 2017, which credited EU-US cooperation and suggested only a few minor improvements (European Commission, 2017b). This agreement is an instance of US dependence on the EU for the continuing flow of data. Of course, member states also welcome the intelligence received from the USA, making it a situation of interdependence. It is noteworthy that the European Commission decided not to pursue the development of an equivalent EU system (European Commission, 2019b).

CONCLUSION

Overall, this chapter has shown that the EU is a significant actor in the case of the EU–US relationship, although it lacks a sense of separateness from the USA, which has taken the leadership in their interactions. Importantly, relations with the USA have led other states to approach the EU to ask for cooperation along the lines of the agreements that the EU has already concluded with the USA. In this way, cooperation with the USA has opened the door to cooperation with others. In so doing, it has contributed to the EU's presence and provided new opportunities for EU action.

Since 2001, the EU has become a credible interlocutor for the USA, even though Washington has at times – depending on the policy area – preferred to cooperate with individual member states. It has also on occasion divided the EU, not only with regard to Iraq, but also when attempting to sign separate agreements with individual member states. Furthermore, disagreements over issues such as data protection have frustrated the USA at times. This appears to have become less of an issue recently, since the European Parliament has become seemingly less resistant overall to security measures emanating from the USA. It will be interesting to see the evolution of the sensitive debates on data protection on both sides of the Atlantic and the impact that this has on EU–US cooperation.

Furthermore, the development of EU counter-terrorism cooperation means that the EU now possesses a high level of capability in its relationship with the USA. Policy formulation has fewer obstacles, but there has been a significant level of dependence on ideas emanating from the USA. There was some resistance in the past from the European Parliament on issues such as data protection – and there are still some concerns in some member states in that respect – but the possibility for policy formulation is now high.

NOTES

- 'Transatlantic relationship' refers in this chapter to the EU and the USA, not including Canada.
- 2. The letter from President George W. Bush to Romano Prodi, the then President of the European Commission, from 16 October 2001 listed over 40 demands for closer EU–US cooperation following 9/11. Details can be found at: http://www.statewatch.org/analyses/no-2-useu.pdf.

5. EU counter-terrorism and South Asia

This chapter establishes and assesses the EU's counter-terrorism role in South Asia, with a specific focus on Afghanistan and Pakistan. This region has been particularly important in the field of counter-terrorism for several reasons. First, it is the place where al-Qaeda emerged and was based prior to 9/11, under the protection of the Taliban. Second, it is where the 'war on terror' began with the US-led invasion of Afghanistan in 2001. Third, it was largely viewed as the epicentre of global terrorism prior to the emergence of Daesh in the Middle East (Gunaratna and Nielsen, 2008) and jihadist groups continue to have a major presence in the region nowadays. Afghanistan and Pakistan are contiguous countries, which explains why the insurgency in Afghanistan has also caused major problems for Pakistan. While Pakistan has been a major victim of terrorist violence, it has also been a well-known sponsor of terrorist groups (Riedel, 2008). After invading Afghanistan without adequately planning for post-conflict reconstruction, the USA and its allies have enabled the Taliban and al-Qaeda leaders to find shelter in the border areas between Pakistan and Afghanistan. After years of stalemate, President Biden's decision to withdraw the remaining US troops led to a swift re-conquest of Afghanistan by the Taliban, who entered Kabul on 15 August 2021 (Bildt, 2021). Thus, two decades after 9/11 and the ensuing Afghanistan War, instability and violence still characterise the region. The hope expressed at the 2001 Bonn Conference that Afghanistan would see 'a broad-based, gender-sensitive, multi-ethnic and fully representative government' (United Nations, 2001a) has remained elusive

In general, the type of role that the EU plays in counter-terrorism depends on the situation in the country. When combined with the other case studies, this chapter demonstrates the diverse range of external tasks that the EU can now perform. Overall, the EU's external role – particularly in Afghanistan and Pakistan – has been largely overlooked by both scholars and journalists. Because of how the EU's presence expanded in countries like Afghanistan, it is difficult to conclude that the EU is still 'an absent friend' (Keohane, 2008), if it ever were. The growth of the EU's role is the most important issue to consider, as it has evolved from a modest regional security actor at best to becoming more globally active. It is questionable whether this level of commitment to global counter-terrorism efforts will be seen again soon now that the Taliban have reconquered Afghanistan. Nevertheless, that does not change the fact that

the EU's involvement in this region developed considerably between 2001 and 2021.

This chapter examines how the EU became involved in Afghanistan after the US-led war in 2001, before analysing the important contributions that it has made to security in the region since then. Almost all EU member states have contributed to the war efforts in Afghanistan, through aid or troops. The activities of the member states are not examined here, as the chapter focuses on the role of the EU institutions, even where member states have seconded their nationals to EU missions, such as with EUPOL Afghanistan. There has been little literature addressing the EU's role in South Asia in general, and even less on EU counter-terrorism activities in the two countries examined here (Duke, 2002; Monar, 2015; Rummel, 2002; Siegel, 2009; Toro, 2010; Wennerholm et al., 2010). A further limitation has been that most of that scholarship has focused on what was previously known as the European Security and Defence Policy (ESDP) and the missions that took place under its auspices. However, in general, such activities have not specifically included counter-terrorism in their mandate.

Nevertheless, it is argued here that the EU has made significant contributions to security and counter-terrorism activities in the region, which deserve further exploration. As argued by Klaus-Peter Klaiber (2007: 10), the EU Special Representative to Afghanistan between December 2001 and July 2002, the EU and its member states account for

over one-third of financial assistance and aid programmes to Afghanistan, contribute over one-third of troops to the International Security Assistance Force (ISAF), and have worked on building up some of Afghanistan's key functions, such as police (Germany), judiciary (Italy), and anti-narcotics (UK) ... [d]espite these facts, the work and contribution of the EU for the reconstruction is hardly ever mentioned in books and articles written about Afghanistan. In the last few years, the UN and a number of individual states have figured prominently in assessments by experts, whereas the EU has been noticeably absent from these reports. This is regrettable given its major contributions.

This view is shared by Cameron (2013: 38–9), who states that 'the EU has played a significant but often unrecognised role in the reconstruction of Afghanistan, both in terms of its financial contributions and institutional support, and in exerting political influence through the appointment of an EU special representative'. Thus, this chapter fills a significant gap in the literature by analyzing the EU's contribution to counter-terrorism efforts and its broader security role in Afghanistan and Pakistan.

Although the success of some of the EU's projects – especially the EUPOL Afghanistan mission – has been questioned within the media and in other circles (*New York Times*, 2009; EUObserver, 2010), this chapter does not specifically

focus on the issue of success. In fact, the success of all actors that operated in Afghanistan – not just the EU – can be seriously questioned, especially since the Taliban's return to power in 2021. It is very unlikely that democracy will flourish in Afghanistan in the coming years. However, the focus of this book is the expanding role and actorness of the EU in counter-terrorism, which has seen a significant strengthening since 2001. Frequently, expectations of EU action have been raised too high. Considering the evolution of the EU in combination with an appreciation for what it is and what it can do can lower expectations, which allows for a more realistic evaluation of EU activity, notably in the field of counter-terrorism – hence, the choice of an analytical framework centred on 'actorness'.

This chapter is structured into two main sections. First, it analyzes the development of the EU's activities in and cooperation with Afghanistan and Pakistan in the field of counter-terrorism. Second, the actorness framework is applied to explore the context, the capability, and the coherence of the EU's counter-terrorism activities related to Afghanistan and Pakistan. In so doing, the chapter develops three key arguments. First of all, the EU came to fulfil several important counter-terrorism-related functions in Afghanistan and Pakistan – particularly in the traditionally 'softer' areas of security, notably capacity-building and funding of national and international projects. Those have been significant, albeit often underappreciated, projects. Furthermore, the evolution of the EU's role in the region, as well as the range and the scope of the activities undertaken, has been impressive by EU standards. Finally, this chapter aptly illustrates how the EU became a significant counter-terrorism actor on the global scene in a short time period following 9/11 and in contrast to its traditional focus on internal matters or on its geographical neighbourhood.

THE EU'S RELATIONS WITH AFGHANISTAN AND PAKISTAN

This section outlines the evolution of the relations between the EU and Afghanistan and Pakistan up to 2001, as well as the security environment and experiences of terrorism within these two states. The EU was primarily a provider of aid and an important trade partner to both countries prior to 9/11. However, these terrorist attacks thrust the EU into developing a broader and heavier presence in the region. The EU had been active in Afghanistan from the 1980s onwards, mainly as a provider of aid, which began during the Soviet–Afghan War. This aid continued until the late 1990s, when the Taliban made a series of unreasonable requests of non-governmental organisations (NGOs) working in the country, which led to the suspension of EU aid. As for direct EU–Afghanistan cooperation at the governmental level, it was almost non-existent prior to 2001, because of the lack of a government and the sub-

sequent Taliban regime. EU-Pakistan relations date back to 1974, but again, this relationship was mainly aid- and trade-related. Pakistan has experienced political upheavals over time and the EU has had to adapt. It is a fragile, but functioning, state. The EU has been Pakistan's most important trade partner and Afghanistan's fourth (EEAS, 2016a; European Commission, 2018b). This should have given the EU considerable leverage over both states, but the situation has been rather more complicated, as shown later in this chapter.

Afghanistan's problems date back much further than the 1970s and include conflicts over the leadership of the state and differences between its centre and periphery (OECD, 2002; 4). It has experienced serious and protracted conflicts over the past few decades. The USSR invaded Afghanistan in 1979, before withdrawing in 1989, leaving terrible destruction, including an estimated one million civilian deaths (Taylor, 2014). During the war, thousands of 'Afghan Arabs' travelled to fight against the Soviets in Afghanistan. Osama bin Laden was one of these. While others left, he stayed in the region to form al-Qaeda in Peshawar, Pakistan, in 1988, with the desire to form a 'jihadist vanguard movement' that would perform three tasks: first, to be a terrorist group in its own right; second, to organise, train and provide logistics for jihadists around the world; and third, to offer leadership, unify, and reorient the global jihadist movement (Byman, 2015: 9). The group was, of course, of greater relevance later on, after al-Qaeda's move to Sudan in 1991–92 and its subsequent return to Afghanistan in 1996. Following the Soviet withdrawal, the country slipped into civil war. This lasted until the Taliban – a group which emerged in 1994 from the country's majority Pashtun ethnic group – seized the capital, Kabul, in 1996, and took control of over 90 per cent of the country by 1998 (Rashid, 2010: 1-5). It was the Taliban who invited bin Laden back to Afghanistan, where he remained until 2001. Thus, 9/11 was planned in Afghanistan. Furthermore, Afghanistan had an entrenched drug economy, with a key role for heroin – a practice supported and extended by the Taliban (Felbab-Brown, 2017).

Resistance to the Taliban mainly revolved around the so-called 'Northern Alliance', which was supported by the US-led coalition during the invasion of 2001. However, many of these warlords were almost as religiously hard-line as the Taliban. Therefore, concern was raised over what would replace the Taliban after the invasion (Gannon, 2004: 37). In the aftermath of the Taliban's downfall, various warlords were given posts in the new government under Hamid Karzai, Afghanistan's prime minister, leading to questions of whether peace and stability were possible in a new Afghanistan or whether the new government would simply repeat the practices of the government that it was supposed to replace (Gannon, 2004).

The EU provided significant amounts of aid to Afghanistan during the 1980s and 1990s through the European Commission Humanitarian Office (ECHO)

based at Peshawar, Pakistan. However, in 1998, the EU suspended all aid to the parts of Afghanistan under Taliban control after it made a series of unreasonable demands to NGOs that were providing assistance, such as requiring all female UN aid workers to be chaperoned by a blood relative and requesting that all NGOs working in Kabul move their offices to ruined buildings outside of the city, which were formerly the polytechnic college (Rashid, 2010: 71–2). Until this point, the EU donated two-thirds of all aid to the country (European Union, 1998). Subsequently, the EU, along with the UN, appears to have criticised the Taliban for their activities and maintained aid sanctions until the US-led invasion of the country in 2001. Ultimately, the OECD (2002: 6) concluded that attempts to use aid incentives and disincentives to influence policy had not been successful and that Afghanistan was not a favourable environment for their use, given how far apart the donors and the Taliban were. In the end, Afghanistan was a war-torn and divided state. Western states had little influence over it with their aid, whilst it was a base for al-Qaeda and its training camps, as well as a major producer of heroin.

As for Pakistan, its cooperation with the EU began under a different set of circumstances compared to Afghanistan. Pakistan and the EU started cooperating in 1974, but only signed an official Cooperation Agreement in 2004. Since 1974, the EU has committed money to various projects and programmes in Pakistan (European External Action Service Website, n.d.). The Cooperation Agreement set out several areas for cooperation between the two partners. Particularly pertinent to counter-terrorism are Article 13 (drug precursor chemicals and money laundering) and Article 17 (evolutive clause) (Council of the European Union, 2004b). However, these are not the only forms of cooperation relevant to counter-terrorism between the EU and Pakistan. Since 2001, Pakistan was also expected to become a key recipient of EU technical assistance (Wennerholm et al., 2010: 8).

In addition, new areas for EU-Pakistan cooperation have opened up over time. Pakistan has seen the EU primarily as a trade actor. Security has not been a significant theme for Pakistan in its relationship with the EU. This is confirmed by examining the Cooperation Agreement, which holds few clauses on security cooperation, and the Europa website, which highlights that most of the money that the EU sends to Pakistan is sent for humanitarian purposes and crises, with few references to security (European External Action Service Website, n.d.). However, Islam (2013: 587) has highlighted the fragility of the state of Pakistan, as well as the fact that Pakistan has been on the periphery of Europe's Asia policy until recently, but that with concerns about terrorism and following US and UK pressure, the EU has begun to play a greater role in Pakistan. Even so, Islamabad has continued to rely primarily upon the USA and the UK for military support.

EU COUNTER-TERRORISM ACTORNESS IN SOUTH ASIA

The chapter now turns to examining the growing role of the EU in the domain of counter-terrorism in the South Asian states of Afghanistan and Pakistan. The section on 'context' shows how 9/11 led to new circumstances, in which EU actors pushed for the development of an external role for the EU in counter-terrorism, whilst Afghanistan and Pakistan were also encouraged to engage with the EU. The section on 'coherence' considers matters related to values, methods, and procedures, before questioning whether EU actors have all pulled in the same direction. Finally, the section on 'capability' analyses the growing range of activities that the EU has performed in Afghanistan and Pakistan over the course of its involvement following 9/11.

Context

Recognition by Afghanistan and Pakistan

Before 9/11, it was actually not clear by whom the EU could be recognised in Afghanistan. Although the Taliban controlled about 90 per cent of the territory of Afghanistan, the Islamic Emirate of Afghanistan that they had established was not recognized by the international community (Ibrahimi, 2017: 956). Following the US-led invasion, Afghanistan had little choice but to accept that the EU would play a role within the state because of the attempts to build up the country. One key document on EU–Afghanistan cooperation was the EU–Afghanistan Joint Declaration of 2005, which was supposed to launch a new partnership. As evidenced by the main headings in this document, it was expected that EU–Afghanistan cooperation would occur in the following areas: political and economic governance, security sector reform and justice sector reform, counter-narcotics, development, human rights, civil society and refugee return, education and culture, and political dialogue (Council of the European Union, 2005b). Thus, several areas for future cooperation were linked to counter-terrorism.

Subsequently, EU–Afghanistan cooperation was placed on a firmer footing with a first agreement in the form of the Cooperation Agreement on Partnership and Development (CAPD). It was ratified by Afghanistan's parliament in July 2017. The CAPD, among other issues, committed the EU and Afghanistan to deal with common challenges, including '(1) the fight against terrorism, international crime and illegal trafficking; (2) non-proliferation, disarmament and nuclear security; (3) Weapons of Mass Destruction (WMD); (4) Small Arms and Light Weapons (SALW); and (5) counter-narcotics' (European Commission, 2015b). It was supplemented by a new strategy on Afghanistan,

which was adopted in October 2017. This strategy emphasised that '[the] EU [was] committed to working with Afghanistan to combat terrorism, the financing of terrorism and terrorist threats in all their forms and to step up dialogue and cooperation to prevent and counter violent extremism, radicalization and recruitment and to address their root causes' (Council of the European Union, 2017: 4). Thus, EU–Afghanistan cooperation – notably in the field of counter-terrorism – developed significantly after 2001.

In contrast, the EU's formal relations with Pakistan have a longer history, as both parties had already concluded a Cooperation Agreement in 2004. This agreement set out several areas for cooperation between the EU and Pakistan, but was very limited when it came to counter-terrorism (Council of the European Union, 2004b). In the EU's post-9/11 efforts to become an exporter of technical assistance, Pakistan was expected to become a key recipient of EU support. This initially did not materialize (Wennerholm et al., 2010: 8) and it is only over time that counter-terrorism cooperation has developed between the EU and Pakistan. To highlight the expanding relationship between the EU and Pakistan across a range of sectors, June 2009 saw the first EU-Pakistan summit, which discussed a range of challenges, from trade to counter-terrorism cooperation (Council of the European Union, 2009c). A second summit took place in 2010, during which a five-year engagement programme was discussed. In 2012, the EU and Pakistan agreed on the programme (EEAS, 2012), which was followed by the 2019 EU-Pakistan Strategic Engagement Plan (SEP). The SEP covers various areas, including peace and security (which includes counter-terrorism cooperation); democracy, rule of law, good governance, and human rights; migration and mobility; and trade and investment (Council of the European Union, 2019). Compared to Afghanistan, Pakistan is a functioning state that has not had to rely on external support to the same extent. Moreover, the Pakistani government appears to view the EU mainly as a trade actor, rather than a security provider or partner. This can notably be explained by the fact that the EU is Pakistan's second most important trading partner, whilst the USA has been able to give Pakistan considerable military assistance in a way that has been beyond the EU's capabilities.

Have circumstances enabled or hindered a role for the EU?

The events of 9/11 significantly changed the environment in which the EU operates, as the 'war on terror' became the defining feature of the first decade of the twenty-first century for many states around the world. Terrorism was to be combated with much greater urgency and on a larger geographical scale than had occurred in the past. In these circumstances, new relationships had to be forged and different policy matters reached the top of the agenda. Combating terrorism globally became an important objective for the EU. Therefore, following the US-led invasion of Afghanistan and as a result of the role of

Pakistan in providing a base for insurgent groups and al-Qaeda, increasing the EU's cooperation with these two states became a major objective. The war in Afghanistan made post-conflict reconstruction necessary, whilst Pakistan was believed to require assistance in order to strengthen its institutions with a view to combating terrorism more effectively within its borders.

Thus, circumstances propelled the EU towards a greater role in global counter-terrorism efforts, with Afghanistan and Pakistan being the most important region in the 'war on terror' – at least initially. For much of the twenty-first century, Afghanistan and Pakistan have been seen as a focal point for the fight against jihadist terrorism, but Afghanistan has also been known as a major supplier of world heroin, producing perhaps as much as 90 per cent of the world's supply (NBC, 2015). Given that these significant challenges need to be tackled, South Asia can arguably be seen as providing a context conducive to EU action. However, in such challenging circumstances, the EU's role has tended to be contingent upon the continuing presence of the USA. As seen in August 2021, once the USA decided to withdraw its last troops from Afghanistan, it was no longer possible for the EU and its member states to maintain their presence in the country. The security situation has been such on the ground that the US withdrawal has not opened space for the EU to act, but rather has significantly hindered its capacity to act.

Whereas Afghanistan has been a priority for the EU, Pakistan had been on the periphery of the EU's Asia policy until the early part of the 2010s, with the true focus of EU activities being China and India as the two major states on the continent (Islam, 2013: 587). However, security links with Afghanistan, threats to Europe, and pressure from the USA and the UK have all coalesced to push the EU to step up its links with Pakistan. In sum, 9/11 created an external environment that has encouraged EU action in external counter-terrorism. Given the problems in the region and concern over leaving the region to once again become a base for terrorists, the circumstances have arguably supported a role for the EU in the region until recently. Now that the Taliban are in power again in Afghanistan, the EU can do little else but offer humanitarian aid and attempt to support rights-based groups (Hassan, 2021). As emphasised by Hassan (2021), '[the] Taliban takeover and the U.S. withdrawal have created profound problems for European policy in Afghanistan'.

Coherence

Shared values within the EU and between the EU, Afghanistan and Pakistan

The EU's involvement in these states has been centred on assisting the USA, combating the Taliban, and preventing Afghanistan and Pakistan from becoming havens for terrorist groups by building a stable Afghan state. However,

there have been divisions over the role of the USA in Afghanistan, tensions between the USA and the EU, and difficulties amongst member states due to their own foreign policy preferences, personnel, and resources. At the same time, there has been significant reluctance to go against the USA, which means that the EU has had to adhere to US actions in Afghanistan, even if those were controversial or ineffective. In this instance, coherence was arguably a greater problem in the early stages when member states were individually deployed in Afghanistan. After European troops left, the EU's position in terms of funding and running projects became somewhat easier.

Several values have drawn together EU member states and have helped sustain the EU's involvement in Afghanistan and Pakistan. The outpouring of sympathy for the USA following 9/11 was significant. In fact, it would have been difficult for the EU and its member states not to help the US in the region following 9/11. NATO invoked Article 5, the mutual defence clause, following 9/11, whilst the EU and its member states pledged to support the USA in Afghanistan. Most European states remained committed to Afghanistan for years after 2001, with many British troops not leaving until 2014. While the invasion of Iraq gave rise to a number of challenges, it did not seriously alter the commitment of European states to Afghanistan. Supporting the USA was clearly of great importance to the EU and its member states. This cannot only be explained by the historical links between the USA and Europe; the widespread perception after 9/11 that al-Qaeda constituted a global threat, which did not only concern the USA, but also Europe and many other nations, was also relevant. This was further emphasized by the Madrid bombings in 2004, which Reinares (2010) has linked back to the al-Qaeda leadership in Pakistan. Afghanistan and Pakistan have continued to be a key area of the world for the planning and execution of terrorist attacks. While not all EU member states have been equally concerned about terrorism related to Afghanistan and Pakistan, these two countries have also come under attention for their role in the illegal drugs trade.

Thus, in the aftermath of 9/11, the EU made a long-term commitment to state-building and to preventing the region from again becoming a haven for terrorists. The EU seems to have been initially driven by the hope of establishing a liberal-democratic Afghanistan. It aligned itself with the UN's ambitious objectives for the country, including its vision for the new Afghan administration, which

should be broad-based, multi-ethnic and fully representative of all the Afghan people and committed to peace with Afghanistan's neighbours; should respect the human rights of all Afghan people, regardless of gender, ethnicity or religion; should respect Afghanistan's international obligations, including by cooperating fully in international efforts to combat terrorism and illicit drug trafficking within and from Afghanistan; and should facilitate the urgent delivery of humanitarian

assistance and the orderly return of refugees and internally displaced persons, when the situation permits. (United Nations, 2001b)

However, optimism decreased over time amongst the EU member states and a more pessimistic – or perhaps realistic – assessment of the situation in Afghanistan began to take shape. Accordingly, the EU's objective became more modest. This is epitomised by the EEAS's (2016a) summary of the EU's position in 2016:

The EU has a long-term commitment to Afghanistan and its people. Its main aim is to develop Afghanistan's institutions so that the progress already made is maintained. The EU also wants to help in establishing a sustainable Afghan state.

Methods and outputs

Both the European Commission and the EU member states have made contributions to the security and reconstruction of Afghanistan following the war in 2001 (European Union, 2009). There were various reasons for this involvement, which notably included the willingness to demonstrate solidarity with the USA and amongst European states. Concerns over illegal drugs also played an important role and can be seen as having the potential to create more cohesion within the EU than terrorism. This is because terrorism has not affected all EU member states in the past, whereas drugs – heroin in this specific case – have been an issue in all EU member states, as drug routes have developed through the Balkans and from Russia into Eastern Europe (Klaiber, 2007).

Nevertheless, some EU member states did have strong concerns related to terrorism. This was particularly the case with the UK, as demonstrated by the fact that the UK's CONTEST Strategy highlights the terrorist threat linked to Pakistan and Afghanistan (HM Government, 2011: 9). As a result, the UK – when it was still an EU member state – encouraged the EU to become more involved in Afghanistan and Pakistan (Council of the European Union, 2009b). Moreover, Europol's 2011 Terrorism Situation and Trend (TE-SAT) report (Europol, 2011: 17) noted that '[v]olunteers are recruited in the EU to support Islamist activities in Afghanistan [and] Pakistan'. This demonstrates that there was some concern at the EU level about the threat posed to Europe by Islamist groups linking back to Afghanistan and Pakistan.

Given that Pakistan is a former British colony, it is perhaps unsurprising that the UK has played a particular role in Pakistan. However, even countries like Poland, with no experience of terrorism, became more aware of the threat of terrorism in Pakistan, notably when a Polish engineer was beheaded while working there (*Guardian*, 2009). Therefore, in several European countries, there was a growing awareness of the threat of terrorism to European countries emanating from this region in the aftermath of 9/11. In addition, attacks on

the West stemming from South Asia and attacks on the people in the region highlighted the volatility of these countries and added to the appreciation that both Pakistan and Afghanistan needed assistance for both their own security and that of Europe. Therefore, the member states, although they may not have all assessed the terrorist threat in the same way, have provided more output in areas relevant to counter-terrorism, which has contributed to enhancing the EU's actorness in counter-terrorism.

In terms of horizontal coherence, there have been some disagreements amongst institutions, although they have all agreed on the need for an EU presence in the region. The appointment of a first EU Special Representative for Afghanistan in 2001 demonstrated the EU's commitment to the country and its importance for the EU. However, the European Commission's external action – particularly on counter-terrorism – may have been truncated by the decision of the European Court of Justice to prevent the European Commission from over-reaching with a border management project in the Philippines. Consequently, the European Commission has been reluctant to use its money and powers at times (Wennerholm et al., 2010: 20).

Even so, all the EU institutions have undoubtedly seen the need to act in some way in Afghanistan and Pakistan. The signing of the CAPD with Afghanistan in February 2017 demonstrated this commitment and the EU's willingness to deepen its cooperation with Afghanistan. Nevertheless, some have questioned whether the resources allocated by the EU have been used wisely or whether they have disappeared due to corruption or ended up in the hands of warlords. Questions were already raised a few years ago, notably by several MEPs, including Thijs Berman during an Election Observation Mission (Burke, 2014: 7), as well as Pino Arlacchi who undertook a report on the EU's activities in Afghanistan in 2010 (European Parliament, 2010). Thus, overall, the various EU institutions have engaged more with Afghanistan in the last few years than they did in the past. Priorities may have also differed amongst member states and institutions, but this has not prevented the EU from taking action in Afghanistan.

By contrast, although several documents have mentioned the need for the EU to increase its counter-terrorism cooperation with Pakistan (e.g. European Commission, 2010b), it appears that neither the EU institutions nor the EU member states have placed as much emphasis on this country as they have on Afghanistan. Although there have been summits between Pakistan and the EU and engagement documents, such as the SEP, EU action in Pakistan has remained minor in comparison with Afghanistan. It can therefore be argued that the EU member states and the EU institutions have shown a certain level of coherence around the idea that it is necessary to be involved in South Asia, particularly in Afghanistan. However, there has been much less evidence of EU member state and EU institution action in Pakistan.

The then CTC Gilles de Kerchove (2009: 472) noted that Pakistan was one of the countries that 'overemphasise the military approach to the fight against terrorism'. Indeed, Pakistan has shown a preference for military aid and similar forms of cooperation. The USA, for instance, gave Pakistan more than \$20 billion in civilian and military aid in 2001–2011, which dwarfed that on offer by the EU − which, by 2011, the EU claims, amounted to more than €500 million since the start of cooperation in 1974 (EEAS Website, n.d.; *Guardian*, 2011). Concerning the EU member states, the UK − which was a member of the EU until 2020 − has maintained close relations with Pakistan as a result of history and because the UK has for years trained the Pakistani security services (Keohane, 2005: 35).

Thus, in its activities, the EU has focused on the 'soft' side of security, as shown by the details of the missions and the projects in which it has been involved. While the EU can perform these functions and has increasingly done so since 2001, it has largely remained in the shadow of the USA. Even so, it is apparent that the EU can now take more autonomous action and in more areas than it could in the past. However, autonomous action has been somewhat inhibited by the lack of commitment of the member states when it comes to sizeable missions. In contrast, EU action has at times been facilitated by the willingness of some member states to 'hide' behind the Union. For example, a country like the UK used to support EU action – when it was still within the EU – in places like Pakistan and Afghanistan because it allowed it to escape the legacy of its colonial past in this area. In fact, one interviewee stated that, for countries like the UK, which have a legacy in South Asia, it was the motives that Pakistan and Afghanistan questioned, not the capabilities. It was the opposite for the EU; countries questioned its capabilities, not its motives (Interview with EU Official C, June 2010). Consequently, although the EU has been hamstrung by the lack of commitment of its member states on occasions, it has increasingly shown that it can perform certain autonomous roles with its own instruments, including through the IcSP.

Capability

According to Hassan (2021: 2), '[since] the 2016 Brussels Conference on Afghanistan, the EU has committed 5 billion euros (...) to the country'. In addition, he noted that '[the] EU and its [member states] collectively were the largest donor to the international [organisations] that helped rebuild Afghanistan, contributing around 34 percent of the funds for these multilateral reconstruction efforts between 2002 and 2021', leading him to conclude that 'Europe's role in the country's reconstruction was not insignificant'. Indeed, there have been many EU projects of note in Afghanistan and to a lesser extent in Pakistan. This section focuses on projects that can be found in the various

Multiannual Indicative Programmes, Country Strategy Papers, and regional documents, such as the 2009 Action Plan for Afghanistan and Pakistan. With regard to the instruments funding these projects, this chapter examines the EIDHR, the IcSP,² the DCI, and the predecessors of these instruments, in order to identify measures that could be seen as related to counter-terrorism. In addition, this chapter considers projects specific to Afghanistan, focusing on EUPOL Afghanistan and the EU's funding of the UN's Law and Order Trust Fund for Afghanistan. This section highlights how the EU has conducted a wide range of projects in Afghanistan and has also increased its cooperation with Pakistan.

European Instrument for Democracy and Human Rights (EIDHR)

Previously known as the European Initiative for Democracy and Human Rights (2000–06), the EIDHR was established in 2007. In the period 2014–20, it had an overall budget of €1.3 billion and five main objectives: (1) support to human rights and human rights defenders in situations where they are most at risk; (2) support to other priorities of the Union in the field of human rights; (3) support to democracy; (4) EU Election Observation Missions; and (5) support to targeted key actors and processes, including international and regional human rights instruments and mechanisms (European Commission, 2017d).

Wennerholm et al. (2010: 12) have noted that the 2007–10 EIDHR strategy paper explicitly mentioned that the instrument would be used for counter-terrorism, because democracy and poverty reduction could be vital to combating terrorism. While this concern with terrorism seems to be absent in later documents, the regulation for the establishment of the EIDHR for the period 2014–20 still discussed the rule of law, security and justice, all of which are relevant to counter-terrorism (European Union, 2014a).

The projects discussed here have been identified on the basis of the compendiums of projects produced by the EU. Projects concerning Afghanistan will be discussed first, followed by those relating to Pakistan. The projects will also be discussed in chronological order. With regard to the 2007–10 compendium, the analysis will focus on the following areas of the EIDHR: 'democracy – rule of law', 'economic, social and cultural rights', 'fighting impunity', 'fundamental rights protection', and 'gender and women's rights' (European Commission, 2011b). In the 2007–10 compendium, four projects were identified as relevant to counter-terrorism. Those were 'justice assistance mechanism', 'promoting human rights and democratic reform through the enhanced role of civil society in supporting transitional justice', 'women connect across conflicts: building accountability for implementation of UN SCR 1325, 1820, 1888, 1889' (which overlaps with Pakistan), and 'Victims' Shuras: women victims of war mobilizing towards reconciliation and justice'. In Pakistan, three projects could be identified, one of which was already mentioned above – 'women connect

across conflicts', which was run as a regional project. The other two were 'police training and capacity building for democracy and human rights' and 'promoting human rights and fundamental freedoms in Pakistan' (European Commission, 2011b). Other projects were run between 2010 and 2013. For example, in Afghanistan, 'mitigating ethnicity: a democratic response to social division in Afghanistan' was a project that could be seen as contributing to addressing the terrorist threat, whereas no non-confidential projects related to counter-terrorism took place in Pakistan between 2010 and 2013 (European Union, n.d.).

Development Cooperation Instrument (DCI)

The DCI was created in 2007. It has been the largest EU instrument with funding of \in 19.6 billion over the period 2014–20 (European Commission, 2017e). This funding was split between three programmes: \in 11.8 billion for geographic programmes; \in 7 billion for thematic programmes; and \in 845 million for the pan-African programme (European Commission, 2017e). The geographic programmes can notably contribute to 'development and security, including conflict prevention', which is certainly of relevance to counter-terrorism, especially in countries where there has been significant terrorist activity, which is clearly the case with Afghanistan and Pakistan (European Union, 2014b). The 2007–13 programme had a smaller budget of almost \in 16.9 billion (European Union, 2006).

Instrument contributing to Stability and Peace (IcSP)

The IcSP is a Commission-based instrument previously known as the IfS (or Stability Instrument) with funding of approximately €2.3 billion for the years 2014–20 (European Commission, 2016c). The decision establishing the IcSP highlighted its three main objectives: (a) in a situation of crisis or emerging crisis, to contribute swiftly to stability by providing an effective response designed to help preserve, establish or re-establish the conditions essential to the proper implementation of the Union's external policies and actions in accordance; (b) to contribute to the prevention of conflicts and to ensuring capacity and preparedness to address pre- and post-crisis situations and build peace; and (c) to address specific global and trans-regional threats to peace, international security, and stability (European Union, 2014a). Counter-terrorism features highly in the regulation, meaning that the IcSP has had a significant role to play in the EU's 'fight against terrorism'.

A number of projects have been run under the IcSP over the last few years in Afghanistan and Pakistan. Using the search tool on the IcSP website (2014–20) reveals that no projects are currently specifically taking place under 'countering violent extremism/counter-terrorism'. However, this appears to result from a narrow interpretation of counter-terrorism. In fact, projects can be

discovered by selecting the search items 'security sector reform', 'rule of law/ legal and judicial development', 'children, youth, and conflict', and 'women, peace and security and gender mainstreaming', all of which are relevant to counter-terrorism (European Commission, n.d.c). Projects that contribute to peace and resolving conflict certainly have a crucial role to play in dealing with terrorism.

A range of projects took place in Afghanistan, some funded by very large budgets. Over €10 million were devoted to five projects that took place between 2014 and 2020. Those included: 'clean and green cities: promoting stabilisation, peace building and local service delivery in three key cities'; 'support to in-country civil society actors in conflict prevention, peace-building, crisis preparedness'; 'strengthening women's role in peace'; 'women building peace: promoting the role of Afghan women in peace and security processes in the prevention of gender-based violence'; and 'building peace through support for women's access to services and decision makers in Jalalabad, Afghanistan'. In Pakistan, two projects were conducted under the IcSP between 2014 and 2020, with funding totalling over €7 million. They were: 'support to Pakistan's action to counter-terrorism, with special reference to Khyber Pakhtunkhwa' and 'building sustainable peace through developing an internal community mechanism' (European Commission, n.d.c). The IfS also ran projects in the region in previous years. One that particularly stands out took place in Pakistan between 2010 and 2011, where a sum of €11.5 million was spent on a project around Islamabad and Punjab to improve law enforcement, focusing particularly on improving civilian law enforcement agencies and the judicial system (European Union, 2010b). In particular, this project has helped set up and train Pakistan's National Counter-Terrorism Authority. There were also two smaller projects in Pakistan, including a project for post-crisis needs assessment (European Union, 2010b).

In short, the EU has undertaken projects in Pakistan that have been relevant to counter-terrorism by aiding the country in improving its police and its judicial system, thereby possibly contributing to reconciliation amongst the population and dealing with crises. There was also an IfS-funded project in Afghanistan conducted from June 2007 for 18 months and at a cost of €1.35 million (European Commission External Relations, 2008: 28). This project was focused on reforming the justice sector by reducing corruption and ensuring appropriate pay for those involved in it (European Commission External Relations, 2008). Because a weak and corrupt judicial system paralyzes a country's ability to combat terrorism and other threats, this could be considered an important area in which the EU has helped Afghanistan to fight terrorism, insurgency, and drugs. Thus, this analysis shows that the IcSP has grown as a tool to be used for combating terrorism along with other threats.

The number of projects in Afghanistan and Pakistan has grown and the amount of money used for some of these projects has been quite significant.

EUPOL Afghanistan

The EUPOL Afghanistan mission started life as a specifically German initiative, the German Police Project Office, which ran between 2002 and 2007. During the German Presidency of the EU in the first half of 2007, this project was brought under the auspices of the ESDP. Simultaneously, the EU was invited by the government of Afghanistan to launch a police mission. The EUPOL Afghanistan mission was essentially a civilian deployment. Originally, it was planned to run for an initial period of three years (from 2007 until 2010) (Council of the European Union, 2007). However, the mission was later extended to continue until the end of 2016. The basic objective of the mission was to

contribute to the establishment of a sustainable and effective civilian police, which works together with the Afghan justice system to improve the local population's safety. More particularly, the mission monitors, mentors, advises and trains at the level of senior management of the Afghan Ministry of Interior, Afghan Ministry of Justice, Afghan Attorney General's Office, in Kabul and in several regions. (European Commission, 2016b)

Information surrounding the mission has become rather difficult to access, as even its website has been shut down. This is likely linked to the fact that the mission had at best 'mixed results', as the European Court of Auditors − possibly generously − entitled their report on the mission (European Court of Auditors, 2015). What is undisputed is that the mission suffered from a number of problems, from insufficient staffing to the very high levels of corruption present in Afghanistan (Burke, 2014). This has been a rather negative result for a project on which the EU spent a significant amount of money. For instance, the annual budget in 2013–14 was €68 million (European Commission, 2016b). The Court of Auditors demonstrated that the project had improved over time and noted how it had been 'largely successful in training-related activities but less so in mentoring and advising' (European Court of Auditors, 2015: 7).

While the EU clearly attempted to play a significant role in building up an effective Afghan police force – which is crucial for success against terrorist campaigns – it has almost certainly not justified the investment. Even so, it is not only the EU that has struggled in the Afghan political environment, as underlined by the US Special Inspector General for Afghanistan Reconstruction (2021) in a report published in August 2021.

Law and Order Trust Fund of Afghanistan (LOTFA)

The LOTFA was established in May 2002 as part of the United Nations Assistance Mission for Afghanistan under the United Nations Development Programme (UNDP) and by request of the new Afghan government. The project has been nationally managed by the Ministry of the Interior, whilst the Ministry of Finance has been the other implementing partner (United Nations Development Programme, 2010). The programme has six priorities: (1) payment of the police force remuneration; (2) institutional development; (3) procurement, maintenance, and operations of non-lethal police equipment and supplies; (4) rehabilitation, maintenance, and operation of police facilities; (5) gender orientation (selection, recruitment, and training of police); and (6) payment of remuneration of uniformed personnel employed by the Central Prisons Department through specially earmarked contributions (United Nations Development Programme, 2010).

The European Commission has been a key contributor to the LOTFA, alongside significant contributions from the EU member states. For instance, between 2003 and 2008, the EU as a whole (including the member states) provided €205 million to LOTFA and was its largest contributor (European Commission External Relations, 2008: 22). This shows the EU's significant contribution to a major Afghan national project. One of the key ideas behind the project was to ensure that the Afghan National Police would be paid, which would − optimistically − ensure that police facilities are improved and that police officers receive training and apply high professional standards. However, this initiative has been plagued by accusations of bad management and corruption, which has remained a significant challenge for Afghanistan (Afghanistan Ministry of Finance, 2009).

Money laundering project in Pakistan

In the aftermath of 9/11, Pakistan was chosen as a priority country for EU technical assistance. As it was decided that countries should be targeted for support under UN Security Council Resolution 1373, an anti-money laundering project was set up in Pakistan under the DCI (European Commission, 2006a). A sum of around €2 million was to be spent on this project (European Commission Website, n.d.). However, it failed to progress. The European Commission has largely blamed the Pakistani government for this negative outcome, claiming that the project cannot materialise without the necessary legislation in place (Wennerholm et al., 2010: 20).

To sum up, in Afghanistan, the EU has been conducting an impressive range of activities. However, now that the Taliban are back in power, it raises important questions as to what exactly the EU has achieved in the country at great cost. Overall, the EU has been considerably less active in Pakistan. It has been mainly hindered by Islamabad's reluctance to cooperate in counter-terrorism,

which can itself be explained by the strong ties that it already enjoys with the USA – making EU funding less attractive than for other countries – and its willingness to raise its own profile in the region and beyond.

CONCLUSION

This chapter has shown that the EU has played a more active counter-terrorism role in South Asia than has been widely perceived, particularly in Afghanistan. Whilst most of the focus has been on military activities, the EU has - more quietly – engaged in a wide range of counter-terrorism activities and has spent a considerable amount of funding in the region. The Taliban's return to power raises important questions as to what the EU has exactly achieved in the region, though. The lesser amount of counter-terrorism cooperation with Pakistan also highlights the necessity for the EU to find favourably inclined interlocutors in order to develop its counter-terrorism activities. Given the considerable financial support that Pakistan receives from the USA, the EU's willingness to cooperate in counter-terrorism has not traditionally been received with much enthusiasm or interest by the Pakistani government. The souring of the US-Pakistan relations around the same time as the US retreat from the region may provide the EU with an overture as Islamabad may seek to develop new relationships. However, it remains to be seen how the EU will now be able to continue to develop its role in the region without American support.

NOTES

- In the aftermath of 9/11, the EU targeted seven countries for technical assistance. However, the countries concerned were not all formally identified. It has been established that Morocco, Algeria, and Pakistan were three of them (Council of the European Union, 2008, 2009b; Wennerholm et al., 2010).
- 2. The IcSP was known as the Instrument for Stability (IfS) prior to 2014, which itself succeeded the EU's Rapid Reaction Mechanism (RRM). The appropriate name is used in relation to the date of the project in question.

6. EU counter-terrorism, Iraq, and Syria

The US-led invasion of Iraq in 2003 will likely go down in history as one of the greatest strategic errors of the early twenty-first century. By invading Iraq, the USA caused tensions with many of its closest allies in Europe and beyond; violated widely respected global norms and opened up space for other states to deviate from them; diverted attention and resources away from Afghanistan and Pakistan, which became a serious problem later on; and opened up a whole new battleground for jihadists in a state that likely had no strong connection to al-Oaeda or the weapons of mass destruction that were used as a justification for the invasion. Indeed, the US-led invasion allowed militants to take advantage of the chaos that erupted in the early stages of the occupation and existing sectarian divides that had festered for decades under Saddam Hussein. He had privileged his own minority Sunni group and marginalised the majority Shia, and notoriously even used chemical weapons against the Kurds in the north of the country in 1988 (BBC, 2013b). The 2003 invasion unleashed a new and extremely violent insurgency in the country, with civilian deaths peaking at nearly 30,000 in 2006 (Iraq Body Count, 2019). The invasion of Iraq is probably best explained by the presence of a 'neo-conservative bloc' in the Bush administration that had championed regime change in Iraq for a number of years. The 9/11 attacks presented the perfect opportunity to justify such an undertaking (Paul, 2007). Indeed, such an objective had been commonplace in neo-conservative circles already in the 1990s, including in think tanks such as the Project for a New American Century, which included among its members a number of Bush administration officials (Rogers, 2014: 16).

At least initially, there were divisions within Europe and the EU as to how to deal with the invasion of Iraq. Donald Rumsfeld, the then US Secretary of Defense, angered many Europeans when he labelled those states that disagreed with the invasion 'old Europe' and those that supported them 'new Europe' (*Guardian*, 2003). The UK, Spain, Italy, and a number of states due to join the EU, including Poland, sent troops to assist with the initial invasion or offered political support or military assistance during the ensuing occupation. The run-up to the invasion was probably the nadir in Europe–US relations following the end of the Cold War, as Cox (2005: 203) warned that a 'more complex and more abrasive kind of relationship [was] in the making'. Nevertheless, the relationship between the EU and the USA eventually improved following the invasion of Iraq, although the Trump presidency was marked by renewed

transatlantic tensions. Lower-level cooperation, however, seems to have generally been unaffected and has probably strengthened despite the aforementioned disagreements, as shown in Chapter 4 on transatlantic relations. For its part, the EU had not always played a significant role in Iraq and Syria over the past two decades. However, as the threat of Daesh to Europe grew, the EU turned its attention towards this region.

Europe has been significantly affected by the conflicts in Syria and Iraq and by Daesh, especially when one considers the proximity of the region to Europe and the fact that small groups of people linked to or inspired by Daesh have committed a number of attacks across the continent. The attacks on Charlie Hebdo and Paris in 2015, Brussels airport in Belgium in 2016, the Berlin Christmas market in Germany in 2016, and Westminster and London Bridge in the UK in 2017 highlighted the threats that European states faced. A further challenge for Europe – and where the EU has been particularly important – has been that posed by foreign fighters and others who travelled to join Daesh. Foreign fighters are not a new phenomenon. They were already a challenge in some older conflicts, but the number of people who travelled to join Daesh was much larger than that of the fighters who joined al-Qaeda or any one of its affiliates (Tammiko, 2018). According to a report by the International Centre for the Study of Radicalisation (2018: 14), 41,490 people from over 80 states travelled to join Daesh, with 5,904 coming from Western Europe and 7,252 from Eastern Europe, including Russia. In 2016, it was estimated that between 3,900 and 4,300 foreign fighters had come from EU member states and that about a third had returned home by then (International Centre for Counter-Terrorism, 2016). With the loss of Daesh's last stronghold, Baghuz, in Syria in March 2019, some of the people who travelled to join Daesh may return home, although many will do so without any intention of continuing the fight. The next crucial question is, what happens to Daesh now – does the failure of the group mark the end of the religious wave or the start of a new stage, perhaps characterised by a more dispersed, or even digital, caliphate? It has even been argued that Daesh might pose a bigger danger as a global terrorist network than as an organised political entity (Sultygov, 2019). Somewhat controversially, Crenshaw (2018) has suggested that it might be time to negotiate with al-Oaeda and Daesh.

This chapter offers an investigation of the EU's role in Iraq and Syria, focusing on the period since 2003. The region has – perhaps surprisingly given its proximity to Europe and the events unfolding in the region – been largely neglected by EU scholars until now. The chapter is structured into two main sections. The first provides a history of the EU's engagement with Iraq and Syria and the challenges faced by the two states until 2011, which saw the start of the Syrian Civil War and the rise of Daesh. The second section examines the region through the conceptual lens of actorness, focusing on the events

after 2011. Once again, the context, capability, and coherence of the EU as an actor in the region are examined in turn. This chapter focuses primarily on the EU's role in the conflict against Daesh. It argues that EU action has noticeably grown in the region since about 2014. Before then, the EU's role was likely hindered by the existence of civil wars in these states, which had also led to the military forces of the states concerned, the USA, Russia, some EU member states, and other involved states to take precedence. As is well known, the EU largely lacks the capabilities to be involved in high-intensity conflicts in anything other than a support capacity. Due to the range of threats posed by Daesh, the EU has also taken a series of robust internal measures to combat external threats. These will be detailed where relevant. It is worth noting that the EU may well become a more important actor in the region following the end of the formal conflict, given that it is generally better suited to dealing with capacity-building and post-conflict situations. It will be important to attempt to prevent the re-emergence of similar problems to those currently affecting the Middle East in future, but the states in the region must also show a genuine commitment to addressing such problems, especially sectarian divides and the poor treatment of minorities. The EU might be able to play a role in that respect, although divisions amongst member states remain when it comes to engaging with Bashar al-Assad or dealing with other challenging regional actors, such as Iran.

EU-SYRIA-IRAQ RELATIONS PRIOR TO 9/11

Historically, the Middle East was part of the Ottoman Empire, but following the end of the First World War, the region was divided up between the victorious Allies, specifically the UK and France, under the secret 1916 Sykes–Picot Plan. After the Second World War, the colonial powers gradually withdrew or were forced out of the region, leaving a number of weak states, border problems, and exacerbated sectarian problems due to the favouring of one ethnic or religious group over others during colonial rule. The legacy of European colonial rule continues to hang over the region, and some European states, as well as the USA, remain intimately involved in the area, for its strategic importance, among other lingering ties. In contrast, some states in the Middle East, such as Iran, were never colonised at all. The relations between member states and Middle Eastern states have varied and have been controversial at times, such as the UK's military ties with Saudi Arabia, which were criticised against the backdrop of the Yemeni Civil War (Stavrianakis, 2018).

Under the rule of President Saddam Hussein (1979–2003), the EU had 'no political or contractual relations with Iraq', but implemented United Nations Security Council sanctions that were introduced after the first Gulf War in 1990–91 and was the second largest donor of humanitarian relief after the UN

from 1992 onwards (European Commission, 2011e: 7). Following the invasion of Iraq in 2003, the EU focused on reconstruction and humanitarian, political, and financial support. Over time, however, the Union has increasingly developed to take on a stronger security role. EU studies dealing with Iraq have focused on the divides in the EU over the invasion of the country in 2003. Little attention has been given to the development of the EU–Iraq relationship and even less to the widening and deepening role of the EU since the invasion of 2003 (Kienzle, 2013; Puetter and Wiener, 2007; Schweiger, 2004). This is puzzling given the history and the continuing importance of the ties between Europe and the Middle East, the EU's substantial financial support and range of projects, and the global focus on the region, especially since 2003.

By contrast, the EU's relationship with Syria was historically more developed than that with Iraq. A cooperation agreement was signed between the two in 1977 and the state was part of the ENP. An EU-Syria Association Agreement was drafted in 2004 and updated and initialled in 2008, but it was never properly concluded. After the start of the civil war, the EU suspended all its bilateral cooperation with the government of Syria in May 2011 because of the escalation of violence and the unacceptable human rights situation. Sanctions have also been put in place. EU-Syria cooperation under the ENP has been put on hold, whilst its membership of and participation in the Union for the Mediterranean (UfM) has been suspended (EEAS, 2016b). Furthermore, the EU has scaled back its role in Damascus, closing its delegation in December 2012 due to safety concerns and in protest at the Assad regime's handling of the conflict, including its suspected use of chemical weapons. While the drastic decline in EU-Syria relations after 2011 has likely decreased the EU's leverage over Damascus and may have even backfired (Turkmani and Haid, 2016), this is arguably different to the extent of EU action in the region, which has actually grown, certainly in relation to counter-terrorism, since 2011. Evidently, the deterioration of ties has been an obstacle to EU activity, as seen with the Assad regime's decision to suspend multiple-entry visas to EU diplomats, which has made it more difficult for the EU to dispense aid in Syria (Reuters, 2019a). Assad's victory in the civil war now seems likely, but the EU remains divided over whether to start dealing with his regime again, rendering a return to close cooperation unlikely for the foreseeable future.

Iraq and Syria were historically well known for being state sponsors of terrorism, with the latter still on the American list in 2018 (Congressional Research Service, 2018). Both states were infamously identified as being part of the 'axis of evil' identified by President George W. Bush in his State of the Union speech in January 2002. They have experienced significant internal strife: internal conflicts and regional wars have occurred at times over the last few decades (e.g. the Iran–Iraq War, 1980–88). Iraq also faced significant sanctions from many states following its invasion of Kuwait in 1990 and the

ensuing Gulf War. There have long been rumours that discussions between Iraq and the al-Qaeda leadership took place in the 1990s, when the group was based in Sudan. However, these talks are thought to have been fruitless (Byman, 2015: 115). Iraq had also possessed WMDs at some point, but it is now widely believed that 'Saddam had no serious WMD capability and certainly not one capable of threatening the US' by the time of the invasion in 2003 (Hinnebusch, 2007: 209).

The US-led invasion unleashed huge sectarian violence from both Sunnis and Shias, from which al-Qaeda in Iraq (AQI) emerged. Abu Musab al-Zarqawi, leader of AOI until he was killed by a US airstrike in 2006, travelled to Iraq to resist the invasion and agitate for a sectarian civil war. The alliance between al-Qaeda core and AQI/Daesh had existed since al-Zarqawi swore allegiance to Osama bin Laden's al-Qaeda in 2004. This enabled AQI to gain resources and a brand and al-Oaeda core a stake in this new front, although the core had little control over AQI – an issue that was the source of significant friction between the two in the ensuing years, especially regarding strategy. AOI was a loose and peculiar network of groups of militants, including foreign fighters, former Baathists in the armed forces under Saddam Hussein that had been disbanded by the USA, as well as Sunni tribespeople, who saw themselves as marginalised in the new democratic Iraq (Byman, 2015). However, the group was extremely brutal, which repelled many people. This led to the 'Awakening', where Sunnis increasingly turned against AQI, as violence decreased in Iraq after 2007. All this time, Syria had allegedly been involved in encouraging, or at least turning a blind eye to, foreign fighters traversing its borders (New York Times, 2007). Al-Qaeda officially severed its ties with Daesh in February 2014 following a dispute concerning the ownership of the al-Nusra Front, which had originally been established by Daesh in Syria to stir up support (Bacon and Grimm Arsenault, 2017: 9). Daesh subsequently became al-Qaeda's main rival, perhaps even superseding its erstwhile ally – at least temporarily (Byman, 2017: 1106).

The USA left Iraq in 2011 following a dispute with the Iraqi government over the status of US forces. In addition to this withdrawal, the ongoing marginalisation of Sunnis by leaders, such as Nouri al-Maliki, who was then prime minister, the Arab Spring, and the Syrian Civil War enabled the resuscitation of AQI. As a result, Iraq and Syria drifted into chaos. The split between AQI and al-Qaeda core and the emergence of Daesh as the leader of the global jihadist movement ensured that Iraq and Syria would once again become the main battleground for jihadists. However, this time, Daesh appealed to a younger generation of jihadists and offered a more messianic message than al-Qaeda had (Kaplan and Costa, 2015).

Against the backdrop of this extremely complicated and fast-moving situation that has involved a wide range of players, the EU has not typically been

viewed as a major security actor in the region. While the EU has been present in the most recent conflict with Daesh, military force used by powerful states, such as the USA and Russia, has been the real focus, rendering the EU's contributions virtually invisible. Inevitably, it is necessary to limit the scope of this chapter, which will therefore not address the Middle East Peace Process or the Saudi Arabia–Iran rivalry, despite their historical importance for the region and their impact on the conflicts under examination. These issues will only be examined where necessary.

The remainder of this chapter will show that the EU currently has limited contact with the Syrian government, but has acquired a growing role in Iraq. The EU has stepped up its efforts to combat terrorism in both states, as well as increasing internal and international cooperation in the face of the threat posed by Daesh. It may also play a more significant role in the aftermath of the conflicts in a bid to prevent such problems from arising again in the future. This will notably require deeper problems in society to be addressed, such as sectarian divides in Iraq.

THE EU AS A COUNTER-TERRORISM ACTOR IN IRAQ AND SYRIA

These two states are inseparable in terms of the terrorist threats that they have faced in recent times, especially Daesh. Indeed, the group was able to take advantage of the Syrian Civil War from Iraq to rejuvenate itself. Because of this, it is unlikely that progress could be made in one without progress in the other. That is the main rationale for examining them together in one case study. With the decline of Daesh, both states turn to problems of regime survival, with Syria's Assad intent on the continuation of his regime, whereas Iraq may or may not stay on the democratic path. Both states will have to deal with demobilisation of armed groups and severe sectarian divides in the future. There is also the extremely controversial question of what will happen to the Kurdish population, which played a key role in the defeat of Daesh and was given substantial support by the West for doing so. Of course, the impact of the conflict has been felt across the world, with people fleeing the conflict as refugees to other Middle Eastern states, Europe, and North America. Indeed, the UN has estimated that there are over 5.5 million registered Syrian refugees and 6 million internally displaced people (UNHCR, 2018a). Iraq has faced a similar problem, with 3 million people displaced in Iraq and 260,000 who are refugees in other states due to the conflict (UNHCR, 2018b). This is an extremely complex issue, upon which this book only briefly touches as it examines the tools created with the purpose of combating terrorism inside and outside of the EU in the region.

Context

The EU used to enjoy fairly close and stable ties with Syria in the past, with this country being involved in both the ENP and the UfM, whereas Iraq's leadership was effectively isolated with the exception of aid up until 2003. Clearly, then, the invasion of the latter in 2003 brought about a huge change and an opportunity for the EU to engage with the new regime. As pointed out before, since 2011, the EU–Syria relationship has basically been frozen. Also, the attention of the EU has largely shifted towards the 'refugee crisis' that occurred as a result of the Syrian Civil War (European Commission, 2019a). By contrast, the relationship with Iraq has grown. The most significant events that occurred following the 2003 invasion of Iraq and changed the context for action were arguably the departure of US troops from Iraq in 2011, the start of the Syrian Civil War in 2011, and the rise of Daesh that followed. After the apparent defeat of Daesh in Iraq and Syria, the USA has quickly shifted its attention towards Iran. Post-conflict issues in Iraq have remained largely neglected, which may further add to the instability of the region.

The invasion of Iraq offered the opportunity of a reset of the EU's relationship with that country. The EU responded fairly swiftly after the invasion, with the 2004 'The EU and Iraq: a framework for engagement' document, a 2005 Joint Declaration on Political Dialogue, as well as the European Commission's 'Recommendations for renewed engagement with Iraq' in 2006. The original framework for engagement outlined a set of ways in which the EU-Iraq relationship might develop, listing a number of priorities for the EU from what was required immediately to the medium-term future, all 'to help lay the foundations for a secure, stable, democratic and prosperous Iraq, at peace with its neighbours and its region, and a full and active member of the international community' (European Commission, 2004a). Then, the 2005 Joint Declaration on Political Dialogue established a political dialogue between the two parties. It was intended that discussions would focus on: 'promotion of good governance; the protection of human rights; rule of law and democracy; dialogue on non-proliferation and counter-terrorism; conflict prevention and resolution; and cultural issues' (Council of the European Union, 2005c). Finally, the 'Recommendations for renewed engagement with Iraq' were published in changed circumstances in the midst of increasing violence. The EU praised the progress that Iraq had made with regard to political and constitutional issues, but also highlighted greater inclusiveness in the political process and government and the security situation as being crucial for greater EU engagement at the time (European Commission, 2006b). In 2012, the EU and Iraq agreed on a Partnership and Cooperation Agreement, the first such agreement between the two states. Interestingly, the 'Title 1' section focused on foreign and security policy, including combating terrorism (Article 4) (European Union, 2012).

Thus, the EU-Iraq relationship has significantly developed in the last few years, going beyond the more ad hoc cooperation of the post-invasion period towards being more settled and dealing with an expanded range of issues. Much of what the EU originally provided was aid-based. It has been a key aid provider to Iraq since the start of the twenty-first century. However, the EU has also taken on a stronger role in a wider range of activities linked to security, as explained below.

The Syrian Civil War evidently caused a sea change in the relations between the EU and Syria. In the past, there had been a certain degree of EU-Syria cooperation. However, European governments expressed criticisms at the Assad regime's handling of the war and a deteriorating security situation contributed to the Union's decision to close its delegation in Damascus in 2012. Whereas Syria was historically part of a number of regional programmes, it was now excluded. This has reduced EU leverage over the state and repairing the ties between the two – should this be seen as desirable – will be difficult. However, if Assad remains in power, the EU will face a difficult choice between re-engagement or maintaining the freeze on relations. For the time being, the EU's leverage over the state is not significant and it will likely need to rebuild its connections with the Assad regime in order to seriously increase the importance of its role in the long term should Assad's regime survive. Furthermore, the Syrian government has been reasserting its control over its territory, thereby decreasing the space for EU action in relation to terrorism in any case.

The rise of Daesh and terrorist attacks in Europe linked to this group were a further contextual change that compelled the EU and European states into action, with the EU responding with elements of a regional strategy for Syria and Iraq in 2015. This was a first for the EU – a strategy focusing so clearly on a particular terrorist group. The overall objectives of the strategy were:

to counter the threat posed by [Daesh] and other terrorist groups to regional and international stability, and simultaneously to create the conditions for an inclusive political transition in Syria and lasting stability in Syria and Iraq as well as in refugee-hosting countries in the region, while alleviating the human suffering caused by the ongoing violence and displacement. (European Commission, 2015c)

The strategy foresaw engagement with not only Syria and Iraq, but also adjacent affected states, such as Jordan. It offered a range of concrete actions involving various EU tools and has been revised on an annual basis since then. Building on this, the EU agreed on a specific strategy for Syria in 2017 and one for Iraq in 2018 (Council of the European Union, 2018b; European Council and Council of the European Union, 2017).

Furthermore, the conflicts in Iraq and Syria presented a significant internal problem for the EU on a hitherto unprecedented scale, namely the challenge of foreign fighters (Council of the European Union, 2015). Indeed, Daesh has acted as a point of attraction for a small, but not insignificant number of people. This has posed serious problems for some European states, as their own citizens normally have the right to return and move around Europe if they choose to do so. It is therefore possible that radicalised individuals might be able to travel back to Europe, spread propaganda, divide communities, recruit others, and even plan and carry out attacks. Nevertheless, the flow of returnees following the declining fortunes of Daesh has not been as significant as feared at one point, although many will pose a long-term threat to their states. It will be challenging to manage this over a long period of time (New York Times, 2017). One of the main concerns is that some individuals or groups will return home and 'professionalise' local groups, as seen in Sri Lanka in the case of the Easter 2019 attacks, which killed nearly 300 people. The terrorist group that has been accused of the attack, the Sri Lankan National Thowheed Jamath, went from small-scale smashing of Buddhist statues in December 2018 to an extremely sophisticated and coordinated large-scale bombing in April 2019 (BBC, 2019b).

In December 2018, the Trump administration made the surprise announcement that US troops would be withdrawn from Syria, following their 'defeat' of Daesh (*The Economist*, 2019). Three years later, the US-led global coalition against Daesh also announced that it had ended its combat mission in Iraq. This US withdrawal has led some observers to fear that it will lead to the increasing influence of Russia and Iran in the region (BBC News, 2018). Furthermore, Daesh has continued to pose a threat, notably with its cells operating in rural areas, which carry out hit-and-run attacks. There is a concern that Daesh may take advantage of the current power vacuum to regroup and grow considerably again in future.

It can be argued that the US withdrawal from the region has created opportunities for the EU and its member states to demonstrate their credentials as security providers. They are no substitute for the USA's immense military power, but they might be able to pressure the Iraqi government into reconciling its Sunni and Shia populations and to assist with building more robust institutions for the long term. For instance, European states have acquired significant experience in managing ethnic, religious, and national divides through, for example, consociational democratic systems. Nevertheless, this remains a very challenging context for EU action, especially if the security situation were to become more unstable again.

To sum up this section, the US invasion of Iraq in 2003 completely changed the context of EU-Iraq relations. Where the Union had not engaged with Iraq under Saddam Hussein, the invasion opened up the way for a possible

improvement in this relationship. This was challenging while the state was existentially threatened, but the considerable weakening of Daesh has led to an enhancement of the EU-Iraq relationship. EU-Iraq ties have certainly expanded in scope since 2003. However, these advances remain fragile and could be threatened should the security situation in Iraq deteriorate again. With regard to Syria, the opposite has occurred since the start of the civil war. The EU suspended its cooperation with the Assad regime, thereby decreasing its leverage on the state, if it had any. Should Assad remain in power, then the EU may eventually have to engage with his regime again in the years to come. It is also likely that Assad will not make re-engagement easy. Moreover, some have criticised the effects of the sanctions imposed by Western states on Syria. which may end up strengthening the Assad regime at the expense of its citizens and allow its allies to further assert themselves in the Middle East (European Council on Foreign Relations, 2019a). Furthermore, the rise and decline of Daesh brought the region to the particular attention of the EU, not least because of foreign fighters and attacks in Europe linked to Daesh. Evidently, the context for action in this region has fluctuated and has been rather challenging for the EU overall. Nevertheless, the US withdrawal from the region may have arguably opened up space for a more significant EU role, although this may remain rather modest given the continued security challenges characterising the region.

Coherence

As previously mentioned, the EU has often been criticised for lacking coherence when it comes to its counter-terrorism policy. Perhaps expectations cannot be set very high due to the disparate experiences of terrorism across Europe. Scholars have frequently commented on the varied perceptions of terrorism within the EU (Bures, 2010a; Monar, 2007a). Foreign fighters and others travelling to join Daesh have somewhat changed this picture. Daesh has brought EU member states closer together on the issue. Notably, there have been reports of terrorist activity in Eastern European states, many of which previously tended to pay only limited attention to terrorism.

An analysis of the coherence of the EU's action must start with the divides over the war in Iraq in 2003. Germany and France famously objected to the US-led invasion of Iraq, while the UK and a set of other states chose to support Washington. Although this was a serious low point for transatlantic ties in the post-Cold War period and caused problems within Europe, the damage did not last for long. Transatlantic cooperation increased in the following period, despite some disagreements. Counter-terrorism cooperation in Europe has only grown since 2002–03. The UK was even able to take a lead on counter-terrorism during its presidency in 2005, with the adoption of the EU's

Counter-Terrorism Strategy and other measures. Thus, the disagreements over Iraq do not appear to have caused serious damage in hindsight. Not only this, but the EU also quickly became involved in post-invasion Iraq, although its role has taken some time to fully develop. Brexit will likely prove much more damaging, dividing European states further, as well as reducing resources and expertise overall.

Even so, the invasion of Iraq created a brand new battleground for jihadis that was geographically closer than Afghanistan and Pakistan. In this sense, it is entirely possible that the US-led invasion of Iraq ultimately led to greater convergence on the issue of terrorism and thus, at least to an extent, cohesion due to the long-term consequences of the invasion significantly affecting Europe in a myriad of ways. In particular, the issue of foreign fighters has been a serious one. This problem existed even prior to the rise of Daesh. For instance, 'Europe's first female suicide terrorist' Muriel Degauque, who killed herself in Iraq in late 2005, is a case in point; she was Belgian and considered a convert to Islam. Even so, foreign fighters from Europe to Iraq were far fewer in number than what occurred following the rise of Daesh. Fighters mainly travelled to Iraq from places other than Europe prior to the emergence of Daesh. With Daesh, the challenge of foreign fighters became significantly more acute for Europe. Almost all European states were affected in some way, even though numbers varied widely, from 2,000 from France to a handful from Bulgaria (BBC, 2019c). Nonetheless, the majority of EU member states, from the historically most to least experienced with terrorism, had some foreign fighters. Indeed, numerous stories of Eastern European foreign fighters have emerged in recent years, raising awareness of the threat of terrorism and at least going some way in reducing the gap between Western and Eastern European states. The threat was varied and still is, but it is now most of Europe that has a similar problem. Even so, differences remain and familiar problems continue to act as obstacles to the EU's approach to terrorism and foreign fighters, including perceptions of the threat posed by terrorism (Bures, 2018). It is also worth being aware of the possibility that terrorism could be used as a way of further marginalising, for example, Roma communities in Europe. Calling someone 'terrorist' or susceptible to radicalisation can open the door to further government intervention. The very existence of an EU strategy focusing on foreign fighters indicates that there is a certain level of coherence on the issue (Council of the European Union, 2015). The main concern, of course, has been that foreign fighters would be able to re-enter the EU, then travel freely within the Schengen zone and commit attacks anywhere.

A further complicating factor has been the 'refugee crisis' triggered by the conflicts in the region and how to deal with it. This also relates to how to deal with Assad and the future of Syria. The EU's stance on this has typically been that it supports a UN-led peace process (United Nations, 2019). However, in

some member states, some voices have begun to call for re-engaging with Assad, notably in order to send refugees home, as the UN-led process has ended up in deadlock (Reuters, 2019b). On Syria, without a single voice, the EU is unlikely to have much influence in practice, even though the withdrawal of the US has potentially opened the door to a stronger role for the EU. A further point is that broader regional concerns will also play a role in Syria. Iran has backed the Assad regime. An Assad victory would therefore empower Iran to some extent, and relations between the USA, the UK, and Iran have been very tense, while the EU has attempted to hold together the nuclear deal agreed with the USA in 2015 (European Council on Foreign Relations, 2019b). Given that Iran has now breached the terms of this deal in protest, it is not entirely clear to what extent it is still alive (BBC, 2019c). Iraq will also come into play here on the basis that Iran and Iraq appear to be more politically united than they were in the past. Therefore, there are a number of linked issues in the Middle East on which the EU has appeared divided at times in recent years.

While terrorism has gone up the agenda and member states have evidently been more concerned about it generally, threat perceptions have not been uniform and are unlikely to ever be so. The decline of Daesh in Iraq and Syria might even decrease concern about the problem, although the group might actually become more dangerous, notably because of uncertainties stemming from its decentralisation. The 'refugee crisis' and what to do about Assad and Syria have been significant questions for the EU, on which it appears to have been quite divided. Thus, even though the US withdrawal may seem to have left more space for EU action in the region, it will be extremely challenging for the EU to develop its role given the challenges that it faces on the ground, in addition to a certain lack of coherence on the key issues.

Capability

The EU has claimed that, between it and the member states, €10.8 billion has been raised for those affected by the Syrian conflict and other states in the region, as well as €1.05 billion for those inside Syria (European Commission, 2018c). However, the EU has suspended direct cooperation with Syria. It has also been criticised for some of the measures that it has taken in order to reduce the flows of asylum-seekers coming from the region, in particular a controversial agreement with Turkey (Deutsche Welle, 2018; Léonard and Kaunert, 2021b). For Iraq, the EU has provided hundreds of millions of euros across a range of sectors (European External Action Service, 2018). This section examines how the EU has grown as a counter-terrorism actor over time in this region, especially to help deal with Daesh, but also to assist with capacity-building and reconciliation. As previously mentioned, the US decision to withdraw troops from the region may facilitate the development

of a more significant role for the Union. In that respect, it is important to recall that Syria is part of the ENP, contrary to Iraq. This has an impact on the instruments that the EU has at its disposal in relation to each of these countries.

Legal basis for cooperation

The legal basis for the EU's relationship with Iraq has been the 2012 Partnership and Cooperation Agreement, which has sanctioned EU-Iraq cooperation on counter-terrorism, although such cooperation had been of concern since the renewal of EU-Iraq relations following the US-led invasion. The frozen and likely abandoned 2004 EU-Syria Association Agreement also included a section on counter-terrorism cooperation (Title 9) (European Commission, 2004b). At that time, hopes were high that reforms in Syria were under way and that such an agreement would strengthen the credibility of them, although there was concern that the clause on weapons of mass destruction might offer a loophole and exit option (Zorob, 2008). Having said this, the existence of such clauses says little about whether any substantive cooperation actually followed these agreements. At present, for instance, EU direct cooperation with the Syrian government is non-existent. Ultimately though, agreement on the importance of counter-terrorism cooperation is clear in the existing agreement with Iraq and in the proposed one with Syria, with the caveat that these might mainly aim to signal some willingness to cooperate with no actual intention or idea as to how this might be carried out in practice.

EU instruments

Development Cooperation Instrument (DCI)

The EU has used the DCI in its cooperation with Syria, but not for the purpose of tackling counter-terrorism directly. It has been used for an agricultural project. Having said that, climate change and drought have been blamed by some for being a cause of the Syrian Civil War as people allegedly travelled to the cities following a drought, although this link has also been disputed (Selby et al., 2017).

In Iraq, longer-term projects have been financed by the DCI. Total development assistance between 2014 and 2017 amounted to €124.4 million. Of course, not all projects have concerned terrorism, but there have been projects concerned with good governance and stabilisation, which very clearly link to the post-conflict challenges and long-term management of sectarian issues in Iraq. The EU has also topped up some of these efforts since 2017 with more funding (EEAS, 2018).

Contributions to UN projects

The EU has also contributed large amounts of money to stabilisation funds to deal with post-conflict challenges and de-mining, including $\[\in \]$ 50.4 million to restore services, to repair the infrastructure, and to support small businesses for restoring economic activity, as well as $\[\in \]$ 10 million for de-mining (European Commission, 2017f). These projects have been implemented following the withdrawal of Daesh from these areas. They have been conducted by the UNDP Funding Facility for Stabilisation and the UN Mine Action Service. In addition, in 2019, the EU contributed $\[\in \]$ 50 million in humanitarian aid and development cooperation to help deal with post-conflict challenges, from healthcare through jobs for vulnerable youth to the restoration of Iraq's cultural heritage, which Daesh had attempted to obliterate (European Commission, 2019d).

The provision of this funding has brought the EU's total humanitarian assistance to €420 million and its development cooperation to €309 million since the beginning of the EU's real entry into the conflict against Daesh in 2015 (European Commission, 2019d). Some may object to the linkages that have been made in this way between development and humanitarian aid, on the one hand, and security and the interests of the EU member states, on the other hand. Indeed, it may be considered a questionable activity to link development and the provision of humanitarian aid to places where it benefits Europe and is in line with security challenges.

Instrument contributing to Stability and Peace (IcSP)

The IcSP has been used by the EU in Syria, with nearly €58 million having been spent in the state on 20 projects since 2015 and over €73 million over the longer term (European Commission, 2019c). Once again, not all of this money or the projects have been relevant to counter-terrorism, but some of the largest ones have related to mine clearing (€10 million) and support for the peace process (€9 million) (European Commission, n.d.c).

In Iraq, the IcSP has been used to offer over €40 million since 2015 and there have been some significant projects relevant to dealing with Daesh. Once again, the EU has not been able to directly combat Daesh, but it has supported institution-building and other forms of assistance, including for tackling post-conflict issues. Of particular note have been projects supporting the police (over €3.9 million) and a human rights compliant counter-terrorism strategy (two phases: (i) 2017–18, €3.5 million; and (ii) 2019–22, €7 million) (European Commission, n.d.c). Evidently, the EU has conducted a range of projects relevant to counter-terrorism under the IcSP.

Regional Trust Fund in Response to the Syrian Crisis

Although it has not been set as a specific counter-terrorism tool, the EU has introduced a regional fund to help deal with the Syrian crisis. Between 2014 and 2021, over €2.3 billion for Syria were raised and used for a range of purposes (European Commission, n.d.e). This underlines again the significant level of funding spent by the EU on a number of projects in the region.

European Union Advisory Mission in support of Security Sector Reform in Iraq

In 2017, the EU launched the EU Advisory Mission in support of Security Sector Reform in Iraq (EUAM Iraq), with an original mandate of one year. This mission was requested by the Iraqi government to advise and offer assistance in implementing the civilian aspects of Iraq's National Security Strategy. It originally had a budget of €14 million, as well as 51 staff, including 36 from the EU (EEAS, 2017). The mission was later extended until April 2020 (EUAM Iraq, 2018) and then again until 30 April 2022 (Council of the European Union, 2020). It 'involves support to institutional reform and efforts to help counter terrorism (including countering violent extremism) and organised crime, with specific reference to border management, financial crime in particular corruption, money laundering and trafficking of cultural heritage goods' (Council of the European Union, 2020).

European Instrument for Democracy and Human Rights (EIDHR)

This is the smallest instrument being used by the EU in Syria, with €16.1 million to support 'human rights, accountability, and independent media' (European Commission, 2019c). Projects funded by this instrument therefore have the potential to contribute to counter-terrorism efforts.

European Neighbourhood and Partnership Instrument (ENPI)/European Neighbourhood Instrument (ENI)

This tool has not been used to Iraq because this country is not covered by the European Neighbourhood Policy, whilst cooperation with the Syrian government was suspended by the EU in 2011. However, the instrument has still been used within Syria to assist the local population and the provision of €268.6 million since 2011 has not been insignificant. It appears to have been used for 'resilience' and 'post-conflict recovery' (European Commission, 2019c), although little further information has been made available regarding the projects run under the ENI.

To conclude, this section has evidenced that the EU's involvement in Syria and Iraq has considerably increased since 2011. The Union has conducted a range of projects in both states and has mobilised a vast range of tools in order to

conduct various projects at least partially linked to counter-terrorism. The EU has not been able to participate in the harder aspects of security. However, the decline of Daesh has enabled an increased role for the EU, which might help at least improve institutions and encourage better governance in the long term – contributions to security that military assets cannot make. The cases of Iraq and Syria have shown how many instruments the EU has developed for projects linked to counter-terrorism. Thus, its capabilities in the field of counter-terrorism have significantly increased. Nevertheless, some may find it unfortunate that development, humanitarian aid, and security have been linked. One should also not overestimate the capabilities of the EU, given the scale of the security challenges on the ground, especially if Daesh goes through a revival in future. Finally, another important factor to consider is whether the EU's interlocutors will be cooperative and receptive to advice and training, which greatly matters when attempting to cooperate with other states.

CONCLUSION

Overall, the Middle East has experienced a significant amount of upheaval since 9/11. The American invasion of Iraq contributed to revive an al-Qaeda that was under significant pressure in Afghanistan and Pakistan and gave it a whole new battleground following its alliance with AQI. The invasion has led to the unleashing of a very high and long-lasting level of violence in the region. The uprisings from 2011 onwards initially took jihadist groups by surprise. However, they have been able to utilise them for their own ends, not least the continuing Syrian Civil War. The area comprising Iraq and Syria has been the focal point of terrorist violence, but the decline of Daesh has given way to a series of post-conflict challenges, such as sectarian problems and rebuilding, as well as concern about growing Iranian power and influence in the region. Global powers have been divided on how to deal with these challenges, which will make it more difficult to resolve them in the near future. Overshadowing this for the EU has also been the so-called 'refugee crisis', which has significantly divided member states (Léonard and Kaunert, 2021b, 2021c).

The issue of foreign fighters has brought the EU member states closer together to a significant extent in order to develop a common approach to dealing with Daesh. EU action has also significantly increased, ranging from humanitarian aid through institution-building to tackling the long-term challenges faced by Middle Eastern states. There has been an opportunity for the EU to play a significant role here, as the USA has begun to shift its attention away from the region. Indeed, as the largest aid donor in the world, the EU may well be expected to play a significant role in the reconstruction of Syria and Iraq. The EU and its member states also possess significant experience in dealing with terrorism, reconciliation, and building robust democratic institu-

tions that can assist with long-term solutions to sectarian divides. Nevertheless, raising its profile as an actor in the region will be a very significant challenge for the EU, in particular if the security situation deteriorates, possibly as a result of an increase in terrorist attacks.

7. Conclusion to *The European Union as* a Global Counter-Terrorism Actor

Our war on terror begins with al-Qaeda, but it does not end there. It will not end until every terrorist group of global reach has been found, stopped and defeated. (George W. Bush's Address to a Joint Session of Congress and the American People, 20 September 2001)

President George W. Bush had just announced the beginning of the 'war on terror' to the American people in the wake of 9/11. He was right in assessing that it would not end with al-Qaeda, as shown by the rise to prominence of Daesh. The EU, as this book has demonstrated, joined in this 'war on terror' and became a global counter-terrorism actor. The evidence presented in this book demonstrates that the EU has significantly grown as a counter-terrorism actor since 2001 in several areas of the world. This concluding chapter starts by restating the aims of the book. Then, it highlights its main research findings and contributions to the literature on EU counter-terrorism and that on European integration and actorness. The following section considers two important challenges that are likely to influence the development of the EU's global counter-terrorism role in the near future, namely the rise of right-wing terrorism and the consequences of Brexit. Finally, the chapter identifies some promising avenues for future research.

REVISITING THE AIMS OF THE BOOK

The main aim of this book was to examine the extent to, if any, and the ways in which the EU has become an actor in global counter-terrorism. It was shown that many scholars have argued that the EU has not been particularly important in counter-terrorism and has mainly played a supporting role. However, this claim had not been substantiated by any systematic and detailed empirical study, particularly at the global level.

Thus, this book has addressed a distinct gap in the scholarship on European security and counter-terrorism. Although a growing body of literature has emerged that provides regional and thematic explorations of European counter-terrorism policies, there has not been any systematic study of the broadened role of the EU as a counter-terrorism actor at the global level. Furthermore, the existing literature on the policies that constitute European

counter-terrorism has tended to discuss them in isolation from one another, in line with the traditional divide between CFSP and AFSJ matters. The critical value of this book is thus twofold. Not only does it provide a comprehensive picture of EU activities in the area, but it also seeks to demonstrate that the above-mentioned divide, initially brought about by the Treaty of Maastricht (European Commission,1992), has led to limitations in the understanding of the dynamics across the spectrum of European counter-terrorism activities. Instead, the position adopted here has been that of exploring various linkages and interaction between these policy fields from a theoretically informed and also substantively empirical perspective.

This book has shown that the EU's role in external counter-terrorism has tended to be underestimated. In contrast to commonly held views in the literature, it has highlighted several interesting and important developments not only for EU counter-terrorism abroad, but also with regard to the progress that the EU has made towards becoming a truly global actor with a broad range of capabilities. In the last two decades, the EU has shown that it can extend its reach beyond trade and economic policy into certain aspects of foreign and security policy, which is a sign that the EU nowadays is a greatly enhanced actor when compared to that of 2001. Evidently, progress has been impressive – albeit in cooperation with some countries and in some policy areas more than others. As has been demonstrated, unprecedented and unexpected counter-terrorism cooperation now takes place internally amongst EU member states and externally between the EU and third states. Diverse forms of terrorism and many terrorist groups presented threats to Europe for much of the twentieth century. However, the more transnational nature of the Islamist threat has encouraged EU member states to deepen their cooperation. Nevertheless, more than two decades after the start of the EU's 'fight against terrorism', European countries still face significant terrorist threats from various external and internal sources, ranging from jihadist terrorists – some of whom are more or less loosely affiliated with al-Oaeda or Daesh – to far-right terrorists.

Also, growing concerns over linkages between criminal gangs, terrorist groups, and a number of other threats has played a central role in raising awareness of terrorism in Europe, as set out in some of the most important EU documents, including the ESS (European Council, 2003a, 2003b). Furthermore, perhaps the clearest assessment of these links has been contained in Europol's various TE-SAT reports. Criminal activities have been extensively used to finance terrorism. Such activities in which terrorist groups are involved, either through affiliation with criminal groups or through their own operations, include the trafficking of illegal goods and substances, such as weapons and drugs, trafficking in human beings, financial fraud, money laundering, and extortion. These links are summed up as a crime–terror nexus, which has led the EU to strengthen its commitment to fighting terrorism and other threats.

Conclusion 127

Many European countries face internal and external challenges to their security. However, it must be remembered that European states have been challenged at a level far lower than many other regions and countries around the world, especially the Middle East and South Asia, which have experienced attacks more frequently and suffered significantly higher casualties over the years. For example, the frequent bombings in Iraq, Syria, Pakistan, and Afghanistan serve as a constant reminder that many more people are killed or injured by terrorism every year in places outside Europe. Even so, with the world being so interconnected through transport and trade links, those trained in Pakistan or Afghanistan – especially those who possess European passports – can return to Europe and set up cells inspired by jihadi terrorism, raise money for such organisations or even perhaps commit a terrorist attack.

MAIN RESEARCH FINDINGS OF THE BOOK

This book has made theoretical and empirical contributions mainly to two strands of the academic literature, namely that on EU counter-terrorism and that on EU integration and actorness. First of all, it has been noted that the threat of international terrorism has made European integration proceed in surprising and interesting directions internally and externally, particularly as the EU has become increasingly involved in counter-terrorism policies by cooperating with a growing number of third countries. This growth in EU counter-terrorism cooperation has been driven by the 'collective securitization' of the transnational threat of terrorism within the EU, whereby the terrorist threat has been seen as a 'European' – rather than 'national' – threat calling for a collective response of the EU member states.

Second, this book has adopted a unique theoretical approach to the study of EU counter-terrorism cooperation in the form of actorness, which it has applied to several case studies in order to analyse the evolution of the EU as a counter-terrorism actor. In addition, this research has taken a broader view of counter-terrorism and external relations. Thus, this project has added new empirical knowledge, which has included an examination of the development of the EU-US relationship following the Treaty of Lisbon and of the largely overlooked role of the EU in Afghanistan and Pakistan, as well as in Syria and Iraq. Overall, this has demonstrated that the literature on the external dimension of EU counter-terrorism has often lacked depth and geographical scope. Furthermore, this project has highlighted that the EU is, in fact, becoming a counter-terrorism actor of growing importance and with an ever-diversifying number of policy options available. Thus, this book has shown how European integration has been profoundly affected by international terrorism and has increasingly taken place in the field of counter-terrorism. In fact, very few observers would have predicted that EU integration in counter-terrorism matters would have progressed so far in such a short space of time. In terms of theoretical contribution, this project has examined EU actorness in several case studies, focusing on the EU's evolution in each case. The book has taken as a starting point that the EU is a *sui generis* actor in the global system (Bretherton and Vogler, 2006; Ginsberg, 2001). In doing so, this research project has viewed the EU as an individual type of actor, and one that cannot in all ways be compared to a state because many of the processes, the functions, and the structure of the organisation are different. In saving this, it is important to recall that the EU is neither a state nor a typical international organisation (Keohane, 2005). It follows that it is important to avoid systematically – and implicitly or explicitly – comparing the EU to a state and to the USA in particular as the EU's significant other. This book has noted that other approaches to international relations have tended to overemphasise relations amongst states, which has led to not treating the EU as an actor. This focus on states has prioritised coercive methods over others, which has led to overlooking a range of activities that the EU performs. As has been demonstrated throughout this book, an actorness-based framework appreciates what the EU is, whilst also avoiding comparisons with other types of actors. Importantly, this book has asked a fundamental question about the EU: is the EU a counter-terrorism actor at all? It has analysed the EU as an actor in several case studies and has highlighted that EU counter-terrorism actorness has varied across the world. This approach to EU counter-terrorism has provided a new and interesting way of assessing it, which has helped construct the EU as an actor by examining what the EU can contribute – and has contributed – to global counter-terrorism, rather than mainly criticising it for what it cannot.

Third, this book has made significant empirical contributions to the literature. The external role of the EU in counter-terrorism has remained widely neglected. Whereas the internal dimension of EU counter-terrorism has seen a large amount of research, the same cannot be said about the external dimension. This lack of empirical data on the external dimension of EU counter-terrorism has largely led to the conclusion that the EU has not been a significant counter-terrorism actor. Importantly, this research project has challenged such an assumption. It has shown that EU cooperation with a range of third states has grown considerably since 2001. More specifically, this research has generated new insights that have contributed to the literature on the external dimension of EU counter-terrorism.

The case of greatest interest in the literature has traditionally been that of the EU-US relations. However, these studies have often been quite limited in their scope and timespan or overly focused on rhetoric more than practice. The argument that the EU has been heavily influenced by the USA in its counter-terrorism measures is not entirely new, but this book has provided a fuller, more nuanced, and more dynamic view of this bilateral relationship.

Conclusion 129

Most interesting of all empirically, perhaps, has been the EU's major contribution to global counter-terrorism efforts in Afghanistan and Pakistan, an area of the world that EU researchers have generally neglected. Similarly, little attention had generally been given in the extant literature to the EU's efforts to combat Daesh in the Middle East. This book has highlighted the significant scope of activities related to counter-terrorism that the EU has conducted in Syria and Iraq.

The USA remains by far the most important counter-terrorism partner for the EU, although the transatlantic relationship has been marked by some tensions in the last few years. At the highest level, the EU's overall approach to counter-terrorism has remained resolutely based on law enforcement, as the EU has attempted to re-frame the effort against jihadist movements as a more low-key 'fight against terrorism', rather than a war, although this move has not been supported by all member states. In the aftermath of 9/11, divisions amongst European states became visible over their relations with the USA, as epitomised by Rumsfeld's distinction between 'old' and 'new' Europe in the run-up to the war in Iraq. Despite these disagreements, the USA has exercised considerable influence over the development of EU counter-terrorism measures, being referred to as a 'norm-maker', whilst the EU largely adopted the role of a 'norm-taker' and some controversial measures - that may not have been acceptable prior to 9/11 – were adopted (Argomaniz, 2009a; Ripoll Servent and MacKenzie, 2012). However, it is important to highlight the role of some internal players in the introduction of such measures (Kaunert et al., 2012). Thus, US influence should not be overestimated overall, as notably observed by de Goede (2008a) who has shown how the EU was a world leader in pre-emptive security practices prior to US intervention. EU-US cooperation is now wide-ranging and has been underpinned by various agreements covering the air travel of individuals, international bank transfers, and trade, amongst others. Given the seemingly decreasing importance of Europe for recent US administrations, it can be argued that EU-US counter-terrorism cooperation is likely to be maintained in future, albeit with limited potential for significant growth.

The 'war on terror' started in Afghanistan and Pakistan, a theatre of which the strategic importance has decreased over time with the withdrawal of Western troops. In August 2021, the US administration withdrew its last remaining troops after a 20-year war, which saw bloodshed and failed to defeat the Taliban (Williams and Parkin, 2021). The EU stepped up its efforts over time and contributed to a number of capacity-building projects. It footed the bill for a significant number of high-profile international projects that cost billions of euros, until it decided in August 2021 to freeze its development funding and government assistance to Afghanistan following the Taliban's takeover. One can question what most Western efforts in Afghanistan, includ-

ing the EU-funded projects, have exactly achieved, especially as the country is now controlled again by the Taliban. In Pakistan, the EU has continued to run projects on a modest scale, but it is unclear what impact those have had. However, it is noteworthy that the EU has undertaken all these activities in an area not in its vicinity. However, given the lacklustre outcome in the region overall, it seems unlikely that the EU will often venture out further afield in the near future. Furthermore, one can argue that the most pressing challenges faced by the EU originate from within the organisation or in its neighbourhood.

Following on from this case of South Asia, the book has highlighted a higher than expected level of EU activity in Syria and Iraq after the rise of Daesh. Since the invasion of Iraq in 2003, the main area of Western efforts to combat terrorism has been the Middle East. Unsurprisingly, the USA has been the dominant player here with a sizeable military presence, but the EU's role in the region has grown significantly over time. It will be interesting to see how it will evolve in the years to come, now that Daesh has been considerably weakened, although the US withdrawal from the region could leave Europe more vulnerable should Daesh undergo a revival in future.

Thus, the overall assessment is that the EU has actually been far from being the 'absent friend' that Keohane (2008) called it in one of the first pieces considering the external dimension of EU counter-terrorism over a decade ago. However, this characterisation of the EU has remained influential overall. As previously discussed, it has notably been fuelled by the use of a rather narrow definition of external action, which has obscured the EU's strengths, emphasised its weaknesses, and implicitly compared the EU to the USA. A more accurate description of the Union nowadays is arguably that it has been 'an invisible friend' - not absent, but working in the less spectacular and less visible aspects of counter-terrorism. The EU has been quietly active and growing in capabilities since the attacks on the USA almost twenty years ago. While not fighting a spectacular and newsworthy 'war on terror' in the style of the USA, it has focused on building up its tools to provide long-term counter-terrorism assistance and capability-building aid, whilst adopting agreements with third states. As demonstrated in this book, the EU has played a complex and diverse role in counter-terrorism around the world. In short, the EU has become one of the most significant, but certainly not the most visible. actors in global counter-terrorism. Nevertheless, this role has been shared with the member states, which also explains why the EU has sometimes remained in the background and has assumed long-term tasks that do not always make news headlines.

While the oft-neglected role of the EU is empirically interesting and worthy of research in itself, much more important has been the rapidity of the EU's progress and how and why this has occurred. Looking at how and why entails drawing upon theoretical tools developed elsewhere to see if they help shed

Conclusion 131

light on this area of global politics. Furthermore, it is possible and illuminating to see if similar patterns can be discerned in the processes of European integration in various policy areas. The events of 9/11, subsequent terrorist attacks, and the general threat of jihadist terrorism have been of undeniable importance in expanding the EU's security role overall. Indeed, the threat of terrorism has changed the EU and its governance substantially. The questions of how and why the EU has grown as a counter-terrorism actor have been addressed through our examination of actorness. What has been particularly significant is the shift towards terrorism being seen as an issue to be dealt with domestically to being an area where the EU has a high level of competence. whilst member states and their citizens want the EU to do more. This has likely occurred because of the material characteristics of the terrorist threat itself, which has indeed become increasingly transnational rather than simply domestic. However, it is important to also highlight the role of the EU, notably the European Commission, in framing it as such. The member states also appear to have increasingly accepted this perception. Thus, this book has shown how issues can move from national to supranational governance, which has also contributed to the theoretical literature on European integration.

To conclude, this study has demonstrated how much the EU has developed as a counter-terrorism actor since 9/11 and has examined the reasons for this growth. The EU has become significant in many of the main conflict zones of the globe, albeit not in a uniform manner. Overall, the growth in the EU's global counter-terrorism role has been remarkable. This chapter now considers two important challenges that are likely to influence the development of the EU's global counter-terrorism role in the near future. They are the rise of right-wing terrorism and the consequences of Brexit for the relations between the EU, the UK, and the USA.

TWO CHALLENGES AND THEIR LIKELY IMPACT ON THE EU'S GLOBAL COUNTER-TERRORISM ROLE

Terrorist threats are forever evolving, and much has changed since 2001. Rapoport's (2001) 'wave theory' has shown how technology has often played a permissive role in such developments. Al-Qaeda declined and Daesh overtook it as the main jihadist threat, although it has now been considerably weakened. The main concern now stems from the fact that Daesh has become fragmented and some of those who travelled to fight for it may have been trying to return home. Small numbers of Muslims have been attracted to the world views of jihadist groups. Some have travelled to fight abroad or carried out attacks against Western states. It is in this changing context that efforts to combat terrorism have taken place. For Rapoport (2001), each 'wave' of terrorism lasted about a generation. Perhaps this one has now run its course

with the dream of a caliphate and Islamic resurgence largely extinguished. Now it must seem like the obstacles to a caliphate are all but insurmountable and high levels of support from the abstract 'community' have not been forthcoming. Each wave appears to last just as long as there is a glimmer of hope. The present wave may have even been prolonged by what some have seen as a counter-productive over-reaction of the USA and its allies after 9/11, including the fateful invasion of Iraq. Through a complex set of events, including the Syrian Civil War, the invasion of Iraq has ultimately led to the emergence of Daesh – a threat that has ended up causing significant difficulties for Europe.

If indeed the demise of Daesh in the Middle East acts as the harbinger of the end of the fourth wave of modern international terrorism (Rapoport, 2001), then this book will have successfully covered the majority of the arc of the EU's activities against terrorism – especially jihadist terrorism – since 9/11. Terrorism will not just disappear after one of its waves has ended – one of the great misunderstandings of the 'war on terror'. More likely, terrorist groups disintegrate as they are hunted down, there is in-fighting within groups and future generations take little interest in carrying on the fight. Looking towards the future, there has been increasing concern about the growth in right-wing terrorism in Europe, notably as a reaction to the 'refugee crisis' and the terrorist attacks that have marked 2015–16 (Léonard and Kaunert, 2021c).

Right-Wing Terrorism and the EU

Right-wing terrorism has been on the rise and growing faster than other types of terrorism (Koehler, 2019). In Western states, the far-right accounted for 17.2 per cent of attacks in 2018 and had risen by 320 per cent over the five previous years (Global Terrorism Index, 2019). Some established terrorism studies scholars have speculated that the 'fifth wave' of terrorism might come from the right (Koehler, 2019). It is open to question whether the EU has hitherto taken this upcoming threat seriously enough and has taken all necessary measures that may become necessary to address it effectively.

Right-wing terrorism derives from a vision of the superiority of one societal group over others. In this specific instance, it is likely that the individuals and groups in question will focus on Muslims, Jews, and regional or state-specific minorities. Moreover, there are certain strategic and tactical trends. On the level of strategy, right-wing terrorists have pioneered the approach of 'lead-erless resistance', which refers to individuals or small groups engaging in terrorism unrelated to a larger group (Kaplan, 1997; Michael, 2012). Questions abound as to whether this is an effective approach in terms of goals, but these individuals and small groups are more than just a mere nuisance; there is notably potential for violent spirals amongst societal groups. Also, the Internet can likely be seen as a game changer for leaderless resistance as more

Conclusion 133

people can access relevant material, such as extreme ideas or instructions for making weapons (Berger, 2019). Recent high-profile attacks by right-wing terrorists, such as Brenton Tarrant, Dylann Roof, and Anders Behring Breivik were carried out by single actors. However, it appears that they had engaged with other individuals – notably online – who offered them support, encouragement, and direction. Rarely have right-wing terrorists organised in large groups or lasted for a long time, at least not in Europe. Indeed, Koehler has pointed to an increasing trend of small unit tactics among German right-wing groups since 2000, whilst 73.2 per cent of groups existed for no longer than a year (Koehler, 2014). In terms of tactics, a wide range of methods have been used, but explosives and arson appear to have dominated in Germany (Koehler, 2014: 54). Shootings have been particularly noticeable in some of the aforementioned recent high-profile attacks. Piazza (2009: 67) has also highlighted the increasing violence of right-wing terrorist groups, which he has found to be generally more violent than nationalist and leftist groups, with 5.1 victims per attack on the basis of data between 1998 and 2005. Finally, it is noteworthy that there has been a tendency not to take responsibility for attacks, which is somewhat surprising since terrorism is usually seen as a form of political communication. It also further renders distinction from hate crime more difficult (Koehler, 2019: 9).

Right-wing terrorism is not a new issue in Europe. Indeed, not only is there abundant inspiration from the fascist regimes of the past, but there have been several right-wing terrorist groups of note since 1945, such as during Italy's Years of Lead, Germany's National Socialist Underground, and neo-Nazi skinhead movements across the continent, the origins of which have been traced back to the UK (Bjorgo, 1995; Hoffman, 1982; Manthe, 2018). The 1990s have typically been seen as a period of declining terrorism. Since 9/11, the issue of major concern has been jihadist terrorism (Blackbourn et al., 2019). Right-wing terrorism in Europe mostly seems to have been of lower importance until the 2010s. However, the difficulty of separating terrorism and hate crime may have played a role by masking the issue to some extent. Anders Behring Breivik's attack in 2011 was particularly influential in generating concern about emerging right-wing terrorism in Europe. The global financial crisis, the ensuing austerity, the 'refugee crisis', an upsurge in populism, and the increasingly blurry lines between centre-right and extreme right all seem to be auspicious circumstances for its rise. Schuurman (2019) has shown that there has been only limited research on right-wing terrorism in general. However, this gap seems to be particularly acute in relation to Europe. Most literature on this issue has been either historical or focused on a limited set of states. There has been some research on Germany as it has come to terms with the upsurge of right-wing terrorism, whilst Pisoiu (2015) has applied subcultural theory to both jihadi and right-wing radicalisation in the country, arguing for a greater focus on agency and subcultural context. In addition to Germany, Scandinavia has been of interest to scholars, perhaps spurred by Breivik, but more generally by powerful far-right groups in the region (Hardy, 2019; Ravndal, 2018).

Furthermore, the linkages between immigration and terrorism in Europe have been explored. For instance, McAlexander has argued that, between 1980 and 2004, increases in migration were positively correlated with increases in right-wing terrorism, notably because the arrival of migrants aggravated the grievances of those on the radical right. In this perspective, terrorism has been ultimately seen as a strategic choice (McAlexander, 2020). Some scholars have speculated that right-wing terrorism might become an increasingly transnational phenomenon, albeit probably not to the same extent as jihadist terrorism (Caiani and Kroll, 2014; Stevenson, 2019). What can be observed here is how patchy and somewhat limited the knowledge and understanding of right-wing terrorism in Europe remains, which adds to the challenges inherent to addressing it. However, this is not the only challenge to be faced by the EU in counter-terrorism in the next few years. It will also have to deal with the consequences of Brexit, which are also intertwined with possible tensions in the transatlantic relationship.

The EU, the UK, and the USA after Brexit

The UK's departure from the EU on 31 January 2020 is likely to have significant and mostly unwelcome consequences for both the EU and the UK in the area of counter-terrorism. In general, it introduces a significant number of hurdles that will need to be overcome before a substantial level of cooperation becomes again a possibility. In addition, the UK used to act as a driving force for the development of EU counter-terrorism when it was still a member state. Its departure from the EU might therefore lead to a loss of impetus in this policy area. The proponents of Brexit may have been somewhat over-confident of what – and at what cost – the UK can achieve on its own, whist the EU faces some challenges after losing such an important contributor to its counter-terrorism cooperation. This may lead to the creation of security gaps.

While not the only important actor in driving the EU's counter-terrorism policy – the Commission and other member states have played key roles at times – the UK has in many ways been vital to the development of EU counter-terrorism at crucial points (Parliament.uk, 2016a; Rees, 2008). In fact, it can be argued that the UK has probably been the most important member state in pushing the EU counter-terrorism agenda forward (HM Government, 2017c). Like other member states, the UK recognised that cooperation at the EU level was urgent after 9/11, not least because a number of the hijackers had passed through London to reach the USA prior to the attacks. Subsequently,

Conclusion 135

the UK played a major role in several ways: it showed leadership in the area; it exported or uploaded domestic measures to the EU; and it consistently championed EU–US cooperation and agreements in the field of counter-terrorism.

First, the UK demonstrated leadership in counter-terrorism with its role in bringing about the EU's Counter-Terrorism Strategy during its presidency in the second half of 2005 – a document that is strikingly similar to the UK's own counter-terrorism strategy. As an interviewee pointed out in 2010, the EU's Counter-Terrorism Strategy is the UK's with just a few different words (Interview with a member state representative, June 2010). Second, UK efforts to upload domestic security measures to the EU level were evident in the cases of the DRD in 2006 (Ripoll Servent, 2013) and the EU PNR system, which was agreed in 2016 (itself originally exported from the USA). The UK had followed the US lead and started collecting and using PNR data in 2004 – many years before any other member state (Bakowski and Voronova, 2015). The UK (along with the European Commission) pushed for an EU PNR system for a number of years. Despite lukewarm support to begin with, amid scepticism and concern about mass data retention, other member states were slowly brought on board due to an apparent recognition of the benefits of collecting the data and the willingness to take action following several attacks (Bakowski and Voronova, 2015; Pawlak, 2009a). Third, the UK consistently supported EU-US counter-terrorism cooperation, such as in the cases of the several EU-US PNR Agreements. De Goede (2008a: 161) once commented that '[i]t becomes clear that, with respect to a number of policies that play a key role in preemptive security practice, including criminalizing terrorist support, data retention, and asset freezing, the European Union is world leader rather than reluctant follower'. Thus, the USA and the UK have been two of the key players pushing for more EU counter-terrorism cooperation. In the last few years, the EU has altered its previously strict data protection standards as it has concluded more agreements with the USA, notably as a result of windows of opportunity being exploited by the UK and others to push for more cooperation at the EU level. To sum up, the UK has evidently been a leader in the area of counter-terrorism in general, but it has also been important in pushing for controversial measures and cooperation with third states. With Brexit, the UK has lost its influence as an EU member state, whilst the EU has lost the driving force of the UK in this policy area.

For the UK, David Davis, the then Secretary of State for Exiting the European Union, identified 'maintaining the strong security co-operation we have with the EU' as one of the government's four main objectives in the negotiations with the EU in 2020 (Hansard, 2016). On the EU side, the appointment of Sir Julian King as the Commissioner for the Security Union following the resignation of Lord Hill after the EU membership referendum in the UK demonstrated the importance of the UK to the other EU member states

in the area of security. With Brexit, the UK has lost both the ability to influence the EU's direction and access to a number of EU agencies and databases, some of which have been considered important to the UK security agencies. As for the EU, the development of its counter-terrorism cooperation may lose some momentum, given the important role that the UK used to play in that policy area, including its contributions to the EU's security information databases. It is possible for the EU and the UK to forge new forms of counter-terrorism cooperation, but this will require creativity and goodwill from all parties, whilst the results may be sub-optimal compared to what could be achieved when the UK was still an EU member state.

Finally, it is important to note that Brexit may also have an impact on the EU-US relationship on counter-terrorism, given the bridging role that the UK traditionally played between continental Europe and the USA when it was still a member of the EU. In general, EU-US relations have improved since Trump's departure from the White House. Nevertheless, Europe has appeared to decline in strategic importance for the USA in recent decades, in parallel with America's pivot to Asia. This suggests that, although EU-US counter-terrorism cooperation may well be maintained, it is unlikely to grow significantly.

AVENUES FOR FUTURE RESEARCH

There are many possibilities for future research stemming from this work, both theoretical and empirical. On the theoretical front, how does the analytical framework map onto other areas of EU cooperation outside of counter-terrorism? Can such an approach help explain why certain issues have been moved to the EU level? Empirically, at this stage, it seems important to further assess EU counter-terrorism in relation to Daesh, which has been largely overlooked to date. How much of a role has the EU played, especially in Iraq and Syria? EU activity in Afghanistan has also remained under-researched. It is not clear why this issue has been so neglected, especially given the scale of the EU's engagement. Perhaps the tools of international relations could be used to examine the EU's relations with Afghanistan, not only in the area of counter-terrorism, but also in relation to aid and trade. This is a fascinating case of significant EU engagement that has remained largely under-researched. Literature on the EU's engagement with Iraq and Syria has also remained limited, possibly because of the continued domination of state-based approaches.

Additionally, a significant amount has been written on the EU–US relationship and the influence that Washington has had on EU counter-terrorism, but what about the main players within the EU? For example, not much attention has been given to the UK's role in the development of the external dimension

Conclusion 137

of EU counter-terrorism. It is now a particularly interesting issue in the wake of Brexit. To date, there have only been passing references to the UK's role in this area, although it has been amongst the main member states pushing for greater counter-terrorism cooperation at the EU level. Leading on from this, how has the 'war on terror' influenced EU member states? There has been limited research on states, such as Germany, which have a history of strong data protection safeguards, and on the ways in which such protection has been eroded as the 'war on terror' has progressed. What influence have attacks such as those that occurred in France in 2015 had on the EU? Has France now become a more active player in EU counter-terrorism due to its own experiences? This is important in relation to Brexit because it raises the question of which member states, if any, will lead in the development of AFSJ cooperation, including counter-terrorism, following the departure of the UK from the EU? More generally, now with a longer-term view, how exactly and how much has European integration been influenced by terrorism? What role have the EU institutions played, possibly by taking advantage of 'windows of opportunities' that have opened following terrorist attacks?

Many possible avenues for future research follow from this book. Having said that, it can be concluded that the EU has become a much more important actor in counter-terrorism on the global stage than it was at the start of the twenty-first century. Terrorism has been a significant factor in fostering greater security cooperation amongst European states and has changed the course of EU integration, particularly since 2001. How will these decades marked by significant crises be viewed in decades to come in relation to European integration? If the EU weathers the current storms, it will emerge a more influential actor than it was before.

- Abdelsamad, H. (2018), 'Terrorism and Counterterrorism in the EU: Overestimated Threat, Oversimplified Approach', accessed 14 April 2020 at https://docresearch.org/2018/08/counterterrorism/#.
- Ackerman, G. A. (2019), 'Chemical, Biological, Radiological and Nuclear (CBRN) Terrorism', in Silke, A. (ed.), *Routledge Handbook of Terrorism and Counterterrorism*, London: Routledge, pp. 240–52.
- Acosta, D. (2009), 'The Good, the Bad and the Ugly in EU Migration Law: Is the European Parliament Becoming Bad and Ugly?' (The Adoption of Directive 2008/15: The Returns Directive), *European Journal of Migration and Law*, 11(1), 19–39.
- Afghanistan Ministry of Finance (2009), Donor Financial Review (No. 1388).
- Aizpeolea, L. R. (2018, 30 April), 'ETA terminó en octubre de 2011', *El País*, accessed 14 April 2020 at https://elpais.com/politica/2018/04/30/actualidad/1525111923 662842.html.
- Albert, E. (2017), 'ASEAN: The Association of Southeast Asian Nation', *Council of Foreign Relations*, accessed 14 April 2020 at https://www.cfr.org/backgrounder/asean-association-southeast-asian-nations.
- Alcantara, C. (2017, 17 July), '46 Years of Terrorist Attacks in Europe, Visualized', Washington Post, accessed 20 June 2018 at https://www.washingtonpost.com/ graphics/world/a-history-of-terrorism-in-europe/.
- Almqvist, J. (2008), 'A Human Rights Critique of European Judicial Review: Counter-terrorism Sanctions', *International and Comparative Law Quarterly*, 57(2), 303–31.
- Alvarez, J. E. (2003), 'Hegemonic International Law Revisited', *The American Journal of International Law*, 97(4), 873–88.
- Amadio Viceré, M. G. (2015), 'From 9/11 to Da'esh: What Role for the High Representative in the External Dimension of EU Counter-Terrorism Policies?' (IAI Istituto Affari Internazionali).
- Amadio Viceré, M. G. (2016), 'The Roles of the President of the European Council and the High Representative in Leading EU Foreign Policy on Kosovo', *Journal of European Integration*, 38(5), 557–70.
- Amadio Viceré, M. G. (2018), The High Representative and EU Foreign Policy Integration, Basingstoke: Palgrave Macmillan.
- Andersson, T. et al. (2003), 'EU Blacklisting: The Renaissance of Imperial Power, but on a Global Scale', European Business Law Review, 14(2), 111–41.
- Andreatta, F. (2011), 'The European Union's International Relations: A Theoretical View', in, Hill, C. and Smith, M. (eds), *International Relations and the European Union*, Oxford: Oxford University Press, 21–43.
- Archik, K. (2012), 'US–EU Co-operation against Terrorism'. CRS Report for Congress, accessed 7 January 2012 at http://www.fas.org/sgp/crs/row/RS22030.pdf.
- Argano, M. E. (2018), 'NATO's Impact on the European International Security and Counterterrorism Policy' (EULOGOS Athena).

Argomaniz, J. (2008), 'Counter-Terrorism Policy-Making in the EU'. PhD thesis, University of Nottingham.

- Argomaniz, J. (2009a), 'Post-9/11 Institutionalisation of European Union Counter-terrorism: Emergence, Acceleration, and Inertia', *European Security*, 18(2), 151–72.
- Argomaniz, J. (2009b), 'When the EU is the "Norm-taker": The Passenger Name Records Agreement and the EU's Internalization of US Border Security Norms', *Journal of European Integration*, 31(1), 119–36.
- Argomaniz, J. (2010), 'Before and After Lisbon: Legal Implementation as the "Achilles heel" in EU Counter-terrorism', *European Security*, 19(2), 297–316.
- Argomaniz, J. (2011), The EU and Counter-Terrorism: Politics, Polity and Policies after 9/11, London: Routledge.
- Argomaniz, J., Bures, O. and Kaunert, C. (2015), 'A Decade of EU Counter-Terrorism and Intelligence: A Critical Assessment', *Intelligence and National Security*, 30(2/3), 191–206.
- ASEAN Convention on Counter Terrorism (2007), accessed 14 April 2020 at https://asean.org/?static_post=asean-convention-on-counter-terrorism.
- ASEAN Declaration on Joint Action to Counter Terrorism (5 November 2001), accessed 14 April 2020 at https://asean.org/?static_post=2001-asean-declaration-on-joint-action-to-counter-terrorism.
- ASEAN Declaration on Transitional Crime (20 December 1997), accessed 14 April 2020 at https://asean.org/?static_post=asean-declaration-on-transnational-crime -manila-20-december-1997.
- ASEAN Plan of Action in Combating Transnational Crime (20 September 2017), accessed 14 April 2020 at https://asean.org/wp-content/uploads/2012/05/ASEAN -Plan-of-Action-in-Combating-TC_Adopted-by-11th-AMMTC-on-20Sept17.pdf.
- ASEAN Work Programme to Implement the ASEAN Plan of Action to Combat Transnational Crime (17 May 2002), accessed 14 April 2020 at https://asean.org/?static_post=work-programme-to-implement-the-asean-plan-of-action-to-combat -transnational-crime-kuala-lumpur-17-may-2002.
- Aspenia (2015), 'EU Counterterrorism Efforts and the Role of High Representative Mogherini', *Aspenia Online*, accessed 7 March 2015 at https://aspeniaonline.it/articolo_aspenia/eu-counterterrorism-efforts-and-the-role-of-high-representative -mogherini/.
- Association of European Airlines (2014), 'Transfer of PNR Data from Europe to Third Countries', accessed at https://www.statewatch.org/media/documents/news/2014/oct/eu-iata-pnr-transfer.pdf.
- Aversano-Stabile, A., Lasconjarias, G. and Sartori, P. (2018), 'NATO–EU Cooperation to Project Stability'. *Istituto Affari Internazionali- IAI*, accessed 14 April 2020 at https://www.iai.it/en/pubblicazioni/nato-eu-cooperation-project-stability.
- Bacon, T. and Grimm Arsenault, E. (2017), 'Al Qaeda and the Islamic State's Break: Strategic Strife or Lackluster Leadership?', *Studies in Conflict and Terrorism*, 42(3), 1–35.
- Baker-Beall, C. (2016), *The European Union's Fight against Terrorism: Discourse, Policies, Identity*, Manchester: Manchester University Press.
- Bakker, E. (2006), 'Difference in Terrorist Threat Perceptions in Europe', in Mahncke, D. and Monar, J. (eds), *International Terrorism: A European Response to a Global Threat*?, Brussels: Peter Lang, pp. 47–62.
- Bakker, E. (2015), 'EU Counter-radicalization Policies: A Comprehensive and Consistent Approach?', *Intelligence and National Security*, 30(2–3), 281–305.

- Bakowski, P. and Voronova, S. (2015), 'The Proposed EU Passenger Name Records (PNR) Directive: Revived in the New Security Context', accessed 24 February 2022 at https://www.europarl.europa.eu/EPRS/EPRS-Briefing-554215-The-EU-PNR-Proposal-FINAL.pdf.
- Balfour, R., Carta, C. and Raik, C. (2015), *The European External Action Service and National Foreign Ministries: Convergence or Divergence?*, Aldershot: Ashgate Publishing.
- Balzacq, T. (2005), 'The Three Faces of Securitization: Political Agency, Audience and Context', European Journal of International Relations, 11(2), 171–201.
- Bantekas, I. (2003), 'The International Law of Terrorist Financing', *The American Journal of International Law*, 97(2), 315–33.
- Bar-Yaacov, N. (2015, 27 November), 'What if Isis Launches a Chemical Attack in Europe?', *The Guardian*, accessed 5 August 2019 at https://www.theguardian.com/global/commentisfree/2015/nov/27/isis-chemical-attack-europe-public.
- Bates, S. (1999, 18 November), 'Solana Calls for Voice to Match EU Might', *The Guardian*, accessed 14 April 2020 at https://www.theguardian.com/world/1999/nov/18/eu.unitednations.
- BBC (2001a), 'Text of Bush's act of war statement', 12 September.
- BBC (2001b), 'Washington presses EU in terror war', 22 October.
- BBC (2003), 'Terror attack "a matter of time", 17 June.
- BBC (2005), 'The Hamburg Connection', 19 August.
- BBC (2011), '2004: Madrid train attacks', 7 September.
- BBC (2013a), 'US marks 9/11 attacks anniversary', 12 September.
- BBC (2013b), 'Iraqi Kurds mark 25 years since Halabja gas attack', 16 March.
- BBC (2016), 'Alexander Litvinenko: Profile of murdered Russian spy', 21 January.
- BBC (2016), 'Staying in EU "exposes UK to terror risk", says Iain Duncan Smith', 21 February.
- BBC (2017), 'Inside Europol', 15 February.
- BBC (2017), 'Who are Britain's jihadists?', 12 October, accessed 15 June 2017 at http://www.bbc.co.uk/news/uk/uk-320026985.
- BBC (2018), 'Russian spy: What happened to Sergei and Yulia Skripal?', 27 September, accessed 2 January 2018 at https://www.bbc.co.uk/news/uk-43643025.
- BBC News (2018), 'Syria conflict: US officials withdraw troops after IS "defeat", 19 December.
- BBC (2019a), 'How many IS foreign fighters are left in Iraq and Syria?', 19 February, accessed 9 July 2019 at https://www.bbc.co.uk/news/world-middle-east-47286935.
- BBC (2019b), 'Sri Lanka reels from deadly attacks', 19 April, accessed 20 May 2019 at https://www.bbc.co.uk/news/live/world-asia-48002165.
- BBC (2019c), 'Iran developing nuclear-capable missiles, European powers warn UN', accessed 9 March 2022 at https://www.bbc.co.uk/news/world-middle-east -50671003.
- BBC News (2021), 'US-led coalition against IS ends combat mission in Iraq', 10 December.
- Béland, D. and Howlett, M. (2016), 'The Role and Impact of the Multiple-Streams Approach in Comparative Policy Analysis', *Journal of Comparative Policy Analysis: Research and Practice*, 18(3), 221–7.
- Belaza, M. C. (2018), 'ETA y el coche bomba: los atentados más sangrientos', *El País*, 30 April, accessed 14 April 2020 at https://elpais.com/politica/2018/04/29/actualidad/1525026725 974459.html.

- Berger, J. M. (2019), 'The Strategy of Violent White Supremacy is Evolving', *The Atlantic*, accessed 2 March 2020 at https://www.theatlantic.com/ideas/archive/2019/08/the-new-strategy-of-violent-white-supremacy/595648/.
- Berthelet, P. (2017), 'How the European Union is Making Major Strides Fighting Terrorism', *The Conversation*, accessed 14 April 2020 at https://theconversation.com/how-the-european-union-is-making-major-strides-fighting-terrorism-82866.
- Beyer, C. (2008), 'The European Union as a Security Policy Actor: The Case of Counterterrorism', *European Foreign Affairs Review*, 13(3), 293–315.
- Bickerton, C. J., Dermot, H. and Uwe, P. (2015), 'The New Intergovernmentalism: EUropean Integration in the Post-Maastricht Era', *JCMS: Journal of Common Market Studies*, 53(4), 703–22.
- Biersteker, T. J. (2009–10), 'Targeted Sanctions and Individual Human Rights', *International Journal*, 65(1), 99–117.
- Biersteker, T. and Eckert, S. (2009), Addressing Challenges to Targeted Sanctions: An Update of the 'Watson Report', Watson Institute for International Studies, Brown University, accessed 10 January 2011 at http://www.watsoninstitute.org/pub/2009_10_targeted_sanctions.pdf.
- Bigo, D. (2006), 'Internal and External Aspects of Security', *European Security*, 15(4), 385–404.
- Bigo, D. and Tsoukala, A. (2008), Terror, Insecurity and Liberty: Illiberal Practices of Liberal Regimes after 9/11, London: Routledge.
- Bildt, C. (2021), 'Afghanistan: How It Could Have Been Different', London: European Council on Foreign Relations.
- Biscop, S. (2008), 'The European Security Strategy in Context: A Comprehensive Trend', in Biscop, S. and Anderson, J. (eds), *The EU and the European Security Strategy: Forging a Global Europe*, London: Routledge, pp. 5–20.
- Biscop, S. (2010), 'The ENP, Security, and Democracy in the Context of the European Security Strategy', in Whitman, R. and Wolff, S. (eds), *The European Neighbourhood Policy in Perspective: Context, Implementation and Impact*, Basingstoke: Palgrave Macmillan, pp. 73–88.
- Bjorgo, T. (ed.) (1995), Terror from the Extreme Right, Abingdon: Routledge.
- Blackbourn, J., McGarrity, N. and Roach, K. (2019), 'Understanding and Responding to Right Wing Terrorism', *Journal of Policing, Intelligence, and Counter-Terrorism*, 14(3), 183–90.
- Boin, A., Ekengren, M. and Rhinard, M. (2013), *The European Union as Crisis Manager*, Cambridge: Cambridge University Press.
- Borelli, M. (2017), 'ASEAN Counter-Terrorism Weakness', Counter Terrorist Trends and Analyses, 9(9), 14–20.
- Bossong, R. (2008a), 'The Action Plan on Combating Terrorism: A Flawed Instrument of EU Security Governance', *Journal of Common Market Studies*, 46(1), 27–48.
- Bossong, R. (2008b), 'The EU's Mature Counterterrorism Policy A Critical Historical and Functional Assessment', LSE Challenge Working Paper.
- Bossong, R. (2013), The Evolution of EU Counter-terrorism: European Security Policy after 9/11, London: Routledge.
- Bossong, R. (2021), 'The Next Steps for EU Counterterrorism Policy', SWP Comment 2021/C 20, Berlin: SWP.
- Bossong, R. and Hegemann, H. (eds) (2015), European Civil Security Governance: Diversity and Cooperation in Crisis and Disaster Management, Basingstoke: Palgrave Macmillan.

- Brader, T. (2011), 'The Political Relevance of Emotions: "Reassessing" Revisited', *Political Psychology*, 32(2), 337–46.
- Brattberg, E. and Rhinard, M. (2012), 'The EU as a Global Counter-terrorism Actor in the Making', *European Security*, 21(4), 557–77.
- Bretherton, C. and Vogler, J. (2006), *The European Union as a Global Actor*, London: Routledge.
- Bretherton, C. and Vogler, J. (2013), 'A Global Actor Past Its Peak?', *International Relations*, 27(3), 375–90.
- Brouwer, E. (2009), 'Towards a European PNR System? Questions on the Added Value and the Protection of Fundamental Rights', accessed 24 February 2022 at https://www.europarl.europa.eu/thinktank/en/document/IPOL-LIBE ET(2009)410649.
- Brown, D. (2010), *The European Union, Counter Terrorism and Police Cooperation,* 1992–2007, Manchester: Manchester University Press.
- Brown, T. (2004), 'Pop Music and Politics: Skinheads and "Nazi Rock" in England and Germany', *Journal of Social History*, 38(1), 157–78.
- Bunn, M. and Wier, A. (2006), 'Terrorist Nuclear Weapon Construction: How Difficult?', *The ANNALS of the American Academy of Political and Social Science*, 607, 133–49.
- Bunyan, T. (1993), 'Trevi, Europol and the European State', available at https://www.statewatch.org/media/documents/news/handbook-trevi.pdf.
- Bures, O. (2006), 'EU Counter-Terrorism Policy: A "Paper Tiger"?', *Terrorism and Political Violence*, 18(1), 57–78.
- Bures, O. (2008), 'Europol's Fledgling Counter-Terrorism Role', *Terrorism and Political Violence*, 20(4), 498–517.
- Bures, O. (2010a), 'Perceptions of the Terrorist Threat among EU Member States', Central European Journal of International and Security Studies, 4(1), 51–80.
- Bures, O. (2010b), 'EU's Fight against Terrorist Finances: Internal Shortcomings and Unsuitable External Models', *Terrorism and Political Violence*, 22(3), 418–37.
- Bures, O. (2010c), 'Eurojust's Fledgling Counterterrorism Role', *Journal of Contemporary European Research*, 6(2), 236–55.
- Bures, O. (2011), EU Counterterrorism Policy A Paper Tiger?, Farnham: Ashgate.
- Bures, O. (2016), 'Intelligence Sharing and the Fight against Terrorism in the EU: Lessons Learned from Europol', *European View*, 15(1), 57–66.
- Bures, O. (2018), 'EU's Response to Foreign Fighters: New Threat, Old Challenges', *Terrorism and Political Violence*, 32(4), 789–806.
- Burke, E. (2014), 'Game Over? The EU's Legacy in Afghanistan', FRIDE, accessed 24 February 2022 at http://ibdigital.uib.es/greenstone/collect/cd2/index/assoc/fride002/8.dir/fride0028.pdf.
- Bush, G. W. (2001a), 'Address to a Joint Session of Congress and the American People', *The Guardian*, 21 September, accessed 20 June 2018 at https://www.theguardian.com/world/2001/sep/21/september11.usa13.
- Bush, G. W. (2001b), 'Address to a Joint Session of Congress Following 9/11 Attacks', American Rhetoric, accessed 24 February 2022 at https://www.americanrhetoric.com/speeches/gwbush911jointsessionspeech.htm.
- Buzan, B. (1991), People, States and Fear: An Agenda for International Security Studies in the Post-Cold War Era, London: Harvester Wheatsheaf.
- Buzan, B. and Wæver, O. (2003), *Regions and Powers: The Structures of International Security*, Cambridge: Cambridge University Press.

- Buzan, B. and Waever, O. (2009), 'Macrosecuritization and Security Constellations: Reconsidering Scale in Securitization Theory', *Review of International Studies*, 35(2), 253–76.
- Buzan, B., Wæver, O. and de Wilde, J. (eds) (1998), Security: A New Framework for Analysis, Boulder, CO; London: Lynne Rienner.
- Byman, D. (2015), Al Qaeda, the Islamic State, and the Global Jihadist Movement, New York: Oxford University Press.
- Byman, D. (2017), 'Explaining Al Qaeda's Demise', *Journal of Politics*, 79(3), 1106–17.
- Byman, D. (2018), 'What Happens When ISIS goes Underground?', *Brookings Institute*, accessed 14 December 2018 at https://www.brookings.edu/blog/markaz/2018/01/18/what-happens-when-isis-goes-underground/.
- Cahn, O. (2017), 'Brexit and the Future of European Criminal Law: A French Perspective', *Criminal Law Forum*, 28, 301–9.
- Caiani, M. and Kroll, P. (2014), 'The Transnationalisation of the Extreme Right and the Use of the Internet', *International Journal of Comparative and Applied Criminal Justice*, 39(4), 331–51.
- Cameron, F. (2007), 'Transatlantic Relations and Terrorism', in Spence, D. (ed.), *The European Union and Terrorism*, London: John Harper.
- Cameron, F. (2013), 'The Evolution of EU–Asia Relations: 2001–2011', in Christiansen, T., Kirchner, E. and Murray, P. (eds), *The Palgrave Handbook of EU–Asia Relations*, Basingstoke: Palgrave Macmillan.
- Cannataci, J. and Bonnici, J. (2009), 'The UK 2007–2008 Data Protection Fiasco: Moving On From Bad Policy and Bad Law?', *International Review of Law, Computers & Technology*, 23(1/2), 47–76.
- CATO Institute (2017), CATO Handbook for Policymakers, Washington, DC: CATO Institute.
- CBS News (2003, 10 June), 'U.S.: Al Qaeda WMD Risk Remains', accessed 10 September 2019 at https://www.cbsnews.com/news/us-al-qaeda-wmd-risk-remains/.
- Chaliand, G. and Blin, A. (2016), *The History of Terrorism: From Antiquity to ISIS*, Berkeley: University of California Press.
- Chalk, P. (1996a), West European Terrorism and Counter-terrorism: The Evolving Dynamic, Basingstoke: Macmillan.
- Chalk, P. (1996b), 'The Maastricht Third Pillar and Liberal Democratic Acceptability', in Chalk, P. (ed.), *West European Terrorism and Counter-terrorism: The Evolving Dynamic*, London: Palgrave Macmillan, pp. 143–63.
- Chalk, P. (1996c), 'The Nature of Contemporary Terrorism: Problems of Definition', in Chalk, P. (ed.), *West European Terrorism and Counter-terrorism: The Evolving Dynamic*, London: Palgrave Macmillan, pp. 9–24.
- Chivers, C. J. (2015, 6 October), 'What an ISIS Chemical Threat Did to One Syrian Family', *The New York Times*, accessed 16 April 2020 at https://www.nytimes.com/2015/10/07/world/middleeast/syrian-familys-agony-raises-specter-of-chemical-warfare.html.
- Ciascai, G. (2011), 'The Impact of the Institutional Innovations Introduced by the Treaty of Lisbon on the European Union's External Action', *CES Working Papers*, 3(2), 182–7.
- CNN (2001, 18 September), 'Chirac: France Stands with US against Terrorism', accessed 16 April 2020 at http://edition.cnn.com/2001/US/09/18/ret.bush.chirac/ index.html.

- CNN (2010, 13 April), 'Obama Hosts Leaders at Nuclear Summit', accessed 3 January 2018 at http://edition.cnn.com/2010/POLITICS/04/12/nuclear.security.summit/ index.html.
- CNN (2016), 'Rumsfeld/Knowns', available at https://www.youtube.com/watch?v=REWeBzGuzCc.
- CNN (2018, 28 November), '17 Years In, Afghan War at a "Stalemate", accessed 17 January 2019 at https://edition.cnn.com/2018/11/27/politics/us-afghan-war-stalemate/index.html.
- Combs, C. C. (2018), Terrorism in the Twenty-first Century, London: Routledge.
- Commission of the European Communities (2003), Communication of the European Commission to the Council and to the European Parliament The European Union and the United Nations: The Choice of Multilateralism, COM 2003 (526), 10 September, Brussels: Commission of the European Communities.
- Commission of the European Communities (2009), 'Communication from the Commission to the European Parliament and the Council on Strengthening Chemical, Biological, Radiological and Nuclear Security in the European Union an EU CBRN Action Plan', COM 2009 (273), 24 June, Brussels: Commission of the European Communities.
- Congressional Research Service (2018), 'State Sponsors of Acts of International Terrorism—Legislative Parameters: In Brief', available at https://sgp.fas.org/crs/terror/R43835.pdf.
- Coolsaet, R. (2010), 'EU Counterterrorism Strategy: Value Added or Chimera?', International Affairs, 86(4), 857–73.
- Coolsaet, R. and Renard, T. (2018), 'The Homecoming of Foreign Fighters in the Netherlands, Germany and Belgium: Policies and Challenges', *International Centre for Counter-Terrorism*, accessed 14 December 2018 at https://icct.nl/publication/the-homecoming-of-foreign-fighters-in-the-netherlands-germany-and-belgium-policies-and-challenges/.
- Cosgrove, C. and Twitchett, K. (eds) (1970), The New International Actors: The UN and the EEC, London: Macmillan.
- Council of the European Union (2001), 'Special Council Meeting, General Affairs', 12 September, Doc. 11795/01 (Presse 318), Brussels: Council of the European Union.
- Council of the European Union (2002), 'Adoption of the Programme to Improve Cooperation in the European Union for Preventing and Limiting the Consequences of Chemical, Biological, Radiological or Nuclear Terrorist Threats', 21 November, Doc. 14627/02, Brussels: Council of the European Union.
- Council of the European Union (2003), 'Fight Against the Proliferation of Weapons of Mass Destruction EU Strategy Against Proliferation of Weapons of Mass Destruction', 10 December, Doc. 15708/03, Brussels: Council of the European Union.
- Council of the European Union (2004a), 'EU Plan of Action on Combating Terrorism', Doc. 10586/04, Brussels: Council of the European Union.
- Council of the European Union (2004b), 'Council Decision of 29 April 2004 Concerning the Conclusion of the Cooperation Agreement between the European Community and the Islamic Republic of Pakistan, (2004/870/EC) OJ L378/22, Brussels: Council of the European Union.
- Council of the European Union (2004c), 'EU Solidarity Programme on the Consequences of Terrorist Threats and Attacks (Revised/Widened CBRN Programme): Adoption', 1 December, Doc. 15480/04, Brussels: Council of the European Union.

Council of the European Union (2004d), 'Declaration on Combatting Terrorism', 29 March, Doc. 7906/04, Brussels: Council of the European Union.

- Council of the European Union (2004e), 'Presidency Conclusions of the Brussels European Council', 12–13 December, Doc. 5381/04, Brussels: Council of the European Union.
- Council of the European Union (2005a), 'The European Union Counter-Terrorism Strategy', 30 November, Doc. 14469/4/05, Brussels: Council of the European Union.
- Council of the European Union (2005b), 'EU-Afghanistan Joint Declaration: Committing to a New EU-Afghan Partnership', Doc. 14519, Brussels: Council of the European Union.
- Council of the European Union (2005c), 'EU-Iraq Joint Declaration on Political Dialogue', Doc. 12547/05, Brussels: Council of the European Union.
- Council of the European Union (2007), 'Council Joint Action 2007/369/CFSP of 30 May 2007 on Establishment of the European Union Police Mission in Afghanistan' (EUPOL Afghanistan) OJ L139/33, Brussels: Council of the European Union.
- Council of the European Union (2008), 'Revised Strategy on Terrorist Financing', 17 July, Doc. 11778/1/08 REV 1, Brussels: Council of the European Union.
- Council of the European Union (2009a), 'EU/US Agreements on Extradition and on Mutual Legal Assistance', Doc. 14826/09, Brussels: Council of the European Union.
- Council of the European Union (2009b), 'EU-Pakistan Summit Joint Statement', Doc. 11117/09, Brussels: Council of the European Union.
- Council of the European Union (2009c), 'Contribution by the UK Delegation on the Strategy for the External Dimension of Justice and Home Affairs', Doc. 11413/09, Brussels: Council of the European Union.
- Council of the European Union (2009d), 'Council Conclusions on Strengthening Chemical, Biological, Radiological and Nuclear (CBRN) Security in the European Union An EU CBRN Action Plan: Adoption', 12 November, Doc. 15505/1/09 REV 1, Brussels: Council of the European Union.
- Council of the European Union (2010), 'EU–U.S. and Member States 2010 Declaration on Counterterrorism', Brussels: Council of the European Union.
- Council of the European Union (2011a), 'Europol's Role in the Framework of the EU-US TFTP Agreement and State of Play of Operational and Strategic Agreements of Europol (Specific Focus: the Agreement on Exchange of Personal Data and Related Information that Europol has with the US)', Doc. 6266/11, Brussels: Council of the European Union.
- Council of the European Union (2011b), 'EU Action Plan on Combating Terrorism', Doc. 17594/1/11, Brussels: Council of the European Union.
- Council of the European Union (2012a), 'Council Decision 2012/392/CFSP of 16 July 2012 on the European Union CSDP Mission in Niger (EUCAP Sahel Niger)', Official Journal of the EU, 17 July, Doc. 187/48, Brussels: Council of the European Union.
- Council of the European Union (2012b), 'Draft Council Conclusions on the New CBRNE Agenda Adoption', 29 November, Doc. 16980/12, Brussels: Council of the European Union.
- Council of the European Union, HOME-JUST (2013), 'The Stockholm Programme An Open and Secure Europe Serving Protecting the Citizens', available at https://ec.europa.eu/anti-trafficking/eu-policy/stockholm-programme-open-and-secure -europe-serving-and-protecting-citizens-0_en.

- Council of the European Union (2014), 'Report on the Implementation of the EU Counter-Terrorism Strategy', 10 October, Doc. 13971/14, Brussels: Council of the European Union.
- Council of the European Union (2015), 'Outline of the Counter-Terrorism Strategy for Syria and Iraq, with Particular Focus on Foreign Fighters', Doc. 5369/15, Brussels: Council of the European Union.
- Council of the European Union (2016), 'Review of the 2010 EU–US MLA Agreement', Brussels: Council of the European Union.
- Council of the European Union (2017), 'Council Conclusions (16 October 2017)', Doc. 13098/17, Brussels: Council of the European Union.
- Council of the European Union (2019), EU-Pakistan Strategic Engagement Plan', Doc. 7857/19, Brussels: Council of the European Union.
- Council of the European Union (2018a), 'European Council Conclusions on the Salisbury Attack', 22 March, Doc. 163/18, Brussels: Council of the European Union.
- Council of the European Union (2018b), 'Council Conclusions on Iraq', Doc. 5285/18, Brussels: Council of the European Union.
- Council of the European Union (2020), 'Iraq: EU Extends Advisory Mission on Security Sector Reform', 7 April, press release 215/20, Brussels: Council of the European Union.
- Council on Foreign Relations (2006), 'Targets for Terrorism: Nuclear Facilities', Council on Foreign Relations, accessed 5 July 2019 at https://www.cfr.org/backgrounder/targets-terrorism-nuclear-facilities.
- Cox, M. (2005), 'Beyond the West: Terrors in Transatlantia', European Journal of International Relations, 11(2), 203–33.
- Crelinsten, R. (2009), Counterterrorism, Cambridge: Polity Press.
- Cremona, M. (2009), 'EC Competence, "Smart Sanctions", and the *Kadi* Case', *Yearbook of European Law*, 28(1), 559–92.
- Crenshaw, M. (2018), 'Time for Peace Talks with ISIS and Al Qaeda?', Foreign Policy, accessed 4 February 2019 at https://foreignpolicy.com/2018/09/19/time-for -peace-talks-with-isis-and-al-qaeda/.
- Cronin, A. (2009), How Terrorism Ends: Understanding the Decline and Demise of Terrorist Campaigns, Princeton, NJ: Princeton University Press.
- Cronin, A. (2016), 'ISIS is More Than a Terrorist Group', *Foreign Affairs*, accessed 14 December 2018 at https://www.foreignaffairs.com/articles/2016-03-23/isis-more-terrorist-group.
- Customs and Border Protection (2018), 'CSI: Container Security Initiative', accessed 15 January 2019 at https://www.cbp.gov/border-security/ports-entry/cargo-security/csi/csi-brief.
- Danzig, R., Sageman, M., Leighton, T., Hough, L., Yuki, H., Kotani, R. and Hosford, Z. M. (2012), Aum Shinrikyo: Insights into How Terrorists Develop Biological and Chemical Weapons, Washington, DC: Center for a New American Security.
- D'Aspremont, J. and Dopagne, F. (2008), 'Kadi: The ECJ's Reminder of the Elementary Divide between Legal Orders', *International Organizational Law Review*, 5, 371–9.
- Davis Cross, M. (2017), 'Counter-terrorism in the EU's External Relations', *Journal of European Integration*, 39(5), 609–24.
- De Búrca, G. (2009), 'The European Courts and the Security Council: Between Dédoublement Fonctionnel and Balancing of Values: Three Replies to Pasquale De Sena and Maria Chiara Vitucci', *The European Journal of International Law*, 20(3), 853–62.

- De Búrca, G. (2010), 'The European Court of Justice and the International Legal Order after Kadi', *Harvard International Law Journal*, 51(1), 1–49.
- de Goede, M. (2008a), 'The Politics of Preemption and the War on Terror in Europe', European Journal of International Relations, 14(1), 161–85.
- de Goede, M. (2008b), 'Risk, Preemption and Exception in the War on Terrorist Financing', in Amoore, L. and de Goede, M. (eds), *Risk and the War on Terror*, London: Routledge, pp. 97–111.
- de Hert, P. and de Schutter, B. (2008), 'International Transfer of Data in the Field of JHA: The Lessons of Europol, PNR and Swift', in Martenczuck, B. and van Thiel, S. (eds), *Justice, Liberty and Security: New Challenges for EU External Relations*, Brussels: VUB Press/Brussels University Press, 303–340.
- de Kerchove, G. (2009), 'Foreword', *Perspectives on European Politics and Society*, 10(4), 470–73.
- De Londras, F. and Doody, J. (2017), *The Impact, Legitimacy and Effectiveness of EU Counter-terrorism*, London: Routledge.
- De Ruyt, J. (2005), 'A Minister for a European Foreign Policy' ['Un Ministre Pour Une Politique Étrangère Européenne'], available at http://cadmus.eui.eu//handle/1814/7068.
- de Vries, G. (2008), 'The Nexus between EU Crisis Management and Counter-Terrorism', in Blockmans, S. (ed.), *The European Union and Crisis Management*, The Hague: T. M. C. Asser Press, pp. 355–72.
- Deflem, M. (2006), 'Europol and the Policing of International Terrorism: Counter-Terrorism in a Global Perspective', *Justice Quarterly*, 23(3), 336–59.
- Della Cananea, G. (2009), 'Global Security and Procedural Due Process of Law between the United Nations and the European Union: Yassin Abdullah Kadi & Al Barakaat International Foundation v. Council', Columbia Journal of European Law, 15(3), 511–30.
- den Boer, M. (2000), 'The Fight against Terrorism and the Third Pillar of the Maastricht Treaty: Complement or Overlap?', in Reinares, F. (ed.), European Democracies against Terrorism: Governmental Policies and Intergovernmental Cooperation, Dartmouth: Aldershot, 211–226.
- den Boer, M. (2015), 'Counter-Terrorism, Security and Intelligence in the EU: Governance Challenges for Collection, Exchange and Analysis', *Intelligence and National Security*, 30(2–3), 402–19.
- den Boer, M. and Monar, J. (2002), '11 September and the Challenge of Global Terrorism to the EU as a Security Actor', *Journal of Common Market Studies*, 40(1), 11–28
- Denza, E. (2012), 'The Role of the High Representative of the Union for Foreign Affairs and Security Policy', in Denza, E., *The European Union after Lisbon*, Dordrecht: Springer, pp. 481–93.
- Deutsche Welle (2010, 1 August), 'US Accesses European Bank Data Under Controversial SWIFT Agreement', accessed 5 November 2010 at http://www.dw-world.de/dw/article/0,,5855750,00.html.
- Deutsche Welle (2018, 18 March), 'The EU-Turkey Refugee Agreement: A Review', accessed 8 July 2019 at https://www.dw.com/en/the-eu-turkey-refugee-agreement-a-review/a-43028295.
- Dhanapala, J. (2005), 'The United Nations' Response to 9/11', *Terrorism and Political Violence*, 17(1–2), 17–23.
- Dietrich, B. (2019), 'Federica Mogherini', *Diplomatic World*, accessed 18 April 2020 at http://diplomatic-world.com/?p=3586.

- Dijkstra, H. (2011), 'Solana and His Civil Servants: An Overview of Political—Administrative Relations', in Müller-Brabdeck-Bocquet, G. and Rüger, C. (eds), *The High Representative for the EU Foreign and Security Policy: Review and Prospects*, Baden-Baden: NOMOS Verlag, 65–85.
- Douglas-Scott, S. (2004), 'The Rule of Law in the European Union Putting the Security into the Area of Freedom, Security and Justice', *European Law Review*, 29(2), 219–42.
- Draghici, C. (2009), 'International Organisations and Anti-Terrorist Sanctions: No Accountability for Human Rights Violations?', *Critical Studies on Terrorism*, 2(2), 293–312.
- Drieskens, E. (2017), 'Golden or Gilded Jubilee? A Research Agenda for Actorness', Journal of European Public Policy, 24(10), 1534–46.
- Drieskens, E. and Van Schaik, L. (2010), 'The European External Action Service: Preparing for Success', Clingendael, The Hague, available at https://www.clingendael.org/sites/default/files/2016-02/20110200 clingendael paper 1.pdf.
- Dryburgh, L. (2008), 'The EU as a Global Actor? EU Policy towards Iran', *European Security*, 17(2–3), 253–71.
- Duke, S. (2002), 'CESDP and the EU Response to 11 September: Identifying the Weakest Link', European Foreign Affairs Review, 7(2), 153–69.
- Dunmire, P. L. (2009), "9/11 Changed Everything": An Intertextual Analysis of the Bush Doctrine', Discourse & Society, 20(2), 195–222.
- Dunne, T. (2011), '9/11 and the Terrorism Industry', *International Affairs*, 87(4), 965–73.
- Dunne, T. (2017), 'EU Drives International Cooperation on Cybersecurity: Tests Ability to Respond to Threats', available at https://eeas.europa.eu/topics/eu-global-strategy/32160/eu-drives-international-cooperation-cybersecurity-tests-ability-respond-threats en.
- Dunne, T. (2018a), 'Implementing the Global Strategy: EU Delivers on Security and Defence', available at https://eeas.europa.eu/topics/eu-global-strategy/47517/implementing-global-strategy-eu-delivers-security-and-defence_en.
- Dunne, T. (2018b), 'A Global Strategy for the European Union', available at https://eeas.europa.eu/topics/eu-global-strategy/49323/global-strategy-european-union en.
- Dunne, T. (2019), 'Shared View: The Demand for Civilian CSDP Will Remain High', EEAS European Commission, 22 July, available at https://eeas.europa.eu/delegations/libya/65729/shared-view-demand-civilian-csdp-will-remain-high tg.
- Duyvesteyn, I. (2004), 'How New is the New Terrorism?', Studies in Conflict and Terrorism, 27(5), 439–54.
- Eckert, S. and Biersteker, T. (2010), '(Mis)Measuring Success in Countering the Financing of Terrorism', in Andreas, P. and Greenhill, K. M. (eds), *Sex, Drugs, and Body Counts: The Politics of Numbers in Global Crime and Conflict*, Ithaca, NY: Cornell University Press, pp. 247–63.
- Eckes, C. (2009), EU Counter-terrorist Policies and Fundamental Rights, Oxford: Oxford University Press.
- Eder, F. (2011), 'The European Union's Counterterrorism Policy towards the Maghreb: Trapped between Democratisation, Economic Interests and the Fear of Destabilisation', *European Security*, 20(3), 431–51.
- EDRI (2017, 8 February), 'Are the US-EU Data Agreements Still Alive?', accessed 14 January 2019 at https://edri.org/are-the-us-eu-data-agreements-still-alive/.
- Edwards, G. and Meyer, C. (2008), 'Introduction: Charting a Contested Transformation', *Journal of Common Market Studies*, 46(1), 1–25.

- EEAS [European External Action Service] (2011), 'Strategy for Security and Development in the Sahel'.
- EEAS [European External Action Service] (2012), 'EU-Pakistan 5-year Engagement Plan', available at https://eeas.europa.eu/archives/docs/pakistan/docs/2012_feb_eupakistan 5 year engagement plan en.pdf.
- EEAS [European External Action Service] (2016), 'A Global Strategy for the European Union', available at https://eeas.europa.eu/headquarters/headquarters-homepage_en/49323/A%20Global%20Strategy%20for%20the%20European%20Union.
- EEAS [European External Action Service] (2016a), 'Afghanistan and the EU', accessed 20 April 2020 at https://eeas.europa.eu/headquarters/headquarters-homepage/18/afghanistan-and-eu en.
- EEAS [European External Action Service] (2016b), 'EU–Syria Relations, Factsheet', accessed 25 March 2019 at https://eeas.europa.eu/headquarters/headquarters-homepage_en/6769/EU-Syria%20relations,%20factsheet.
- EEAS [European External Action Service] (2017), 'EU and NATO Cooperation to Expand to New Areas, including Counter-terror; Military Mobility; Women, Peace and Security', accessed 20 April 2020 at https://eeas.europa.eu/headquarters/headquarters-homepage/36854/eu-and-nato-cooperation-expand-new-areas-including-counter-terror-military-mobility-women en.
- EEAS [European External Action Service] (2018), 'The EU and Iraq'.
- EEAS [European External Action Service Website] (n.d.), 'Pakistan', accessed 17 November 2011 at http://eeas.europa.eu/pakistan/index en.htm.
- Eeckhout, P. (2007), 'Community Terrorism Listings, Fundamental Rights, and UN Security Council Resolutions: In Search of the Right Fit', *European Constitutional Law Review*, 3(2), 183–206.
- Einsiedel, S. (2016), 'Assessing the UN's Efforts to Counter-terrorism', Tokyo: United Nations University.
- Ekengren, M., Matzen, N., Rhinard, M. and Svantesson, M. (2006), 'Solidarity or Sovereignty? EU Cooperation in Civil Protection', *Journal of European Integration*, 28(5), 457–76.
- El País (2000, 7 June), 'El comisario europeo Patten acusa a Solana de "usurpar" sus tareas en política exterior', accessed 20 April 2020 at https://elpais.com/diario/2000/06/08/internacional/960415205 850215.html.
- Eling, K. (2007), 'The EU, Terrorism and Effective Multilateralism', in Spence, D. (ed.), *The European Union and Terrorism*, London: John Harper, pp. 105–23.
- Engene, J. O. (2007), 'Five Decades of Terrorism in Europe: The TWEED Dataset', *Journal of Peace Research*, 44(1), 109–21.
- English, R. (2009), Terrorism: How to Respond, Oxford: Oxford University Press.
- Errera, P. (2005), 'Three Circles of Threat', Survival, 47(1), 71–88.
- EUAM Iraq (2018), 'Introduction to EUAM Iraq', accessed 9 July 2019 at https://eeas.europa.eu/csdp-missions-operations/euam-iraq/36076/euam-iraq-factsheet en.
- EUObserver (2007), 'EU Anti-Terror Coordinator to Step Down', available at https://euobserver.com/justice/23472.
- EUObserver (2010), 'EU Police Mission in Afghanistan Still Understaffed', accessed 9 March 2022 at https://euobserver.com/justice/29337.
- EUR-Lex (2018), 'Counter-Terrorism Strategy', accessed 16 January 2019 at https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=LEGISSUM%3Al33275.
- Euractiv (2007, 20 September), 'Belgian Appointed New EU Counter-Terrorism Co-Ordinator', accessed 20 April 2020 at https://www.euractiv.com/section/justice -home-affairs/news/belgian-appointed-new-eu-counter-terrorism-co-ordinator/.

- Euractiv (2009, 10 December), 'Solana Leaves Ashton "Impossible Job Description", accessed 20 April 2020 at https://www.euractiv.com/section/future-eu/news/solana -leaves-ashton-impossible-job-description/.
- Eurojust (2019a), 'History of Eurojust', accessed 14 January 2019 at http://www.eurojust.europa.eu/about/background/Pages/history.aspx.
- Eurojust (2019b), 'Third States and Organisations', accessed 14 January 2019 at http://www.eurojust.europa.eu/about/Partners/Pages/third-states.aspx.
- European Commission (1992), 'Treaty of Maastricht on European Union', 29 July, available at https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=LEGISSUM %3Axv0026.
- European Commission (1997), 'Treaty of Amsterdam', available at https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=CELEX%3A11997D%2FTXT.
- European Commission (2002), 'EU Rules on Terrorist Offences and Related Penalties', available at https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=LEGISSUM %3Al33168.
- European Commission (2004a), 'The European Union and Iraq A Framework for Engagement', COM (2004) 417 final.
- European Commission (2004b), 'Proposal for a Council Decision On the Signature on Behalf of the European Community and Provisional Application of Certain Provisions of a Euro-Mediterranean Association Agreement between the European Community and its Member States and the Syrian Arab Republic', COM (2004) 808 final.
- European Commission (2004c), 'Commission Decision 2004/535/EC of 14 May 2004 on the Adequate Protection of Personal Data Contained in the Passenger Name Record of Air Passengers Transferred to the United States Bureau of Customs and Border Protection', OJ L235/11.
- European Commission (2006a), 'Pakistan–European Community Country Strategy Paper for 2007–2013', available at https://eeas.europa.eu/archives/docs/pakistan/csp/07 13 en.pdf.
- European Commission (2006b), 'Recommendations for Renewed European Union Engagement with Iraq'.
- European Commission (2007), 'Treaty of Lisbon', available at https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=CELEX%3A12007L%2FTXT.
- European Commission (2009), *The European Union—United Nations: A Partnership for a Better World*, Brussels: European Commission, External Relations, accessed 10 January 2011 at http://www.europa-eu-un.org/documents/en/2009_EU_UN Partnership for a better world.pdf.
- European Commission (2010a), 'European Instrument for Democracy and Human Rights (EIDHR) Strategy Paper 2011–2013', available at https://by.odb-office.eu/files/eidhr_strategy_paper_2011_2013_com_decision_21_april_2011_text_published_on_internet_en[1].pdf.
- European Commission (2010b), 'Communication from the Commission to the European Parliament and the Council The EU Counter-Terrorism Policy: Main Achievements and Future Challenges', COM (2010) 386.
- European Commission (2010c), 'The EU Internal Security Strategy in Action: Five Steps Towards a More Secure Europe', COM (2010) 673, Brussels: European Commission.
- European Commission (2011a), 'Proposal for a Directive of the European Parliament and of the Council on the Use of Passenger Name Record Data for the Prevention,

- Detection, Investigation, and Prosecution of Terrorist Offences and Serious Crime', COM/2011/32 final.
- European Commission (2011b), 'European Instrument for Democracy and Human Rights Compendium 2007–2010'.
- European Commission (2011c), 'Co-operation between the European Union and Iraq', available at https://eeas.europa.eu/archives/docs/iraq/docs/2011_2013_jsp_nip_en_.pdf.
- European Commission (2011d), 'Proposal for a Decision of the European Parliament and of the Council on Serious Cross-Border Threats to Health', COM (2011) 866, 8 December, Brussels: European Commission.
- European Commission (2011e), 'Joint Strategy Paper, 2011–2013', accessed 9 March 2022 at https://eeas.europa.eu/archives/docs/iraq/docs/2011 2013 jsp nip en.pdf.
- European Commission (2012a), 'Treaty on European Union', available at https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=celex%3A12012M%2FTXT.
- European Commission (2012b), 'Progress Report on the Implementation of the EU CBRN Action Plan (public version)', May, Brussels: European Commission.
- European Commission (2014), 'Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions on a New EU Approach to the Detection and Mitigation of CBRN-E risks', COM (2014) 247, 5 May, Brussels: European Commission.
- European Commission (2015a), 'Instrument Contributing to Stability and Peace', accessed 7 January 2016 at http://ec.europa.eu/dgs/fpi/what-we-do/instrument _contributing_to_stability_and_peace_en.htm.
- European Commission (2015b), 'Joint Proposal for a Council Decision on the Signing, on Behalf of the European Union, of the Cooperation Agreement on Partnership and Development between the European Union and the Islamic Republic of Afghanistan 2015/0303 (NLE)'.
- European Commission (2015c), 'Joint Communication to the European Parliament and the Council: Elements for an EU Regional Strategy for Syria and Iraq as well as the Da-esh Threat'.
- European Commission (2015d), 'The European Agenda on Security', COM (2015) 185, Brussels: European Commission.
- European Commission (2015e), 'Proposal for a Directive of the European Parliament and of the Council on Combating Terrorism and Replacing Council Framework Decision 2002/475/JHA on Combating Terrorism', COM (2015) 625, 2 December, Brussels: European Commission.
- European Commission (2016a), 'Development Cooperation Instrument (DCI)', accessed 7 January 2016 at https://ec.europa.eu/europeaid/how/finance/dci_en.htm_en
- European Commission (2016b), 'European Neighbourhood and Partnership Instrument (ENPI)', accessed 7 January 2016 at http://ec.europa.eu/europeaid/funding/european -neighbourhood-and-partnership-instrument-enpi en.
- European Commission (2016c), 'Instrument Contributing to Stability and Peace, Preventing Conflict Around the World', accessed 28 July 2017 at http://ec.europa.eu/dgs/fpi/what-we-do/instrument_contributing_to_stability_and_peace_en.htm.
- European Commission (2016d), 'The European Union Police Mission in Afghanistan', accessed 11 August 2017 at http://ec.europa.eu/dgs/fpi/showcases/police_mission in afghanistan en.htm.
- European Commission (2017a), 'Report from the Commission to the European Parliament and the Council on the Joint Review of the Implementation of the

- Agreement between the European Union and the United States of America on the Processing and Transfer of Passenger Name Records to the United States Department of Homeland Security'.
- European Commission (2017b), 'Report from the Commission to the European Parliament and the Council on the Joint Review of the Implementation of the Agreement between the European Union and the United States of America on the Processing and Transfer of Financial Messaging Data from the European Union to the United States for the Purposes of the Terrorist Finance Tracking Program'.
- European Commission (2017c), 'Afghanistan', accessed 11 July 2017 at https://ec.europa.eu/europeaid/countries/afghanistan en.
- European Commission (2017d), 'European Instrument for Democracy & Human Rights (EIDHR)', accessed 9 August 2017 at https://ec.europa.eu/europeaid/how/finance/eidhr en.htm en.
- European Commission (2017e), 'Development Cooperation Instrument (DCI)', accessed 11 August 2017 at https://ec.europa.eu/europeaid/how/finance/dci_en.htm en.
- European Commission (2017f), 'EU Steps Up Its Assistance for the Stabilisation of Iraq', accessed 8 August 2017 at https://ec.europa.eu/europeaid/news-and-events/eu-steps-its-assistance-stabilisation-iraq en.
- European Commission (2017g), 'Resilience, Deterrence and Defence: Building Strong Cybersecurity for the EU' (Joint Communication to the European Parliament and the Council-JOIN/2017/0450 final), Brussels: European Commission.
- European Commission (2017h), 'Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions: Action Plan to Enhance Preparedness Against Chemical, Biological, Radiological and Nuclear Security Risks', COM (2017) 610, 18 October, Brussels: European Commission.
- European Commission (2018a), 'Countries and Regions: United States', accessed 10 January 2019 at http://ec.europa.eu/trade/policy/countries-and-regions/countries/united-states/index en.htm.
- European Commission (2018b), 'Pakistan', accessed 18 January 2019 at http://ec.europa.eu/trade/policy/countries-and-regions/countries/pakistan/index_en.htm.
- European Commission (2018c), 'European Neighbourhood Policy and Enlargement Negotiations Syria', accessed 8 July 2019 at https://ec.europa.eu/neighbourhood-enlargement/neighbourhood/countries/syria en.
- European Commission (2018d), 'Security Union: Strengthening Europol's Cooperation with Third Countries to Fight Terrorism and Serious Organized Crime', accessed 20 April 2020 at https://ec.europa.eu/home-affairs/news/security-union-strengthening -europols-cooperation-third-countries-fight-terrorism-and-serious_en.
- European Commission (2018e), 'Communication from the Commission to the European Parliament, the European Council and the Council: Seventeenth Progress Report towards an Effective and Genuine Security Union', COM (2018) 845, 11 December, Strasbourg: European Commission.
- European Commission (2018f), 'rescEU: Commission Welcomes Provisional Agreement to Strengthen EU Civil Protection', press release IP/18/6766, 12 December, Brussels: European Commission.
- European Commission (2019a), 'Passenger Name Record', accessed 15 January 2019 at https://ec.europa.eu/home-affairs/what-we-do/policies/police-cooperation/information-exchange/pnr en.

- European Commission (2019b), 'Terrorist Finance Tracking Programme', accessed 15 January 2019 at https://ec.europa.eu/home-affairs/what-we-do/policies/crisis-and-terrorism/tftp en.
- European Commission (2019c), 'Syria', accessed 4 February 2019 at https://ec.europa.eu/echo/where/middle-east/syria en.
- European Commission (2019d), 'Iraq: EU Announces Additional €50 Million in Humanitarian Aid and Development Cooperation', accessed 8 July 2019 at http://europa.eu/rapid/press-release IP-19-1289 en.htm.
- European Commission (2019e), 'Eurostat: EU Population Up to Over 513 Million on 1 January 2019', accessed 20 April 2020 at https://ec.europa.eu/eurostat/documents/2995521/9967985/3-10072019-BP-EN.pdf/e152399b-cb9e-4a42-a155-c5de6dfe25d1.
- European Commission (n.d.), 'EU in Action: A Partner for Peace'.
- European Commission (n.d.a), 'What is EIDHR?', accessed 30 June 2016 at http://www.eidhr.eu/whatis-eidhr.
- European Commission (n.d.b), 'Instrument for Democracy and Human Rights Worldwide Multiannual Indicative Programme (2014–2017)'.
- European Commission (n.d.c), 'Instrument Contributing to Stability and Peace', accessed 4 August 2017 at https://icsp.insightonconflict.org/.
- European Commission (n.d.d), 'Service for Foreign Policy Instruments IcSP Map', accessed 9 July 2019 at https://icspmap.eu/.
- European Commission (n.d.e), 'EU Regional Trust Fund in Response to the Syrian Crisis', accessed 24 May 2022 at https://ec.europa.eu/trustfund-syria-region/index en.
- European Commission and High Representative of the Union for Foreign Affairs and Security Policy (2016), 'Joint Communication to the European Parliament and the Council Joint Framework on Countering Hybrid Threats: A European Union Response', JOIN (2016) 18, 6 April, Brussels: European Commission and High Representative of the Union for Foreign Affairs and Security Policy.
- European Commission and High Representative of the Union for Foreign Affairs and Security Policy (2018), 'Joint Communication to the European Parliament, the European Council and the Council: Increasing Resilience and Bolstering Capabilities to Address Hybrid Threats', JOIN (2018) 16, 13 June, Brussels: European Commission and High Representative of the Union for Foreign Affairs and Security Policy.
- European Commission External Relations (2008), 'From Warning to Action: Reportage on the EU's Instrument for Stability', Luxembourg: Office for Official Publications of the European Communities.
- European Commission Website (n.d.), 'Multiannual Indicative Programme 2007–2010 Pakistan', available at https://eeas.europa.eu/archives/docs/pakistan/csp/mip_07_10 en.pdf.
- European Council (2001a), Conclusions and Plan of Action of the Extraordinary European Council Meeting on 21 September 2001, Doc. SN 140/01, 21 September, Brussels.
- European Council (2001b), *Anti-Terrorism Roadmap*, Doc. SN 4019/01, 26 September, Brussels.
- European Council (2003a), 'A Secure Europe in a Better World European Security Strategy', 12 December, Brussels.
- European Council (2003b), 'European Security Strategy'.

- European Council (2004a), 'Declaration on Combating Terrorism', 25 March, Brussels: European Council.
- European Council (2004b), 'Summary of Remarks by Javier Solana, EU High Representative for the CFSP, on Terrorism and Intelligence Co-Operation', available at https://www.consilium.europa.eu/ueDocs/cms_Data/docs/pressdata/EN/declarations/80852.pdf.
- European Council (2005), 'The European Union Counter-Terrorism Strategy', 30 November, Brussels.
- European Council (2008), Report on the Implementation of the European Security Strategy: Providing Security in a Changing World, Brussels, 11 December, S407/08.
- European Council (2010), 'European External Action Service', accessed 20 April 2020 at https://eur-lex.europa.eu/legal-content/EN/LSU/?uri=celex:32010D0427.
- European Council and Council of the European Union (2017, 3 April), 'Council Adopts EU Strategy on Syria', accessed 21 May 2019 at https://www.consilium.europa.eu/en/press/press-releases/2017/04/03/fac-conclusions-syria/.
- European Council on Foreign Relations (2019a), 'The Economic War on Syria: Why Europe Risks Losing', accessed 21 May 2019 at https://www.ecfr.eu/article/commentary the economic war on syria why europe risks losing.
- European Council on Foreign Relations (2019b), '60 Days to Save the JCPOA', accessed 12 July 2019 at https://www.ecfr.eu/article/commentary_60_days_to_save the jcpoa iran nuclear deal.
- European Court of Auditors (2015), 'The EU Police Mission in Afghanistan: Mixed Results', available at https://www.eca.europa.eu/Lists/ECADocuments/SR15_07/SR_EUPOL_AFGHANISTAN_EN.pdf.
- European Parliament (2004), Resolution on the Relations between the European Union and the United Nations (2003/2049(INI)), P5-TA (2004) 0037, 29 January, available at https://www.europarl.europa.eu/doceo/document/TA-5-2004-0037 EN.html.
- European Parliament (2006), 'Temporary Committee on the Alleged Use of European Countries by the CIA for the Transport and Illegal Detention of Prisoners', available at https://www.europarl.europa.eu/doceo/document/A-6-2007-0020 EN.html.
- European Parliament (2010), 'Report on a New Strategy for Afghanistan' (2009/2217(INI)) A7-0333/2010, available at https://www.europarl.europa.eu/doceo/document/A-7-2010-0333 EN.html.
- European Parliament (2012), 'Parliament Gives Green Light to Air Passenger Data Deal with the US', accessed 14 January 2019 at http://www.europarl.europa.eu/news/en/press-room/20120419IPR43404/parliament-gives-green-light-to-air-passenger-data-deal-with-the-us.
- European Parliament (2018), 'The Return of Foreign Fighters to EU Soil', accessed 20 April 2020 at https://www.europarl.europa.eu/RegData/etudes/STUD/2018/621811/EPRS_STU(2018)621811_EN.pdf.
- European Parliament and European Council (1995), 'Directive 95/46/EC of the European Parliament and of the Council of 24 October 1995 on the Protection of Individuals with Regard to the Processing of Personal Data and on the Free Movement of Such Data L 281'.
- European Parliament and European Council (2017), 'Directive (EU) 2017/541 on Combating Terrorism and Replacing Council Framework Decision 2002/475/JHA and Amending Council Decision 2005/671/JHA', Official Journal of the European Union L 88/6, accessed 20 June 2018 at https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=CELEX%3A32017L0541.

- European Parliamentary Research Service (2015), 'ISIL/Da'esh and "Non-conventional" Weapons of Terror', Briefing, PE 572.806, Brussels: European Parliament.
- European Union (1998), 'Declaration by the Presidency on Behalf of the European Union on the Relocation of NGO Offices in Kabul, Afghanistan 10583/98'.
- European Union (2001), 'Joint Declaration by the Heads of States and Government of the European Union, the President of the European Parliament, the President of the European Commission, and the High Representative for the Common Foreign and Security Policy', DOC/01/12, 14 September, Brussels: European Union.
- European Union (2006), 'Financing Instrument for Development Cooperation DCI (2007–2013)', accessed 11 August 2017 at http://eur-lex.europa.eu/legal-content/EN/TXT/HTML/?uri=LEGISSUM:114173&from=EN.
- European Union (2007a), 'EU Counter-Terrorism Coordinator'.
- European Union (2007b), 'Three Questions to the New EU Counter-Terrorism Coordinator'.
- European Union (2009), 'EU Blue Book 2009: The EU and Afghanistan', Delegation of the European Commission to the Islamic Republic of Afghanistan.
- European Union (2010a), 'Agreement between the European Union and the United States of America on the Processing and Transfer of Financial Messaging Data from the European Union to the United States for the Purposes of the Terrorist Finance Tracking Program', OJ L195/5.
- European Union (2010b), 'Summary of EU-Pakistan Co-operation'.
- European Union (2012), 'Partnership and Cooperation Agreement between the European Union and its Member States, of the one part, and the Republic of Iraq, of the other part'.
- European Union (2014a), 'Regulation (EU) No 230/2014 of the European Parliament and of the Council of 11 March 2014 Establishing an Instrument Contributing to Stability and Peace'.
- European Union (2014b), 'Regulation (EU) No 233/2014 of the European Parliament and of the Council of 11 March 2014 Establishing a Financing Instrument for Development Co-Operation for the Period 2014–2020'.
- European Union (n.d.), Annex 1 List of EIDHR Non-confidential Projects 2007–2013'. European Union External Action Service (2012), 'Strategy for Security and Development in the Sahel'.
- European Union External Action Service (2017a), 'EU Sets Out New Measures to Counter Terrorism', available at https://eeas.europa.eu/headquarters/headquarters-homepage/34141/eu-sets-out-new-measures-counter-terrorism en.
- European Union External Action Service (2017b), 'EU Advisory Mission in Support of Security Sector Reform in Iraq'.
- Europol (2011), 'TE-SAT 2011: EU Terrorism Situation and Trend Report'.
- Europol (2016), 'European Union Terrorism Situation and Trend Report 2016' (TE-SAT), The Hague: Europol.
- Europol (2017a), 'Europol Staff Numbers', accessed 14 January 2019 at https://www.europol.europa.eu/about-europol/statistics-data.
- Europol (2017b), 'EU Terrorism Situation and Trend Report (TE-SAT) 2017'.
- Europol (2018), 'European Union Terrorism Situation and Trend Report 2018' (TE-SAT), The Hague: Europol.
- Europol (2021), 'European Union Terrorism Situation and Trend Report 2021', Luxembourg: Publications Office of the European Union.

- Europol Joint Supervisory Body (2011), 'Report on the Inspection of Europol's Implementation of the TFTP Agreement', conducted in November 2010 by the Europol Joint Supervisory Body, JSB/Ins. 11-07 (No. JSB/Ins. 11-07).
- Felbab-Brown, V. (2017), 'Afghanistan's Opium Production Is Through the Roof—Why Washington Shouldn't Overreact', Brookings Institution, accessed 18 January 2019 at https://www.brookings.edu/blog/order-from-chaos/2017/11/21/afghanistans-opium-production-is-through-the-roof-why-washington-shouldnt-overreact/.
- Ferreira-Pereira, L. and Oliveira Martins, B. (2012), 'The External Dimension of the European Union's Counter-terrorism: An Introduction to Empirical and Theoretical Developments', *European Security*, 21(4), 459–73.
- Fijnaut, C. (1992), 'Policing Western Europe: Interpol, Trevi and Europol', *Police Studies: International Review of Police Development*, 15(3), 101–106.
- Financial Times (2001, 6 December), 'Common Criminals'.
- Floyd, R. (2016), 'Extraordinary or Ordinary Emergency Measures: What, and Who, Defines the "Success" of Securitisation?', *Cambridge Review of International Affairs*, 29(2), 677–94.
- Flynn, E. J. (2007), 'The Security Council's Counter-terrorism Committee and Human Rights', *Human Rights Law Review*, 7(2), 371–84.
- Foot, R. (2007), 'The United Nations, Counter Terrorism, and Human Rights: Institutional Adaptation and Embedded Ideas', *Human Rights Quarterly*, 29(2), 489–514.
- Foster, S. (2018), 'Sarajevo Incident', in 1914–1918 Online: International Encyclopedia of the First World War, Freie Universität Berlin, available at https://ueaeprints.uea .ac.uk/72080/.
- Freedman, Lawrence (2008), A Choice of Enemies: America Confronts the Middle East, New York: Public Affairs.
- Fuchs-Drapier, M. (2011), 'The European Union's Solidarity Clause in the Event of a Terrorist Attack: Towards Solidarity or Maintaining Sovereignty?', *Journal of Contingencies and Crisis Management*, 19(4), 184–97.
- Fuster, G. G., de Hert, P. and Gutwirth, S. (2008), 'SWIFT and the Vulnerability of Transatlantic Data Transfers', *International Review of Law, Computers, and Technology*, 22(1–2), 191–202.
- Gale, R. P. and Armitage, J. O. (2018), 'Are We Prepared for Nuclear Terrorism?', The New England Journal of Medicine, 378, 1246–54.
- Gannon, K. (2004), 'Afghanistan Unbound', Foreign Affairs, 83(3), 35–46.
- Garrison, A. H. (2004), 'Defining Terrorism: Philosophy of the Bomb, Propaganda by Deed and Change through Fear and Violence', *Criminal Justice Studies*, 17(3), 259–79.
- Gaub, F. and Pauwels, A. (2017), 'Counter-terrorism Cooperation with the Southern Neighborhood', European Union Institute for Security Studies, accessed 20 April 2020 at https://www.iss.europa.eu/content/counter-terrorism-cooperation-southern-neighbourhood.
- Gehring, T., Urbanski, K. and Oberthur, S. (2017), 'The European Union as an Inadvertent Great Power: EU Actorness and the Ukraine Crisis', *Journal of Common Market Studies*, 55(4), 727–43.
- General Assembly (2018), 'Activities of the United Nations System in Implementing the United Nations Global Counter-Terrorism Strategy', United Nations.
- George Washington University (2018), 'Europol Executive Director Rob Wainwright', accessed 15 January 2019 at https://cchs.gwu.edu/europol-executive-director-rob-wainwright.

- Gerstl, A. (2010), 'The Depoliticisation and "ASEANisation" of Counter-Terrorism Policies in Southeast Asia: A Weak Trigger for a Fragmented Version of Human Security', *ASEAS Austrian Journal of South-East Asian Studies*, 3(1), 48–75.
- Ginsberg, R. (1999), 'Conceptualizing the European Union as an International Actor: Narrowing the Theoretical Capability–Expectations Gap', *Journal of Common Market Studies*, 37(3), 429–54.
- Ginsberg, R. (2001), *The European Union in International Politics Baptism by Fire*, New York: Rowman and Littlefield.
- Global Terrorism Database (2018), 'National Consortium for the Study of Terrorism and Responses to Terrorism (START)', accessed 14 December 2018 at https://www.start.umd.edu/gtd/terms-of-use/CitingGTD.aspx.
- Global Terrorism Index (2019), accessed 17 March 2020 at http://visionofhumanity.org/app/uploads/2019/11/GTI-2019web.pdf.
- Goldsmith, J. and Posner, E. (2008, 25 November), 'Does Europe Believe in International Law?', *The Wall Street Journal*, accessed 14 January 2011 at http://online.wsj.com/article/SB122757164701554711.html.
- Gonçalves, D. (2016), 9/11: Culture, Catastrophe and the Critique of Singularity, Berlin: Walter De Gruyter.
- Gowan, R. (2008), 'The European Security Strategy's Global Objective: Effective Multilateralism', in Biscop, S. and Andersson, J. J. (eds), *The EU and the European Security Strategy*, London: Routledge, pp. 42–61.
- Gregory, S. (2003), 'France and the War on Terrorism', *Terrorism and Political Violence*, 15(1), 124–47.
- Grieco, J. (1997), 'Realist International Theory and the Study of World Politics', in Doyle, M. and Ikenberry, G. (eds), *New Thinking in International Relations Theory*, Boulder, CO: Westview Press, 163–201.
- Griller, S. and Ziller, J. (2008), *The Lisbon Treaty: EU Constitutionalism without a Constitutional Treaty?*, Dordrecht: Springer.
- Groenleer, M. and van Schaik, L. (2007), 'United We Stand? The European Union's International Actorness in the Cases of the International Criminal Court and the Kyoto Protocol', *Journal of Common Market Studies*, 49(5), 969–98.
- Gruszczak, A. (2016a), Intelligence Security in the European Union: Building a Strategic Intelligence Community, Dordrecht: Springer.
- Gruszczak, A. (2016b), 'Police and Customs Cooperation Centres and Their Role in EU Internal Security Governance', in Bossong, R. and Carrapico, H. (eds), *EU Borders and Shifting Internal Security*, Dordrecht: Springer, pp. 157–75.
- Guardian (2003, 24 January), 'Anger at Rumsfeld Attack on "Old Europe", accessed 24 March 2019 at https://www.theguardian.com/world/2003/jan/24/germany.france.
- Guardian (2006, 7 September), 'We are No Longer All Americans', accessed 21 December 2010 at http://www.guardian.co.uk/commentisfree/2006/sep/07/yamani.
- Guardian, (2009, 8 February), 'Taliban Hostage-takers Behead Polish Engineer', accessed 20 April 2020 at https://www.theguardian.com/world/2009/feb/08/taliban-pakistan-stancza.
- Guardian (2011, 10 July), 'US Suspends Pakistan Military Aid as Diplomatic Relations Worsen', accessed 20 April 2020 at https://www.theguardian.com/world/2011/jul/ 10/us-suspends-pakistan-military-aid.
- Guardian (2016, 25 March), 'Brexit Would Increase Terrorist Threat to UK, Says Ex-minister', accessed 29 June 2017 at https://www.theguardian.com/politics/2016/mar/25/britain-not-safer-terrorists-eu-pauline-neville-jones.

- Guardian (2017a, 13 November), 'Two Years on from Paris Attacks, Suspect Salah Abdeslam Remains Silent', accessed 13 November 2017 at https://www.theguardian.com/world/2017/nov/13/two-years-on-from-paris-attacks-suspect-salah-abdeslam-remains-silent.
- Guardian (2017b, 18 September), 'UK Calls for Treaty on Post-Brexit Security Cooperation with EU', accessed 13 November 2017 at https://www.theguardian.com/politics/2017/sep/18/uk-calls-for-treaty-on-post-brexit-security-cooperation-with-eu.
- Guardian (2018, 13 December), 'Strasbourg Attack Suspect Killed by Police', accessed 14 December 2018 at https://www.theguardian.com/world/2018/dec/13/strasbourg-attack-suspect-killed-by-police-reports-say.
- Guild, E. (2008), 'The Uses and Abuses of Counter-Terrorism Policies in Europe: The Case of the Terrorist Lists', *Journal of Common Market Studies*, 46(1), 173–93.
- Guild, E. (2010), EU Counter-Terrorism Action: A Fault Line between Law and Politics, Brussels: Centre for European Policy Studies.
- Guild, E. and Brouwer, E. (2006), 'The Political Life of Data: The ECJ Decision on the PNR Agreement between the EU and the US', accessed 21 December 2010 at http://www.ceps.eu/system/files/book/1363.pdf.
- Gunaranta, R. (2018), 'ASEAN's Greatest Counter-Terrorism Challenge: The Shift from "Need to Know" to Smart to Share', in Echle, C., Gunaratna, R., Rueppel, P. and Sarmah, M. (eds), *Combatting Violent Extremism and Terrorism in Asia and Europe: From Cooperation to Collaboration*, Konrad-Adenauer-Stiftung and S. Rajaratnam School of International Studies, pp. 111–28.
- Gunaratna, R. and Nielsen, A. (2008), 'Al Qaeda in the Tribal Areas of Pakistan and Beyond', *Studies in Conflict and Terrorism*, 31(9), 775–807.
- Haacke, J. and Williams, P. D. (2008), 'Regional Arrangements, Securitization, and Transnational Security Challenges: The African Union and the Association of Southeast Asian Nations Compared', Security Studies, 17(4), 775–809.
- Halberstam, D. and Stein, E. (2009), 'The United Nations, the European Union, and the King of Sweden: Economic Sanctions and Individual Rights in a Plural World Order', *Common Market Law Review*, 46(1), 13–72.
- Hamilton, K., Colonnese, F. and Dunaiski, M. (2016), Children and Counter Terrorism, United Nations Interregional Crime and Justice Research Institute-UNICRI.
- Hansard (2016), 'Parliamentary Scrutiny of Leaving the EU, Vol. 615', accessed 26 June 2017 at https://hansard.parliament.uk/Commons/2016-10-12/debates/ F327EC64-3777-4D40-A98D-BEC2E11763A2/ParliamentaryScrutinyOfLe avingTheEU#contribution-4581E073-B39B-400F-A1ED-5098EA703726.
- Hardy, K. (2019), 'Countering Right-Wing Extremism: Lessons from Germany and Norway', *Journal of Policing, Intelligence, and Counter-Terrorism*, 14(3), 262–79.
- Hassan, O. (2021), 'Reassessing the European Strategy in Afghanistan', Brussels: Carnegie Europe.
- Hayes, B. and Jones, C. (2015), 'Taking Stock: The Evolution, Adoption, Implementation and Evaluation of EU Counter-Terrorism Policy', in de-Landras, F. and Doody, J. (eds), *The Impact, Legitimacy and Effectiveness of EU Counter-Terrorism*, London: Routledge, pp. 13–39.
- Heiberg, M. (2007), 'ETA: Euskadi 'ta Askatasuna', in Heiberg, M. et al. (eds), *Terror, Insurgency, and the State: Ending Protracted Conflicts*, Philadelphia: University of Pennsylvania Press, pp. 19–49.

- Helwig, N. (2015), 'The High Representative of the Union: The Quest for Leadership in EU Foreign Policy', in Spence, D. and Batora, J. (eds), *The European External Action Service*, Dordrecht: Springer, pp. 87–104.
- Helwig, N. (2018), New Tasks for EU–NATO Cooperation: An Inclusive EU Defence Policy Requires Close Collaboration with NATO, SWP-Berlin.
- Helwig, N. and Rüger, C. (2014), 'In Search of a Role for the High Representative: The Legacy of Catherine Ashton', *The International Spectator*, 49(4), 1–17.
- Heupel, M. (2009), 'Multilateral Sanctions against Terror Suspects and the Violation of Due Process Standards', *International Affairs*, 85(2), 307–21.
- Hill, C. (2004), 'Renationalizing or Regrouping? EU Foreign Policy since 11 September 2001', *Journal of Common Market Studies*, 42(1), 143–63.
- Hinnebusch, R. (2007), 'The US Invasion of Iraq: Explanations and Implications', *Critique: Critical Middle Eastern Studies*, 16(3), 209–28.
- Hinojosa Martínez, L. M. (2008), 'Bad Law for Good Reasons: The Contradictions of the Kadi Judgment', *International Organizations Law Review*, 5, 339–57.
- HM Government (2011), 'CONTEST: The United Kingdom's Strategy for Countering Terrorism'.
- HM Government (2016), 'The UK's Cooperation with the EU on Justice and Home Affairs, and on Foreign Policy and Security Issues', available at https://www.gov.uk/government/publications/the-uks-cooperation-with-the-eu-on-justice-home-affairs-foreign-policy-and-security-issues-background-note.
- HM Government (2017a), 'The United Kingdom's Exit from and New Partnership with the European Union', available at https://www.gov.uk/government/publications/the -united-kingdoms-exit-from-and-new-partnership-with-the-european-union-white -paper.
- HM Government (2017b), 'Foreign Policy, Defence and Development: A Future Partnership Paper', available at https://www.gov.uk/government/publications/foreign-policy-defence-and-development-a-future-partnership-paper.
- HM Government (2017c), 'Security, Law Enforcement And Criminal Justice', available at https://www.gov.uk/government/publications/security-law-enforcement-and-criminal-justice-a-future-partnership-paper.
- Hoffman, B. (1982), 'Right-Wing Terrorism in Europe', RAND, accessed 18 March 2020 at https://www.rand.org/pubs/notes/N1856.html.
- Hoffman, B. (1999), 'Is Europe Soft on Terrorism?', Foreign Policy, 115, 62–77.
- Hoffman, B. (2009), 'The First Non-state Use of a Chemical Weapon in Warfare: The Tamil Tigers' Assault on East Kiran', *Small Wars & Insurgencies*, 20(3–4), 463–77.
- Honig, O. and Yahel, I. (2017), 'A Fifth Wave of Terrorism? The Emergence of Terrorist Semi-States', Terrorism and Political Violence, 31(6), 1210–28.
- Hopkins, N. (2012, 25 June), 'MI5 Warns Al-Qaida Regaining UK Toehold after Arab Spring', *The Guardian*, accessed 20 April 2020 at https://www.theguardian.com/uk/2012/jun/25/mi5-uk-terrorism-threat-warning.
- Hopkins, N. and Wintour, P. (2013, 20 December), 'Latest Snowden Revelations Spark Anger at European Commission', *The Guardian*, available at https://www .theguardian.com/world/2013/dec/20/snowden-revelations-almunia-anger-european -commission.
- House of Lords (2011), 'European Union Committee Seventeenth Report: The EU Internal Security Strategy', available at https://publications.parliament.uk/pa/ld201012/ldselect/ldeucom/149/14902.htm.
- House of Lords (2016), 'Corrected Oral Evidence: Brexit: Future UK-EU Security and Policing Co-operation', accessed 28 June 2017 at http://data.parliament.uk/

- writtenevidence/committeeevidence.svc/evidencedocument/eu-home-affairs -subcommittee/brexit-future-ukeu-security-and-policing-cooperation/oral/41072 .html.
- Howlett, M., McConnell, A. and Perl, A. (2015), 'Streams and Stages: Reconciling Kingdon and Policy Process Theory', European Journal of Political Research, 54(3), 419–34.
- Howorth, J. (2010), 'CESDP After 11 September: From Short-Term Confusion to Long-Term Cohesion?', available at https://www.researchgate.net/publication/29991078_CESDP_After_11_September_From_Short-term_Confusion_to_Long_-Term Cohesion.
- Howorth, J. (2011), 'The "New Faces" of Lisbon: Assessing the Performance of Catherine Ashton and Herman van Rompuy on the Global Stage', *European Foreign Affairs Review*, 16(3), 303–23.
- Hughes, C. W. (1998), 'Japan's Aum Shinrikyo, the Changing Nature of Terrorism, and the Post-Cold War Security Agenda', Global Change, Peace & Security, 10(1), 39–60.
- Ibrahimi, S. Y. (2017), 'The Taliban's Islamic Emirate of Afghanistan (1996–2001): "War-Making and State-Making" as an Insurgency Strategy', Small Wars & Insurgencies, 28(6), 947–72.
- Inkster, N. (2016), 'Brexit, Intelligence, and Terrorism', Survival, 58(3), 23–30.
- Institute for Economics and Peace, Global Terrorism Index (2019), accessed 17 March 2020 at http://visionofhumanity.org/app/uploads/2019/11/GTI-2019web.pdf.
- International Centre for Counter-Terrorism (2016), 'The Foreign Fighters Phenomenon in the European Union', available at https://icct.nl/app/uploads/2016/03/ICCT -Report_Foreign-Fighters-Phenomenon-in-the-EU_1-April-2016_including -AnnexesLinks.pdf.
- International Centre for the Study of Radicalisation (2018), 'From Daesh to 'Diaspora': Tracing the Women and Minors of Islamic State', available at https://icsr.info/wp-content/uploads/2018/07/ICSR-Report-From-Daesh-to-%E2%80%98Diaspora %E2%80%99-Tracing-the-Women-and-Minors-of-Islamic-State.pdf.
- Ioannides, I. (2014), 'Inside-out and Outside-in: EU Security in the Neighbourhood', The International Spectator, 49(1), 113–32.
- Iraq Body Count (2019), 'Documented Civilian Deaths from Violence', accessed 24 March 2019 at https://www.iraqbodycount.org/database/.
- Irish Independent (2017), 'Brexit Calamitous Act of Self-Harm, Warns Patten', accessed 12 November 2017 at https://www.independent.ie/business/brexit/brexit -calamitous-act-of-selfharm-warns-patten-36208445.html.
- Irish Times (2003, 23 September), 'EU Security Strategy Aims to Build a Safer, Fairer, More United World', accessed 20 April 2020 at https://www.irishtimes .com/opinion/eu-security-strategy-aims-to-build-a-safer-fairer-more-united-world-1 .377974.
- ISAF (2010), 'International Security Assistance Force: Key Facts and Figures'.
- Islam, S. (2013), 'EU-Pakistan Relations: The Challenge of Dealing with a Fragile State', in Christiansen, T., Kirchner, E. and Murray, P. (eds), *The Palgrave Handbook of EU-Asia Relations*, Basingstoke, Palgrave Macmillan, 587–602.
- Jackson, R. (2007), 'An Analysis of EU Counterterrorism Discourse Post-September 11', Cambridge Review of International Affairs, 20(2), 233–47.
- Joffe, G. (2008), 'The European Union, Democracy and Counter-Terrorism in the Maghreb', *Journal of Common Market Studies*, 46(1), 147–71.

- Jupille, J. and Caporaso, J. (1998), 'States, Agency, and Rules: The European Union in Global Environmental Politics', in Rhodes, C. (ed.), *The European Union in the World Community*, London: Lynne Rienner, 213–229.
- Kaddous, C. (2008), 'Role and Position of the High Representative of the Union for Foreign Affairs and Security Policy under the Lisbon Treaty', in Griller, S. and Ziller, J. (eds), *The Lisbon Treaty: EU Constitutionalism without a Constitutional Treaty?*, Schriftenreihe Der Österreichischen Gesellschaft Für Europaforschung (ECSA Austria)/European Community Studies Association of Austria Publication Series, Vienna: Springer Vienna, pp. 205–21.
- Kagan, R. (2003), Of Paradise and Power: America vs. Europe in the New World Order, New York: Knopf.
- Kaplan, J. (1997), 'Leaderless Resistance', Terrorism and Political Violence, 9(3), 80–95.
- Kaplan, J. (2007), 'The Fifth Wave: The New Tribalism?', Terrorism and Political Violence, 19(4), 545–70.
- Kaplan, J. and Costa, C. (2015), 'The Islamic State and the New Tribalism', *Terrorism and Political Violence*, 27(5), 926–69.
- Karlsrud, J. (2017), 'Towards UN Counter-Terrorism Operations?', *Third World Quarterly*, 38(6), 1215–31.
- Kasapoglu, C. (2018), 'A Guide for EU–NATO Security Cooperation on Foreign Terrorist Fighters in the Euro-Mediterranean Region', *Konrad Adenauer Stiftung*, accessed 20 April 2020 at https://www.kas.de/single-title/-/content/foreign-terrorist -fighters-in-the-euro-mediterranean-area1.
- Kaunert, C. (2005), 'The Area of Freedom, Security and Justice: The Construction of a "European Public Order", *European Security*, 14(4), 459–83.
- Kaunert, C. (2007), 'Without the Power of Purse or Sword: The European Arrest Warrant and the Role of the Commission', *Journal of European Integration*, 29(4), 387–404.
- Kaunert, C. (2010a), 'The External Dimension of EU Counter-Terrorism Relations: Competences, Interests, and Institutions', *Terrorism and Political Violence*, 22(1), 41–61.
- Kaunert, C. (2010b), 'Europol and EU Counterterrorism: International Security Actorness in the External Dimension', Studies in Conflict and Terrorism, 33(7), 652–71.
- Kaunert, C. (2010c), European Internal Security: Towards Supranational Governance in the Area of Freedom, Security and Justice?, Manchester: Manchester University Proces
- Kaunert, C. (2010d), 'The Area of Freedom, Security and Justice in the Lisbon Treaty: Commission Policy Entrepreneurship?', *European Security*, 19(2), 169–89.
- Kaunert, C. (2012), 'Conclusion: Assessing the External Dimension of EU Counter-Terrorism Ten Years On', *European Security*, 21(4), 578–87.
- Kaunert, C., Callander, B. and Léonard, S. (2021), 'The Collective Securitization of Aviation in the European Union through Association with Terrorism', *Global Affairs*, available at https://doi.org/10.1080/23340460.2021.2002099.
- Kaunert, C. and Della Giovanna, M. (2010), 'Post 9/11 EU Counter-terrorist Financing Cooperation: Differentiating Supranational Policy Entrepreneurship by the Commission and the Council Secretariat', *European Security*, 19(2), 275–95.
- Kaunert, C. and Léonard, S. (2011a), 'EU Counterterrorism and the European Neighbourhood Policy: An Appraisal of the Southern Dimension', *Terrorism and Political Violence*, 23(2), 286–309.

- Kaunert, C. and Léonard, S. (2011b), 'Introduction: European Security Governance after the Lisbon Treaty: Neighbours and New Actors in a Changing Security Environment', *Perspectives on European Politics and Society*, 12(4), 361–70.
- Kaunert, C. and Léonard, S. (2012), 'Introduction: Supranational Governance and European Union Security after the Lisbon Treaty – Exogenous Shocks, Policy Entrepreneurs and 11 September 2001', Cooperation and Conflict, 47(4), 417–32.
- Kaunert, C. and Léonard, S. (2013), European Security, Terrorism and Intelligence: Tackling New Security Challenges in Europe, Dordrecht: Springer.
- Kaunert, C. and Léonard, S. (2019a), 'The Collective Securitization of Terrorism in the EU', West European Politics, 41(2), 261–77.
- Kaunert, C. and Léonard, S. (2019b), 'The European Union's Response to the CBRN Terrorist Threat: A Multiple Streams Approach', *Politique Européenne*, 65(3), 148–77.
- Kaunert, C., Léonard, S. and MacKenzie, A. (2012), 'The Social Construction of an EU Interest in Counter-terrorism: US Influence and Internal Struggles in the Cases of PNR and SWIFT', European Security, 21(4), 474–96.
- Kaunert, C., Léonard, S. and MacKenzie, A. (2015), 'The European Parliament in the External Dimension of EU Counter-terrorism: More Actorness, Accountability and Oversight 10 Years on?', *Intelligence and National Security*, 30(2–3), 357–76.
- Kaunert, C., Léonard, S. and Wertman, O. (2019), 'EU Counter-terrorism Cooperation with the Middle East and North Africa', in Cusumano, E. and Hofmaier, S. (eds), Projecting Resilience across the Mediterranean, Basingstoke: Palgrave Macmillan, pp. 87–102.
- Kaunert, C. and Wertman, O. (2019), 'Counter-Terrorism Cooperation between NATO and the EU', in Lindstrom, G. and Tardy, T. (eds), *The EU and NATO: The Essential Partners*, European Union Institute for Security Studies EUISS, pp. 63–71.
- Kellner, D. (2004), '9/11, Spectacles of Terror, and Media Manipulation: A Critique of Jihadist and Bush Media Politics', *Critical Discourse Studies*, 1(1), 41–64.
- Keohane, D. (2005), *The EU and Counter-Terrorism*, London: Centre for European Reform.
- Keohane, D. (2008), 'The Absent Friend: EU Foreign Policy and Counter-Terrorism', Journal of Common Market Studies, 46(1), 125–46.
- Keukeleire, S. and Delreux, T. (2014), *The Foreign Policy of the European Union*, Basingstoke: Palgrave Macmillan.
- Kfir, I. (2008), 'Islamic Radicalism in East Africa: Is There a Cause for Concern?', Studies in Conflict & Terrorism, 31(9), 829–55.
- Kienzle, B. (2013), 'The Role of Ideas in EU Responses to International Crises: Comparing the Cases of Iraq and Iran', *Cooperation and Conflict*, 48(3), 424–43.
- Kingdon, J. (1984), Agendas, Alternatives, and Public Policies, Boston, MA: Little, Brown & Co.
- Kissack, R. (2010), Pursuing Effective Multilateralism: The European Union, International Organisations and the Politics of Decision Making, Basingstoke: Palgrave Macmillan.
- Klaiber, K. (2007), 'The European Union in Afghanistan: Lessons Learned', *European Foreign Affairs Review*, 12(1), 7–11.
- Koehler, D. (2014), 'German Right-Wing Terrorism in Historical Perspective: A First Quantitative Overview on the "Database on Terrorism in Germany (Right-Wing Extremism DTG Project", *Perspective on Terrorism*, 8(5), 48–58.
- Koehler, D. (2019), Violence and Terrorism from the Far-Right: Policy Options to Counter an Elusive Threat, International Centre for Counter-Terrorism, accessed 24

- February 2020 at https://icct.nl/wp-content/uploads/2019/02/Koehler-Violence-and -Terrorism-from-the-Far-Right-February-2019.pdf.
- Krause, K. and Williams, M. C. (1996), 'Broadening the Agenda of Security Studies? Politics and Methods', *Mershon International Studies Review*, 40(2), 229–54.
- Kunoy, B. and Dawes, A. (2009), 'Plate Tectonics in Luxembourg: The Ménage à Trois Between EC Law, International Law and the European Convention on Human Rights Following the UN Sanctions Cases', Common Market Law Review, 46(1), 73–104.
- Laqueur, W. (1999), The New Terrorism: Fanaticism and the Arms of Mass Destruction, Oxford: Oxford University Press.
- Laqueur, W. (2017), A History of Terrorism, London: Routledge.
- Larive, M. (2012), 'From Speeches to Actions: EU Involvement in the War in Afghanistan through the EUPOL Afghanistan Mission', European Security, 21(2), 185–201.
- Larsen, H. (2005), Analysing the Foreign Policy of Small States in the EU, London: Palgrave Macmillan.
- Lavranos, N. (2006), 'Judicial Review of UN Sanctions by the Court of First Instance', European Foreign Affairs Review, 11(4), 471–90.
- Lebl, L. S. (2006), 'Working with the European Union', *Orbis*, 50(1), 117–32.
- Léonard, S. (2010), 'The Use and Effectiveness of Migration Controls as a Counterterrorism Instrument in the European Union', *Central European Journal of International and Security Studies*, 4(1), 32–50.
- Léonard, S. (2015), 'Border Controls as a Dimension of the European Union's Counter-terrorism Policy: A Critical Assessment', *Intelligence and National Security*, 30(2–3), 306–32.
- Léonard, S. and Kaunert, C. (2011), 'Reconceptualising the Audience in Securitization Theory', in Balzacq, T. (ed.), *Securitization Theory: How Security Problems Emerge and Dissolve*, London: Routledge, pp. 57–76.
- Léonard, S. and Kaunert, C. (2021a), 'The Development of Europol's External Relations: Towards Supranationalism?', *Maastricht Journal of European and Comparative Law*, 28(2), 229–44.
- Léonard, S. and Kaunert, C. (2021b), 'De-centring the Securitisation of Asylum and Migration in the European Union: Securitisation, Vulnerability and the Role of Turkey', *Geopolitics*, available at https://doi.org/10.1080/14650045.2021.1929183.
- Léonard, S. and Kaunert, C. (2021c), 'Refugee Flows and Terrorism in the European Union: Securitization through Association', *International Politics*, available at https://doi.org/10.1057/s41311-021-00359-4.
- Lustick, I. S. (2006), Trapped in the War on Terror, Philadelphia: University of Pennsylvania Press.
- MacKenzie, A. (2010), 'The European Union's Increasing Role in Foreign Policy Counterterrorism', *Journal of Contemporary European Research*, 6(2), 148–64.
- MacKenzie, A. (2012), 'The External Dimension of European Homeland Security', in Kaunert, C., Léonard, S. and Pawlak, P. (eds), European Homeland Security A European Strategy in the Making?, London: Routledge, pp. 95–110.
- MacKenzie, A., Bures, O., Kaunert, C. and Léonard, S. (2013a), 'The European Union Counter-Terrorism Coordinator and the External Dimension of the European Union Counter-Terrorism Policy', *Perspectives on European Politics and Society*, 14(3), 325–38.

- MacKenzie, A., Kaunert, C. and Léonard, S. (2013b), 'EU Counterterrorism and the Southern Mediterranean Countries after the Arab Spring: New Potential for Cooperation?', *Democracy and Security*, 9(1–2), 137–56.
- Mahncke, D. and Monar, J. (eds) (2006), *International Terrorism: A European Response to a Global Threat?*, Brussels: P.I.E. Peter Lang.
- Manthe, B. (2018), 'On the Pathway to Violence: West German Right-Wing Terrorism in the 1970s', *Terrorism and Political Violence*, 33(1), 49–70.
- Marcus, J. (2019, 7 July), 'Iran Nuclear Deal: Is There Any Way It Can Be Saved?', BBC, accessed 12 July 2019 at https://www.bbc.co.uk/news/world-middle-east -48900432.
- Martin, M. and Solomon, H. (2017), 'Islamic State: Understanding the Nature of the Beast and its Funding', *Contemporary Review of the Middle East*, 4(1), 18–49.
- Martins, B. O. and Ziegler, M. (2018), 'Counter-Radicalization as Counter-Terrorism: The European Union Case', in Steiner, K. and Önnerfors, A. (eds), *Expressions of Radicalization: Global Politics, Processes and Practices*, Basingstoke: Palgrave Macmillan, pp. 321–52.
- McAlexander, R. (2020), 'How are Immigration and Terrorism Related? An Analysis of Right- and Left-Wing Terrorism in Western Europe 1980–2004', *Journal of Global Security Studies*, 5(1), 179–95.
- McLeish, C. (2017), 'Recasting the Threat of Chemical Terrorism in the EU: The Issue of Returnees from the Syrian Conflict', *European Journal of Risk Regulation*, 8(4), 643–57.
- McLeish, C. and Revill, J. (2018), 'Novi-whats and the Chemical Weapons Convention', *CBRNe World*, April, 16–18.
- Mearsheimer, J. (1990), 'Why We Will Soon Miss the Cold War', *The Atlantic Monthly*, 266, 35-50.
- Mendelsohn, B. (2009), Combating Jihadism: American Hegemony and Interstate Cooperation in the War on Terrorism, Chicago, IL: University of Chicago Press.
- Merelli, A. (2015, 25 November), 'Charted: Terror Attacks in Western Europe from the 1970s to Now', *Quartz*, accessed 20 June 2018 at https://qz.com/558597/charted-terror-attacks-in-western-europe-from-the-1970s-to-now/.
- Merskin, D. (2005), 'Making Enemies in George W. Bush's Post-9/11 Speeches', Peace Review: A Journal of Social Justice, 17(4), 373–81.
- Mesterhazy, A. (2017), 'NATO–EU Cooperation after Warsaw (NATO Parliamentary Assembly, Defence and Security Committee)', accessed 20 April 2020 at https:// www.nato-pa.int/document/2017-eu-and-nato-cooperation-mesterhazy-report-163 -dsctc-17-e-rev1-fin.
- Meulenbelt, S. E. and Nieuwenhuizen, M. S. (2015), 'Non-state Actors' Pursuit of CBRN Weapons: From Motivation to Potential Humanitarian Consequences', International Review of the Red Cross, 97(899), 831–58.
- Michael, M. (2012), 'Leaderless Resistance: The New Face of Terrorism', *Defence Studies*, 12(2), 257–82.
- Mitsilegas, V. and Gilmore, B. (2007), 'The EU Legislative Framework Against Money Laundering and Terrorist Finance: A Critical Analysis in the Light of Evolving Global Standards', *International and Comparative Law Quarterly*, 56(1), 119–40.
- Mogherini, F. (2015a), 'Remarks by High Representative/Vice-President Federica Mogherini at the Press Conference after the Foreign Affairs Council', available at https://www.avrupa.info.tr/en/news/remarks-high-representativevice-president-federica-mogherini-press-conference-after-foreign.

- Mogherini, F. (2015b), 'Speech of the HR/VP Federica Mogherini The EU Internal—External Security Nexus: Terrorism as an Example of the Necessary Link between Different Dimensions of Action'.
- Mogherini, F. (2017), 'Speech by the High Representative/Vice-President Federica Mogherini at the Conference "Hybrid Threats and the EU: State of Play and Future Progress", accessed 20 April 2020 at https://eeas.europa.eu/topics/eu-global-strategy/33141/speech-high-representative-vice-president-federica-mogherini-conference-hybrid-threats-and-eu en.
- Monaco, L. (2017), 'Preventing the Next Attack: A Strategy for the War on Terrorism', Foreign Affairs, accessed 20 April 2020 at https://www.foreignaffairs.com/articles/ world/2017-10-16/preventing-next-attack.
- Monar, J. (2005), 'A New Area of Freedom, Security and Justice for the Enlarged EU? The Results of the European Convention', in Henderson, K. (ed.), *The Area of Freedom, Security and Justice in the Enlarged Europe*, Basingstoke: Palgrave, 110–134.
- Monar, J. (2007a), 'Common Threat and Common Response? The European Union's Counter-Terrorism Strategy and its Problems', *Government and Opposition*, 42(3), 292–313.
- Monar, J. (2007b), 'The EU's Approach Post-September 11: Global Terrorism as a Multidimensional Law Enforcement Challenge', Cambridge Review of International Affairs, 20(2), 267–83.
- Monar, J. (2010), 'The Rejection of the EU–US SWIFT Interim Agreement by the European Parliament: A Historic View and its Implications', *European Foreign Affairs Review*, 15(2), 143–51.
- Monar, J. (2014), 'EU Internal Security Governance: The Case of Counter-Terrorism', European Security, 23(2), 195–209.
- Monar, J. (2015), 'The EU as an International Counter-Terrorism Actor: Progress and Constraints', *Intelligence and National Security*, 30(2–3), 333–56.
- Moore, A. (2014), 'CSDP Police Missions: Comparing Bottom-up and Top-down Approaches', *European Foreign Affairs Review*, 19(2), 283–306.
- Morillas, P. (2012), 'Institutionalization or Intergovernmental Decision-Taking in Foreign Policy: The Implementation of the Lisbon Treaty', in Cardwell, P. J. (ed.), *EU External Relations Law and Policy in the Post-Lisbon Era*, The Hague: T. M. C. Asser Press, pp. 119–34.
- Mortera-Martinez, C. (2017), Hard Brexit, Soft Data: How to Keep Britain Plugged into EU Databases, Centre for European Reform Insight.
- Mounier, G. (2009), 'Europol: A New Player in the External Policy Field?', *Perspectives on European Politics and Society*, 10(4), 49–73.
- Mowatt-Larssen, R. (2010, 25 January), 'Al Qaeda's Pursuit of Weapons of Mass Destruction', *Foreign Policy*, accessed 10 September 2019 at https://foreignpolicy.com/2010/01/25/al-qaedas-pursuit-of-weapons-of-mass-destruction/.
- Mueller, J. (2005), 'Six Rather Unusual Propositions about Terrorism', *Terrorism and Political Violence*, 17(4), 487–505.
- Mueller, J. (2006), Overblown: How Politicians and the Terrorism Industry Inflate National Security Threats, and Why We Believe Them, New York: Simon & Schuster.
- Mukhopadhyay, A. (2004), 'EU Counter-Terrorism Coordinator Office: A Non-Starter?', Observer Research Foundation.
- Müller-Wille, B. (2008), 'The Effect of International Terrorism on EU Intelligence Co-Operation', *Journal of Common Market Studies*, 46(1), 49–73.

- NATO (2001, 12 September), 'Statement by the North Atlantic Council', accessed 20 June 2018 at http://www.nato.int/docu/pr/2001/p01-124e.htm.
- NATO (2018a), 'Countering Terrorism', accessed 20 April 2020 at https://www.nato.int/cps/en/natohq/topics 77646.htm.
- NATO (2018b), 'Joint Declaration on EU-NATO Cooperation', accessed 20 April 2020 at https://www.nato.int/cps/en/natohq/official texts 156626.htm.
- NATO Multimedia Library (n.d.), 'Intelligence/Information Sharing in Combating Terrorism', accessed 20 April 2020 at http://www.natolibguides.info/intelligence.
- NBC (2015, 5 July), 'As Heroin Use Grows in U.S., Poppy Crops Thrive in Afghanistan', accessed 21 July 2017 at http://www.nbcnews.com/news/world/heroin-use-grows-u-s-poppy-crops-thrive-afghanistan-n388081.
- Neumann, P. R. (2009), Old and New Terrorism, Cambridge: Polity.
- Neumann, P. R. (2016), Radicalized: New Jihadists and the Threat to the West, London: I. B. Tauris.
- New Transatlantic Agenda (1995), 'New Transatlantic Agenda', available at https://www.europarl.europa.eu/cmsdata/124321/new_transatlantic_agenda_en.pdf.
- New York Times (1995, 31 December), 'McVEIGH'S MIND: A Special Report; Oklahoma Bombing Suspect: Unravelling of a Frayed Life', accessed 11 January 2019 at https://www.nytimes.com/1995/12/31/us/mcveigh-s-mind-special-report-oklahoma-bombing-suspect-unraveling-frayed-life.html.
- New York Times (2007, 7 October), 'Syria Reportedly Encourages Sunni Insurgents', accessed 2 April 2019 at https://www.nytimes.com/2007/10/07/world/africa/07iht-syria.1.7781943.html.
- New York Times (2009), 'Training of Afghan Police by Europe is Found Lacking', accessed 9 March 2022 at https://www.nytimes.com/2009/11/18/world/europe/18training.html.
- New York Times (2017, 22 October), 'ISIS Fighters Are Not Flooding Back Home to Wreak Havoc as Feared', accessed 20 May 2019 at https://www.nytimes.com/2017/10/22/us/politics/fewer-isis-fighters-returning-home.html.
- Newsome, B. O. and Jarmon, J. A. (2016), A Practical Introduction to Homeland Security and Emergency Management: From Home to Abroad, Thousand Oaks, CA: CQ Press.
- Niemann, A. and Bretherton, C. (2013), 'EU External Policy at the Crossroads: The Challenge of Actorness and Effectiveness', *International Relations*, 27(3), 261–75.
- Nilsson, Hans G. (2006), 'The EU Action Plan on Combating Terrorism: Assessment and Perspectives', in Mahncke, D. and Monar, J. (eds), *International Terrorism: A European Response to a Global Threat?*, Frankfurt: Peter Lang, pp. 73–81.
- Nollkaemper, A. (2008), 'The European Courts and the Security Council: Between "Dédoublement Fonctionnnel" and the Balancing of Values Three Replies to Pasquale De Sena and Maria Chiara Vitucci', *European Journal of International Law*, 20(3), 853–87.
- Norman, P. (2001, 6 December), 'Common Criminals', Financial Times.
- Occhipinti, J. (2003), *The Politics of EU Police Cooperation: Towards a European FBI?*, Boulder, CO: Lynne Rienner.
- Occhipinti, J. (2008), 'Progress Amid Difference: The EU and US Counter-terrorism', Paper presented at the Annual Conference of the International Studies Association (ISA), San Francisco, CA, March 2008.
- Occhipinti, J. (2010), 'Partner or Pushover? EU Relations with the US on Internal Security', in Hamilton, D. S. (ed.), Shoulder to Shoulder: Forging a Strategic US-EU

- *Partnership*, Washington, DC: Johns Hopkins University Centre for Transatlantic Relations, pp. 121–38.
- Occhipinti, J. (2015), 'Still Moving Toward a European FBI? Re-Examining the Politics of EU Police Cooperation', *Intelligence and National Security*, 30(2–3), 234–58.
- O'Connor, S. (2010), 'The Accidental Diplomat', *E! Sharp*, January/February, 12–14. OECD (2002), 'The Limits and Scope for the Use of Development Assistance Incentives and Disincentives for Influencing Conflict Situations'.
- Oliveira Martins, B. and Ferreira-Pereira, L. (2012), 'Stepping Inside? CSDP Missions and EU Counter-Terrorism', *European Security*, 21(4), 537–56.
- Olivier, C. (2004), 'Human Rights Law and International Fight against Terrorism: Could Security Council Regulations Impact on States' Obligations under International Human Rights Law?', *Nordic Journal of International Law*, 73, 399–419.
- O'Neill, M. (2012), *The Evolving EU Counter-Terrorism Legal Framework*, London: Routledge.
- Open Europe (2010), 'Spain Claims Support from EU's Big Three for New Counter-Terrorism Body'.
- Parliament.uk (2016a), 'Chapter 2: Europol and Eurojust', accessed 14 January 2019 at https://publications.parliament.uk/pa/ld201617/ldselect/ldeucom/77/7705.htm.
- Parliament.uk (2016b), 'Brexit: Future UK–EU Security and Police Cooperation', accessed 26 June 2017 at https://www.publications.parliament.uk/pa/ld201617/ldselect/ldeucom/77/7704.htm#footnote-198-backlink.
- Parliament.uk (2017), 'The Government's Negotiating Objectives: The Rights of UK and EU Citizens', accessed 28 June 2017 at https://www.publications.parliament.uk/pa/cm201617/cmselect/cmexeu/1071/107101.htm.
- Parliament.uk (2018), 'UK-EU Security Cooperation after Brexit: 3 Europol', accessed 14 January 2019 at https://publications.parliament.uk/pa/cm201719/cmselect/cmhaff/635/63506.htm.
- Paul, A. and Seyrek, D. M. (2018), 'Two Years after the Brussels Attacks, the Terrorist Threat Remains Very Real', *EURACTIV.COM*.
- Paul, D. (2007), 'The Siren Song of Geopolitics: Towards a Gramscian Account of the Iraq War', *Millennium Journal of International Studies*, 36(1), 51–76.
- Pawlak, P. (2009a), *Made in the USA? The Influence of the US on the EU's Data Protection Regime*, Brussels: Centre for European Policy Studies.
- Pawlak, P. (2009b), 'The External Dimension of the Area of Freedom, Security and Justice: Hijacker or Hostage of Cross-pillarization?', *Journal of European Integration*, 31(1), 25–44.
- Pawlak, P. (2009c), 'Transatlantic Networks and Transatlantic Homeland Security Cooperation', *Perspectives on European Politics and Society*, 10(4), 560–81.
- Pawlak, P. (2010), 'Transatlantic Homeland Security Cooperation: The Promise of New Modes of Governance in International Affairs', *Journal of Transatlantic Studies*, 8(2), 142–60.
- Peers, S. (2001), 'Proposed Framework Decision on European Arrest Warrants', Statewatch Post-11.09.01 Analyses, No. 3, available at www.statewatch.org.
- Piazza, J. (2009), 'Is Islamist Terrorism More Dangerous?: An Empirical Study of Group Ideology, Organisations, and Goal Structure', *Terrorism and Political Violence*, 21(1), 62–88.
- Piris, J. C. (2010), *The Lisbon Treaty: A Legal and Political Analysis*, Cambridge: Cambridge University Press.

- Pisoiu, D. (2015), 'Subcultural Theory Applied to Jihadi and Right-Wing Radicalisation in Germany', *Terrorism and Political Violence*, 27(1), 9–28.
- Plater-Zyberk, H. (2017), Intelligence Security in the European Union: Building a Strategic Intelligence Community, Oxford: Oxford University Press.
- Poli, S. and Tzanou, M. (2009), 'The *Kadi* Rulings: A Survey of the Literature', *Yearbook of European Law*, 28(1), 533–58.
- Portela, C. (2009–10), 'National Implementation of United Nations Sanctions', *International Journal*, 65(1), 13–30.
- Post, J. M. (2005), 'The Psychology of WMD Terrorism', *International Studies Review*, 7(1), 148–51.
- Post, J. M., McGinnis, C. and Moody, K. (2014), 'The Changing Face of Terrorism in the 21st Century: The Communications Revolution and the Virtual Community of Hatred', *Behavioural Sciences and the Law*, 32(3), 306–34.
- Puetter, U. and Wiener, A. (2007), 'Accommodating Normative Divergence in European Foreign Policy Co-ordination: The Example of the Iraq Crisis', *Journal of Common Market Studies*, 45(5), 1065–88.
- Quillen, C. (2016), 'The Islamic State's Evolving Chemical Arsenal', Studies in Conflict & Terrorism, 39(11), 1019–30.
- Raik, K. and Järvenpää, P. (2017), A New Era of EU–NATO Cooperation: How to Make the Best of a Marriage of Necessity, International Centre for Defence and Security.
- Rapoport, D. (2001), 'The Fourth Wave: September 11 in the History of Terrorism', Current History, 100(650), 419–24.
- Rapoport, D. (2012), 'The Four Waves of Modern Terrorism', in Horgan, J. and Braddock, K. (eds), *Terrorism Studies: A Reader*, Abingdon: Routledge, pp. 41–60.
- Rashid, A. (2010), *Taliban: The Power of Militant Islam in Afghanistan and Beyond*, London: I. B. Tauris.
- Rasmussen, M. (2002), "A Parallel Globalization of Terror": 9/11, Security and Globalization, *Cooperation and Conflict*, 37(3), 323–49.
- Ravndal, J. (2018), 'Right-Wing Terrorism and Militancy in the Nordic Countries: A Comparative Case Study', *Terrorism and Political Violence*, 30(5), 772–92.
- Raymonda, C. Z. (2006), 'Maritime Terrorism in Southeast Asia: A Risk Assessment', Terrorism and Political Violence, 18, 239–57.
- Rees, W. (2006), Transatlantic Counter-terrorism Cooperation: The New Imperative, London: Routledge.
- Rees, W. (2008), 'Inside Out: The External Face of EU Internal Security Policy', Journal of European Integration, 30(1), 97–111.
- Rees, W. (2009), 'Securing the Homelands: Transatlantic Co-operation after Bush', The British Journal of Politics and International Relations, 11(1), 108–21.
- Rees, W. and Aldrich, R. (2005), 'Contending Cultures of Counterterrorism: Transatlantic Divergence or Convergence?', *International Affairs*, 81(5), 905–23.
- Reich, J. (2008), 'Due Process and Sanctions Targeted against Individuals Pursuant to UN Resolution 1267 (1999)', Yale Journal of International Law, 33(2), 505–11.
- Reinares, F. (ed.) (2000), European Democracies against Terrorism: Government Policies and Intergovernmental Cooperation, Aldershot: Ashgate.
- Reinares, F. (2010), 'The Madrid Bombings and Global Jihadism', *Survival*, 52(2), 83–104.
- Reuters (2009, 20 November), 'EU Says It Has Solved the Kissinger Question', accessed 20 April 2020 at https://www.reuters.com/article/us-eu-president-kissinger -idUSTRE5AJ00B20091120.

- Reuters (2016, 2 December), 'Europe Must Brace For New, Varied Islamic State Attacks – Security Chief', accessed 13 November 2017 at http://uk.reuters.com/ article/uk-eu-security/europe-mustbrace-for-new-varied-islamic-state-attacks -security-chief.
- Reuters (2019a, 24 January), 'Syria's Assad Suspends Special Visas for EU Diplomats', accessed 21 May 2019 at https://uk.reuters.com/article/uk-mideast-crisis-syria-eu/syrias-assad-suspends-special-visas-for-eu-diplomats-idUKKCN1PI1K2.
- Reuters (2019b, 28 February), 'As Assad Holds on in Syria, EU's "Strategic Patience" Tested', accessed 12 July 2019 at https://www.reuters.com/article/us -mideast-crisis-syria-eu/as-assad-holds-on-in-syria-eus-strategic-patience-tested -idUSKCN1OH1RO.
- Rhinard, M., Boin, A. and Ekengren, M. (2007), 'Managing Terrorism: Institutional Capacities and Counter-terrorism Policy in the EU', in Spence, D. (ed.), *The European Union and Terrorism*, London: John Harper Publishing, 88–104.
- Riedel, B. (2008), 'Pakistan and Terror: The Eye of the Storm', *Annals of the American Academy of Political and Social Science*, 618, 31–45.
- Ripoll Servent, A. (2009), 'Setting Priorities: Functional and Substantive Dimensions of Irregular Immigration and Data Protection Under Co-decision', *Journal of Contemporary European Research*, 5(2), 225–42.
- Ripoll Servent, A. (2010), 'Point of No Return? The European Parliament after Stockholm and Lisbon', *European Security*, 19(2), 191–207.
- Ripoll Servent, A. (2013), 'Holding the European Parliament Responsible: Policy Shift in the Data Retention Directive from Consultation to Codecision', *Journal of European Public Policy*, 20(7), 972–87.
- Ripoll Servent, A. (2015), *Institutional and Policy Change in the European Parliament* Deciding on Freedom, Security, and Justice, Basingstoke: Palgrave Macmillan.
- Ripoll Servent, A. and MacKenzie, A. (2011), 'Is the EP Still a Data Protection Champion? The Case of SWIFT', *Perspectives on European Politics and Society*, 12(4), 390–406.
- Ripoll Servent, A. and MacKenzie, A. (2012), 'The European Parliament as a "Norm-taker"? EU–US Relations after the SWIFT Agreement', *European Foreign Affairs Review*, 17(2), 71–86.
- Ripoll Servent, A. and MacKenzie, A. (2017), 'Eroding Germany's Commitment to Data Protection: Policy Entrepreneurs and Coalition Politics in EU Passenger Name Records', *German Politics*, 26(3), 398–413.
- Roe, P. (2008), 'Actor, Audience(s) and Emergency Measures: Securitization and the UK's Decision to Invade Iraq', *Security Dialogue*, 39(6), 615–35.
- Rogers, P. (2014), 'Lost Cause: Consequences and Implications of the War on Terror', *Critical Studies on Terrorism*, 6(1), 13–28.
- Rosand, E. (2003), 'Security Council Resolution 1373, the Counter-terrorism Committee, and the Fight against Terrorism', *The American Journal of International Law*, 97(2), 333–41.
- Rosand, E. (2004), 'The Security Council's Efforts to Monitor the Implementation of Al Qaeda/Taliban Sanctions', *The American Journal of International Law*, 98(4), 745–63.
- Rotella, S. (2016, 18 October), 'How Europe Left Itself Open to Terrorism', *Frontline*, accessed 20 April 2020 at https://www.pbs.org/wgbh/frontline/article/how-europe -left-itself-open-to-terrorism/.
- Roy, O. (2017), *Jihad and Death: The Global Appeal of Islamic State*, London: Hurst & Company.

- Rudischhauser, W. (2015), 'Could ISIL Go Nuclear?', NATO Review, accessed 11 October 2019 at https://www.nato.int/docu/Review/2015/ISIL/ISIL-Nuclear -Chemical-Threat-Iraq-Syria/EN/index.htm.
- Rummel, R. (2002), 'From Weakness to Power with ESDP', European Foreign Affairs Review, 7(4), 453–71.
- Sandholtz, W. (1993), 'Institutions and Collective Action: The New Telecommunications in Western Europe', *World Politics*, 45(2), 242–70.
- Sandholtz, W. and Stone Sweet, A. (eds) (1998), European Integration and Supranational Governance, Oxford: Oxford University Press.
- Sandler, T. (2015), 'Terrorism and Counterterrorism', Oxford Economic Papers, 67(1), 1–20.
- Schmidt, J. (2011), 'The High Representative, the President and the Commission—Competing Players in the EU's External Relations: The Case of Crisis Management', in Cardwell, P. J. (ed.), *EU External Relations Law and Policy in the Post-Lisbon Era*, Dordrecht: Springer, pp. 161–80.
- Schmimmelfennig, F. (2004), Liberal Intergovernmentalism, in Wiener, A. and Diez, T. (eds), European Integration Theory, Oxford: Oxford University Press, 75–94.
- Schultz, R. and Vogt, A. (2003), 'It's War! Fighting Post-11 September Global Terrorism Through a Doctrine Of Preemption', *Terrorism and Political Violence*, 15 (1), 1–30.
- Schuurman, B. (2019), 'Topics in Terrorism Research: Reviewing Trends and Gaps, 2007–2016', *Critical Studies in Terrorism*, 12(3), 463–80.
- Schweiger, C. (2004), 'British-German Relations in the European Union after the War on Iraq', German Politics, 13(1), 35–55.
- Selby, J., Dahl, O., Frohlich, C. and Hulme, M. (2017), 'Climate Change and the Syrian Civil War Revisited', *Political Geography*, 60, 232–44.
- Shepherd, A. (2006), 'Irrelevant or Indispensable? ESDP, the "War on Terror" and the Fallout from Iraq', *International Politics*, 43(1), 71–92.
- Siegel, S. (2009), 'Bearing Their Share of the Burden: Europe in Afghanistan', *European Security*, 18(4), 461–82.
- Sjöstedt, G. (1977), *The External Role of the European Community*, London: Saxon House.
- Smith, H. (2002), European Union Foreign Policy: What It is and What It Does, London: Pluto.
- Soare, S. R. (ed.) (2020), *Turning the Tide: How to Rescue Transatlantic Relations*, Paris: European Union Institute for Security Studies.
- Solana, J. (2004, 25 March), 'Three Ways for Europe to Prevail against the Terrorists', Financial Times.
- Sommers, J. (2011, 5 September), 'EU Anti-Terror Chief Highlights Arab Spring Risks', Reuters, available at https://www.reuters.com/article/us-eu-arabspring-risk-idUSTRE7843TV20110905.
- Special Inspector General for Afghanistan Reconstruction (2021), 'What We Need to Learn: Lessons from Twenty Years of Afghanistan Reconstruction', Arlington, VA: SIGAR.
- Spence, D. (2007a), 'Conclusion: The Continuing Quest for Coherence: Sovereignty, Human Rights, and EU Co-ordination', in Spence, D. (ed.), *The European Union and Terrorism*, London: John Harper, 168–176.
- Spence, D. (2007b), 'Introduction: International Terrorism The Quest for a Coherent EU Response', in Spence, D. (ed.), *The European Union and Terrorism*, London: John Harper, 1–29.

- Spence, D. (ed.) (2007c), *The European Union and Terrorism*, London: John Harper. Sperling, J. and Webber, M. (2016), 'NATO and the Ukraine Crisis: Collective Securitisation', *European Journal of International Security*, 2(1), 19–46.
- Sperling, J. and Webber, M. (2019), 'The European Union: Security Governance and Collective Securitisation', *West European Politics*, 42(2), 228–60.
- Stavrianakis, A. (2018), 'When "Anxious Scrutiny" of Arms Exports Facilitates Humanitarian Disaster', *The Political Quarterly*, 89(1), 92–9.
- Stern, J. (1999), The Ultimate Terrorists, Cambridge, MA: Harvard University Press.
- Steuter, E. and Wills, D. (2010), "The Vermin Have Struck Again": Dehumanizing the Enemy in Post 9/11 Media Representations', *Media, War & Conflict*, 3(2), 152–67.
- Stevenson, J. (2019), 'Right-Wing Extremism and the Terrorist Threat', Survival, 61(1), 233-44.
- Stone Sweet, A., Fligstein, N. and Sandholtz, W. (2001), 'The Institutionalisation of European Space', in Stone Sweet, A., Sandholtz, W. and Fligstein, N. (eds), *The Institutionalisation of Europe*. Oxford: Oxford University Press, 1–28.
- Strack, C. (2017), 'The Evolution of the Islamic State's Chemical Weapons Efforts', CTC Sentinel, 10(9), 19–23.
- Stritzel, H. (2007), 'Towards a Theory of Securitisation: Copenhagen and Beyond', European Journal of International Relations, 13(3), 357–83.
- Stromseth, J. E. (2003), 'The Security Council's Counter-terrorism Role: Continuity and Innovation', Proceedings of the American Society of International Law, 97, 41–5.
- Sultygov, A. K. (2019, 4 March), 'Did Trump Really Defeat the Islamic State?', *National Interest*, accessed 2 April 2019 at https://nationalinterest.org/feature/did-trump-really-defeat-islamic-state-46127.
- SWIFT (n.d.), 'Discover SWIFT', accessed 15 January 2019 at https://www.swift.com/about-us/discover-swift.
- Talmon, S. (2005), 'The Security Council as World Legislature', *The American Journal of International Law*, 97(4), 873–88.
- Tammiko, T. (2018), 'The Threat of Returning Foreign Fighters: Finnish State Responses to the Volunteers in the Spanish and Syria–Iraq Civil Wars', *Terrorism and Political Violence*, 30(5), 844–61.
- Tan, S. S. (2018), 'Sending in the Cavalry: The Growing Militarization of Counterterrorism in Southeast Asia', PRISM, 7(4), 138–47.
- Tan, S. S. and Nasu, H. (2016), 'ASEAN and the Development of Counter-Terrorism Law and Policy in South Asia', *UNSW Law Journal*, 39(3), 1219–38.
- Tappeiner, I. (2005), 'The Fight against Terrorism: The Lists and the Gaps', *Utrecht Law Review*, 1(1), 97–125.
- Taylor, A. (2014), 'The Soviet War in Afghanistan, 1979–1989', *The Atlantic*, accessed 18 January 2019 at https://www.theatlantic.com/photo/2014/08/the-soviet-war-in -afghanistan-1979-1989/100786/.
- Telegraph (2017, 6 June), 'Khuram Butt, Rachid Redouane and Youssef Zaghba Named: Everything We Know about the London Bridge Terrorists', accessed 13 November 2017 at http://www.telegraph.co.uk/news/2017/06/05/ringleader-london-bridge-terror-attack-named-khuram-butt/.
- Terpan, F. (2013), 'Soft Law in the European Union: The Changing Nature of EU Law', *Sciences Po Grenoble Working Paper No. 7*, Grenoble: Sciences Po Grenoble.
- The Economist (2019), 'After America Leaves Syria, What Next?', accessed 21 May 2019 at https://www.economist.com/middle-east-and-africa/2019/01/03/after-america-leaves-syria-what-next.

- The Hague Centre for Strategic Studies (2016), *The Increasing Threat of Biological Weapons: Handle with Sufficient and Proportionate Care*, The Hague: The Hague Centre for Strategic Studies.
- The Parliamentary Office of Science and Technology (2004), 'Terrorist Attacks on Nuclear Facilities', July 2004, Number 222, London: The Parliamentary Office of Science and Technology.
- The Soufan Group (2015), 'Foreign Fighters: An Updated Assessment of the Flow of Foreign Fighters into Syria and Iraq'.
- Times Online (2004), 'EU Divided Over Proposal for New Anti-Terror Czar'.
- Torelli, S. M. (2017), European Union and the External Dimension of Security: Supporting Tunisia as a Model in Counter-Terrorism Cooperation, European Institute of the Mediterranean.
- Toro, C. (2010), 'External State Partners in ESDP Missions: Third Country Participation', European Foreign Affairs Review, 15(3), 325–45.
- Tranholm-Mikkelsen, J. (1991), 'Neo-functionalism: Obstinate or Obsolete? A Reappraisal in the Light of the New Dynamism of the EC', *Millennium: Journal of International Studies*, 20(1), 1–22.
- Transatlantic Declaration (1990), 'Transatlantic Declaration on EC-US Relations'.
- Tridimas, T. (2009), 'Terrorism and the ECJ: Empowerment and Democracy in the EC Legal Order', *European Law Review*, 34(1), 103–26.
- Tridimas, T. and Gutierrez-Fons, J. A. (2008), 'EU Law, International Law, and Economic Sanctions against Terrorism: The Judiciary in Distress?', *Fordham International Law Journal*, 32(2), 660–730.
- Turkmani, R. and Haid, M. (2016), *The Role of the EU in the Syrian Conflict*, Friedrich Ebert Stiftung, accessed 20 April 2020 at https://www.fes-europe.eu/fileadmin/public/editorfiles/events/Maerz_2016/FES_LSE_Syria_Turkmani_Haid_2016_02 23.pdf.
- Tzanou, M. (2015), 'The War Against Terror and Transatlantic Information Sharing: Spillovers of Privacy or Spillovers of Security?', Utrecht Journal of International and European Law, 31(80), 87–103.
- UN News (2019, 6 March), 'Security Council Discusses Chemical Weapons Use in Syria Following Latest Global Watchdog Report', accessed 10 March 2019 at https://news.un.org/en/story/2019/03/1034201.
- UNHCR (2018a, 7 March), 'Seven Facts about the Syria Crisis', accessed 4 February 2019 at https://www.unhcr.org/ph/13418-seven-facts-syria-crisis.html.
- UNHCR (2018b), 'Iraq Emergency', accessed 21 May 2019 at https://www.unhcr.org/ uk/iraq-emergency.html.
- United Nations (2001a), 'Agreement on Provisional Arrangements in Afghanistan Pending the Re-establishment of Permanent Governing Institutions', accessed 17 January 2019 at http://www.un.org/News/dh/latest/afghan/afghan-agree.htm.
- United Nations (2001b), Resolution 1378, accessed 20 April 2020 at http://unscr.com/en/resolutions/1378.
- United Nations (2001c, 12 September), 'Opening its Fifty-Sixth Session, General Assembly Condemns Heinous Acts of Terrorism Perpetrated in Host City and Washington', UN Press Release GA/9903, accessed 20 June 2018 at http://www.un.org/News/Press/docs/2001/ga9903.doc.htm.
- United Nations (2019), 'European Union Investing "As Never Before" in Cooperation with United Nations System, High Representative Tells Security Council', accessed 12 July 2019 at https://www.un.org/press/en/2019/sc13735.doc.htm.

- United Nations Development Programme (2010), 'Law and Order Trust Fund for Afghanistan (LOTFA) Phase V', accessed 25 January 2011 at http://www.undp.org.af/whoweare/undpinafghanistan/Projects/sbgs/prj_lotfa.htm.
- United Nations General Assembly Resolution 71/291 (2017), accessed 20 April 2020 at https://undocs.org/pdf?symbol=en/a/res/71/291.
- United Nations Office on Drugs and Crime UNODC (2018), 'Supporting Legal Responses and Criminal Justice Capacity to Prevent Counter Terrorism' (United Nations).
- United Nations Security Council Resolution 748 (1992), accessed 20 April 2020 at https://www.webcitation.org/6lazEBgO0?url=http://www.treasury.gov/resource-center/sanctions/Documents/748.pdf.
- United Nations Security Council Resolution 1070 (1996), accessed 20 April 2020 at https://www.refworld.org/docid/3b00f20b8.html.
- United Nations Security Council Resolution 1267 (1999), accessed 20 April 2020 at https://www.undocs.org/S/RES/1267(1999).
- United Nations Security Council Resolution 1368 (2001), accessed 20 April 2020 at https://undocs.org/S/RES/1368(2001).
- United Nations Security Council Resolution 1373 (2001), accessed 20 April 2020 at https://www.unodc.org/pdf/crime/terrorism/res 1373 english.pdf.
- United Nations Security Council Resolution 1540 (2004), accessed 20 April 2020 at http://archive.ipu.org/splz-e/civ1540/1540.pdf.
- United Nations Security Council Resolution 2178 (2014), accessed 20 April 2020 at http://www.securitycouncilreport.org/atf/cf/%7B65BFCF9B-6D27-4E9C-8CD3 -CF6E4FF96FF9%7D/s res 2178.pdf.
- United Nations Security Council Resolution 2322 (2016), accessed 20 April 2020 at https://undocs.org/S/RES/2322(2016).
- Van den Herik, L. J. (2007), 'The Security Council's Targeted Sanctions Regimes: In Need of Better Protection for the Individual', *Leiden Journal for International Law*, 20(4), 69–79.
- Van den Herik, L. J. and Schrijver, N. (2008), 'Eroding the Primacy of the UN System of Collective Security: The Judgment of the European Court of Justice in the Cases of Kadi and Al Barakaat', *International Organizations Law Review*, 5(2), 329–38.
- Van Thiel, S. (2008), 'UN Anti-terrorism Sanctions and EU Human Rights: The Lessons of European Integration', in Martenczuk, B. and van Thiel, S. (eds), *Justice*, *Liberty, Security: New Challenges for EU External Relations*, Brussels: VUB Press/ Brussels University Press, pp. 437–92.
- Vanhoonacker, S. and Reslow, N. (2010), 'The European External Action Service: Living Forwards by Understanding Backwards', European Foreign Affairs Review, 15(1), 1–18.
- Vlcek, W. (2009), 'Hitting the Right Target: EU and Security Council Pursuit of Terrorist Financing', Critical Studies on Terrorism, 2(2), 275–91.
- Vogel, J. (2001), 'Abschaffung Der Auslieferung? Kritische Anmerkungen Zur Reform Des Auslieferungsrechts In Der Europäischen Union', *Juristenzeitung*, 56(19), 937–43.
- Vohra, A. (2019, 8 March), 'Europe Doesn't Even Agree on Assad Anymore', Foreign Policy, accessed 21 May 2019 at https://foreignpolicy.com/2019/03/08/europe -doesnt-even-agree-on-assad-anymore/.
- Wæver, O. (1995), 'Securitization and Desecuritization', in Lipschutz, R. D. (ed.), *On Security*, New York: Columbia University Press, pp. 48–86.

- Wagner, W. (2003), 'Building an Internal Security Community: The Democratic Peace and the Politics of Extradition in Western Europe', *Journal of Peace Research*, 40(6), pp.695–712.
- Walt, S. (2015), 'ISIS as Revolutionary State', Foreign Affairs, 94, 42-51.
- Waltz, K. (1979), Theory of International Politics, New York: McGraw-Hill.
- Washington Post (2018), 'U.S. Diplomats Hold Talks with Taliban on Ending Afghanistan War', accessed 17 January 2019 at https://www.washingtonpost.com/world/us-diplomats-hold-talks-with-taliban-on-ending-afghan-war/2018/12/17/c25be5d0-01f3-11e9-9122-82e98f91ee6f_story.html?noredirect=on&utm_term=c0f799c1b6a0.
- Watt, Nicholas (2008, 18 March), 'Extent of Secret Links between Government and IRA Revealed', *The Guardian*, accessed 20 June 2018 at https://www.theguardian.com/politics/2008/mar/18/northernireland.past.
- Weinberg, L. (2011), The End of Terrorism, London: Routledge.
- Weinberg, L. and Eubank, W. (2010), 'An End to Fourth Wave of Terrorism?', Studies in Conflict and Terrorism, 33(7), 594–602.
- Weiss, M. and Dalferth, S. (2009), 'Security Re-Divided: The Distinctiveness of Policy-Making in ESDP and JHA', *Co-operation and Conflict*, 44(3), 268–87.
- Weissman, T. (2005), 'The Spectacle of Trauma: 9/11 in the Museum', Visual Resources, 21(2), 155–70.
- Wennerholm, P., Brattberg, E. and Rhinard, M. (2010), 'The EU as a Counter-terrorism Actor Abroad: Finding Opportunities, Overcoming Constraints', *European Policy Centre Issue Paper*, 60, Brussels: European Policy Centre.
- Wensink, W., Warmenhoven, B., Haasnoot, R., Wesselink, R., Van Ginkel, B., Wittendorp, S., Paulussen, C., Douma, W., Boutin, B., Gueven, O. and Rijken, T. (2017), The European Union's Policies on Counter-Terrorism Relevance, Coherence and Effectiveness, European Parliament: Policy Department for Citizens' Rights and Constitutional Affairs.
- Wessel, R. A. (2006), 'Editorial: The UN, the EU and Jus Cogens', International Organizations Law Review, 3(1), 1–6.
- White House (2002), The National Security Strategy of the United States of America, Washington, DC: White House.
- White House (2006), *National Strategy for Combating Terrorism*, Washington, DC: White House.
- Whittaker, D. (2003), The Terrorism Reader, London: Routledge.
- Wilkinson, P. (2006), Terrorism versus Democracy, London: Routledge.
- Wilkinson, P. (2007), 'International Dimensions of Homeland Security', in Wilkinson, P. (ed.), *Homeland Security in the UK: Future Preparedness for Terrorist Attack since 9/11*, London: Routledge, 371–378.
- Williams, A. and Parkin, B. (2021, 31 August), 'US Withdraws from Afghanistan, Bringing an End to 20-Year War', *Financial Times*.
- Withnall, A. (2015, 19 November), 'Paris Attacks: Isis "Chemical Weapons" Warning Issued by French PM Manuel Valls', *Independent*.
- Wolff, S. (2009), 'The Mediterranean Dimension of EU Counter-terrorism', *Journal of European Integration*, 31(1), 137–56.
- Wolff, S., Wichmann, N. and Mounier, G. (2009), 'The External Dimension of Justice and Home Affairs: A Different Security Agenda for the EU?', *Journal of European Integration*, 31(1), 9–23.
- Wouters, J. and Chané, A. L. (2016), *Brussels Meets Westphalia: The European Union and the United Nations*, Oxford: Hart Publishing.

- Wouters, J., de Jong, S. and De Man, P. (2010), 'The EU's Commitment to Effective Multilateralism in the Field of Security: Theory and Practice', *Yearbook of European Law*, 29(1), 164–89.
- Wright, S. (2007), 'Terrorists and Biological Weapons: Forging the Linkage in the Clinton Administration', *Politics and the Life Sciences*, 25(1), 57–115.
- Zimmermann, D. (2006), 'The European Union and Post-9/11 Counter-Terrorism: A Reappraisal', *Studies in Conflict and Terrorism*, 29(2), 123–45.
- Zorob, A. (2008), 'The Syrian-European Association Agreement and its Potential Impact on Enhancing the Credibility of Reform', *Mediterranean Politics*, 13(1), 1–21.

Index

actorness 10–11, 13–16, 19, 21, 23–34, 45, 74, 76, 92, 95, 100, 109, 125, 127–128, 131, 148, 156–157, 161–162, 166	Bush, G. W. 17, 39–41, 66, 77, 80, 83, 89, 108, 111, 125, 140, 142, 144, 148, 162, 164, 168
African Union (AU) 23, 158 aid 5, 91–94, 97, 101, 111, 114–115, 121, 123, 130, 136, 153, 157 al-Assad, B. 110, 111, 113, 115, 117, 119, 169, 173 Al-Maliki, N. 112 al-Qaeda 1, 2, 7, 12, 26, 30, 38, 48, 67, 78-80, 90, 93, 94, 96, 98, 108, 109, 112, 123, 125–126, 131 al-Qaeda in Iraq (AQI) 112	Canada 12, 30, 46, 53, 60, 74, 89 Central Intelligence Agency (CIA) 56, 154 China 71, 80, 97 co-decision 54, 169 Common Foreign and Security Policy (CFSP) 9, 28, 59, 63–64, 74, 126, 145, 154 see also Common Security and Defence Policy (CSDP); European Security and
al-Qaeda in the Islamic Maghreb (aQIM) 12, 67 al-Zarqawi, A. M. 112 Algeria 11, 12, 61, 75, 78, 107 Arab League 70 Area of Freedom, Security and Justice (AFSJ) 3, 7, 42, 126, 137 see also Justice and Home Affairs (JHA); third pillar Asia-Europe Meeting (ASEM) 61 Association of Southeast Asian Nations (ASEAN) 23–24, 138–139, 141, 171	Defence Policy (ESDP); second pillar Common Security and Defence Policy (CSDP) 12, 33, 65, 70, 145, 148–149, 165, 167 see also Common Foreign and Security Policy (CFSP); European Security and Defence Policy (ESDP); second pillar consent 54–55, 73, 86–88 Container Security Initiative (CSI) 11, 45, 53, 85, 146
Australia 12, 30, 46, 63, 70, 84 Balkans 70, 99 Belgium 37, 87, 109, 144	Court of Justice of the EU 12, 31, 43, 45, 53–54, 85–86, 100, 147, 173 Customs and Border Protection (CBP) 83, 85, 146, 150
Beigium 37, 87, 109, 144 Biden, J. 17, 80–81, 90 border 7, 8, 31, 38, 50, 82–83, 85, 90, 100, 110, 122, 139, 146, 150–151, 163 Brazil 12 Brexit 17, 81, 83, 118, 125, 131, 134–137, 143,158, 160, 165, 167 see also United Kingdom (UK)	Daesh 1–2, 13, 15, 17, 30, 80, 90, 109–110, 112–121, 123, 125–126, 129–132, 136, 160 data protection 20, 54–55, 73, 82, 85–86, 88–89, 135, 137, 143, 167, 169 Data Protection Directive (DPD) 85–86 de Kerchove, G. 46, 56, 59–61, 66–69, 101, 147 de Vries, G. 38, 46, 55–59, 66, 147

Index 177

democracy 33, 52, 77, 81, 92, 96, Gulf War 110, 112 102-103, 114, 122, 141, 150-153, 161, 164, 172, 174 human rights 33, 52, 81, 95–96, 98, Denmark 59 102-103, 111, 114, 121-122, 138, 141, 148, 150–153, 156, 163, 167, Egypt 67 170, 173 ethnonationalist terrorism 36-37, 39, 48 Hussein, S. 108, 110, 112, 117 European Counter Terrorism Centre (ECTC) 35, 47-48 see also implementation 3, 6, 8, 41, 44–47. Europol 57-61, 65, 66, 68, 102-103, 139, Eurojust 33, 65, 81, 84, 142, 150, 167 146, 151–152, 154, 156, 158, 165, European Arrest Warrant (EAW) 7, 44, 168-169 161, 167 India 61, 79, 97 European Neighbourhood Policy intelligence 38, 41, 47-48, 56, 65-66, (ENP) 33, 52, 111, 114, 120, 74, 78, 83, 88, 139, 141–142, 147, 122, 141, 152, 162 European 154, 157–158, 160, 162, 163, Neighbourhood Policy Instrument 165-168 (ENPI) 33, 51-52, 122, 151 Interpol 37, 156 European Security and Defence Policy Iran 28, 110–111, 113–114, 116, 119, (ESDP) 91, 105, 170, 172, 174 140, 148, 154, 162, 164 see also Common Foreign and Iraq 1, 2, 13, 15–17, 21, 30–31, 73, Security Policy (CFSP): Common 81-82, 89, 98, 108-124, 127, Security and Defence Policy 129-130, 132, 140, 145-146, (CSDP); second pillar 149-153, 155, 159-160, 162, Europol 1, 19, 28, 33, 35, 47–48, 57, 167-172 65, 78, 81–83, 88, 99, 140, 142, Islamist terrorism 19, 39, 47, 167 145, 147, 155–157, 161, 165, Italy 3, 11, 19, 35–37, 44, 73, 91, 108, 167 see also European Counter 133 Terrorism Centre (ECTC); Terrorist Financing and Tracking Jordan 115 Programme (TFTP) judicial cooperation 33, 82, 84, 86 extradition 37, 41, 43, 74, 81, 84, 118, Justice and Home Affairs (JHA) 3-4, 9, 145 42-43, 51, 59, 78, 147, 151, 154, 174 see also Area of Freedom, FBI 83, 166-167 Security and Justice (AFSJ); third first pillar 9, 45, 51, 53 pillar Five Eyes alliance 74 foreign fighters 2, 47, 109, 112, 116-118. law enforcement 5, 30, 32, 35, 38, 65, 123, 142, 144, 146, 154, 160, 83, 87, 104, 129, 159, 165 171-172 left-wing terrorism 3, 19, 36–37, 39, 133, France 2-3, 10-11, 19, 36-37, 39, 73, 164 75, 78, 110, 117–118, 137, 144, Mali 10, 12, 67 Frontex 65 Mexico 12 Middle East 2, 4, 15, 70, 79, 90, 110-111, Germany 3, 19, 32, 35–37, 59, 73, 76, 113, 117, 119, 123, 127, 129–130, 91, 109, 117, 133, 134, 137, 142, 132, 156, 162, 164 144, 157–158, 163, 168–169 migration 134, 138, 163-164, 169 Gulf Cooperation Council 70 Morocco 11, 107

```
multilateral(ism) 23, 37, 62, 101, 144,
                                           South Asia 12, 15–17, 21, 81, 90-91, 97,
       157, 159, 162, 174
                                                  101, 107, 127, 130, 171
mutual legal assistance (MLA) 84.
                                           Spain 35-37, 39, 42, 73, 108, 167
       145-146
                                           Syria 1, 2, 13, 15–17, 21, 108–124, 127,
mutual recognition 42, 43
                                                  129–130, 136, 140, 146, 149,
                                                  151-154, 166, 169-172
                                                Syrian Civil War 109, 112-115, 120,
NATO 40, 69, 71, 74, 98, 139, 149, 159,
                                                       123, 132, 170
       161-162, 164, 166, 168, 170
New Zealand 74, 79
                                           Taliban 17, 90, 92-95, 97, 106, 129-130,
                                                  157, 168–169, 174
Obama 17, 80, 144
                                           Tampere European Council 3, 42
                                           terrorist attacks
Passenger Name Records (PNR) 7.
                                                Barcelona 67-68, 70
       11-12, 20-21, 30-31, 45-46,
                                                Berlin 67, 70, 109
      53-55, 69, 72-73, 85-87, 135,
                                                Brussels 35, 39, 67, 70, 109
      139-140, 142, 147, 153, 158, 162
                                                London 16, 29, 39, 50, 65-67, 70,
police cooperation 33, 38, 82, 142,
                                                       109
       166-167
                                                Manchester 67, 70
Police Working Group on Terrorism
                                                Madrid 1, 4, 16, 29, 39, 44-46, 50,
      (PWGOT) 38
                                                       55-56, 65, 68, 98, 140, 168
                                                New York City 38, 72
'refugee crisis' 67, 114, 119, 123,
                                                Nice 39, 67
       132 - 133
                                                Paris 2, 39, 67-68, 70, 109, 158, 174
right-wing terrorism 1, 3, 19, 125-126,
                                                Pentagon 38, 72
       131–134, 141, 149, 158–159,
                                           Terrorist Financing and Tracking
       163-164, 168, 171
                                                  Programme (TFTP) 87–88, 145,
Russia 12, 31, 61, 74, 80, 99, 109–110,
                                                  153, 156 see also Europol
      113, 116, 140
                                           third pillar 3, 9, 45–46, 51, 53, 58, 84,
                                                  86, 143, 147 see also Area of
Sahel 12, 14, 61, 67, 145, 149, 155
                                                  Freedom, Security and Justice
Saudi Arabia 110, 113
                                                  (AFSJ); Justice and Home Affairs
Schengen 29, 38, 118
                                                  (JHA)
Schröder, G. 41
                                           trafficking 47, 95, 98, 122, 126, 146
second pillar 9, 45 see also Common
                                           Treaty of Amsterdam 3, 63, 66, 83, 150
      Foreign and Security Policy
                                           Treaty of Lisbon 28, 46, 50-54, 62-63,
      (CFSP); Common Security and
                                                  73, 83, 86, 88, 127, 143, 150 see
      Defence Policy (CSDP); European
                                                  also Treaty on European Union
      Security and Defence Policy
                                           Treaty of Maastricht 3, 74, 126, 150
      (ESDP)
                                           Treaty of Nice 27, 63
securitization 13-14, 19-49, 140, 143,
                                           Treaty on European Union 27, 84, 151
      158, 161–163, 169, 173
                                                  see also Treaty of Lisbon
sovereignty 8, 37, 72, 76, 149, 156, 170
                                           Trevi Group 3, 37-38, 142, 156
Society for Worldwide Interbank
                                           Trump, D. 17, 71, 80-81, 108, 116, 136,
      Financial Telecommunication
                                                  171
      (SWIFT) 11, 20-21, 31, 46,
                                           Tunisia 67, 172
      53–55, 72, 87–88, 147, 156, 162,
      165, 169, 171
                                           Union for the Mediterranean (UfM) 111
```

Index 179

United Kingdom (UK) 3, 19, 31, 35, 37, 59, 61, 67, 73, 74, 81–83, 87, 91, 94, 97, 99, 101, 108–110, 117–119, 131, 133–137, 140–141, 143, 145–146, 148, 156, 160, 167, 169, 174 see also Brexit
United Nations (UN) 27, 40, 64, 90, 99, 105–106, 110, 119, 144, 147–150.

105–106, 110, 119, 144, 147–150, 154, 156, 158, 168, 172–174 Counter-Terrorism Committee

Counter-Terrorism Committee Executive Directorate (CTED) 62

United Nations Development Programme (UNDP) 105, 106, 173

United Nations Security Council 60, 63–64, 102, 106, 110, 147, 149, 166–167, 169, 171, 173 United States of America (USA) 1–6, 9–12, 15–17, 19–21, 27, 29–32, 38, 40–41, 44–46, 48, 53–55, 60, 71–90, 94, 96–99, 101, 106–108, 110, 112–114, 119, 123, 128–132, 134–136, 167

Vitorino, A. 42, 171

'war on terror' 1–2, 4, 12, 15, 17, 30, 39–40, 42, 44, 54, 71–73, 75, 90, 96–97, 125, 129–130, 132, 137, 147, 157, 163–165, 169–170 weapon of mass destruction (WMD) 95, 112, 143, 168

Yemeni Civil War 110